

Chronic Wet Cough in Children and Further Exploration of Protracted Bacterial Bronchitis

Danielle Fortunée Wurzel

Bachelor of Medicine Bachelor of Surgery

A thesis submitted for the degree of Doctor of Philosophy at

The University of Queensland in 2015

School of Medicine

Abstract

Chronic wet cough (i.e. lasting > 4 weeks) is common in children and is associated with significant burden on children, families and the healthcare system. Although some clinicians view chronic cough in children as trivial, for parents, chronic cough is associated with significant quality of life impairment, which improves when the cough resolves. Further, chronic cough in children may signify the presence of underlying lung disease, such as bronchiectasis.

Prior to this thesis, my supervisors were the first to define the clinical entity of protracted bacterial bronchitis (PBB) in children. PBB is characterised by chronic wet cough, response to antibiotics (suggestive of bacterial infection of the airways) and absence of pointers indicating an alternative cause for cough. This early research demonstrated the existence of a recognisable pattern of lower airway inflammation and bacterial infection in children with chronic wet cough. However, many questions relating to chronic wet cough and PBB remained.

This thesis explored the clinical, demographic, immunological and microbiological characteristics of children with chronic wet cough and PBB, with specific focus on viruses and viral-bacterial interactions. A longitudinal 2-year cohort study was also undertaken to describe the natural history of PBB. The medium-term risk of bronchiectasis (BE), in children with PBB, was evaluated; with the over-arching aim of identifying a link between these two conditions and identifying specific risk factors for a diagnosis of bronchiectasis.

The objectives of the thesis were, to:

- 1. Examine and contrast the lower airway findings of children with different types of cough (wet and dry) compared to no cough, with respect to infection and inflammation.
- Evaluate the sensitivity, specificity and predictive values of nasopharyngeal aspirate (NPA) for detecting viruses in bronchoalveolar lavage (BAL); and determine the relationship between infection and lower airway neutrophilic inflammation in children with non-acute respiratory illness.

- 3. Evaluate the major human adenovirus genotypes/species, and relationships to bacterial co-infection, in children with PBB and BE.
- 4. Delineate the clinical and laboratory characteristics of children with PBB with a specific focus on viruses.
- 5. Determine the 2-year outcomes and risk factors for BE and recurrent PBB, in children with PBB.
- 6. Undertake a systematic review on the efficacy of short-course antibiotic therapy for bronchiectasis in children and review specific treatment options for cough in children in general practice.

Five of these objectives have been published (as 6 papers) and presented as chapters. The remaining chapter (objective 5) has been submitted to a peer-reviewed journal. An overview of the major findings, with respect to each chapter is detailed below:

Objective-1 is a study on 232 children undergoing bronchoscopy for evaluation of respiratory symptoms. Children were grouped using a cough nature symptom-based approach into "wet cough," "dry cough" and "no cough". Children with wet cough were more likely to have lower airway infection (bacterial, viral and viral-bacterial). Viral-bacterial co-infection was associated with maximal lower airway neutrophilic inflammation.

Objective-2 included extended viral panel analyses on paired NPA and BAL specimens from 75 children with chronic respiratory symptoms. We found that the agreement between NPA and BAL, with respect to virus detection using PCR, was dependent on viral species. A good agreement was seen for adenoviruses and poor agreement for rhinoviruses. Adenovirus, but not rhinovirus, positivity in the airway was associated with lower airway neutrophilic inflammation.

Objective-3 examined BAL samples from 245 children with PBB and BE and found a strong association between human adenovirus (HAdV) positivity on PCR and bacterial lower airway infection. Genotyping, performed on a subset of HAdV positive samples, showed that HAdV species C is the major detectable species in the BAL of children with PBB and BE.

Objective-4, which included 104 children with PBB, showed that PBB typically affects very young boys and is characterised by prolonged wet cough and parent-reported wheeze. Compared to controls (n=49), children with PBB had higher rates of childcare attendance, elevated NK-cell levels in blood and increased virus detection rates (particularly adenovirus) in BAL.

Objective-5 showed that in 161 children with PBB, 106 of whom were followed for 2 years, bronchiectasis was diagnosed in 13 (8.1%). On multi-variate analysis, risk factors for BE diagnosis at 2-years were: recurrent episodes of PBB (p=0.003) and *H. influenzae* lower airway infection in the BAL (p=0.013).

Objective-6 consists of two publications. The first is a review of drug treatments of childhood cough. The second is a systematic review of the efficacy of short-course (i.e. \leq 4 weeks) antibiotics for treatment of bronchiectasis in adults and children. In this review, we concluded that there is insufficient evidence, in the current literature, to support the use of short course antibiotics in bronchiectasis and further RCTs are needed.

The limitations of the studies have been outlined in their respective chapters. The studies arising from this thesis provide new knowledge in relation to:

- The significance of wet cough, as a symptoms and sign, in relation to lower airway infection and inflammation.
- The agreement between NPA and BAL with respect to virus detection on PCR; and, the association between pathogen type (viral and/or bacterial) and neutrophilic lower airway inflammation.
- Further description of PBB, including systematic evaluation of clinical, demographic, infective and inflammatory characteristics.
- Identification of the potential role of adenovirus C in chronic suppurative lung diseases in children.
- The medium-term risk of BE diagnosis in children with PBB; and, risk factors for BE.
- Evaluation of current evidence supporting the use of short course antibiotics in adults and children with BE.

Declaration by author

This thesis is composed of my original work, and contains no material previously published or written by another person except where due reference has been made in the text. I have clearly stated the contribution by others to jointly-authored works that I have included in my thesis.

I have clearly stated the contribution of others to my thesis as a whole, including statistical assistance, survey design, data analysis, significant technical procedures, professional editorial advice, and any other original research work used or reported in my thesis. The content of my thesis is the result of work I have carried out since the commencement of my research higher degree candidature and does not include a substantial part of work that has been submitted to qualify for the award of any other degree or diploma in any university or other tertiary institution. I have clearly stated which parts of my thesis, if any, have been submitted to qualify for another award.

I acknowledge that an electronic copy of my thesis must be lodged with the University Library and, subject to the policy and procedures of The University of Queensland, the thesis be made available for research and study in accordance with the Copyright Act 1968 unless a period of embargo has been approved by the Dean of the Graduate School.

I acknowledge that copyright of all material contained in my thesis resides with the copyright holder(s) of that material. Where appropriate I have obtained copyright permission from the copyright holder to reproduce material in this thesis.

Publications during candidature

Peer-reviewed papers:

- Wurzel DF, Marchant JM, Yerkovich ST, Upham JW, Masters IB, Chang AB. Prospective Characterisation of Protracted Bacterial Bronchitis (PBB) in Children. Chest 2014 Jun 1;145(6):1271-8. doi: 10.1378/chest.13-2442.
- Wurzel DF, Mackay IM, Marchant JM, Upham JW, Yerkovich ST, Petsky H, Vaughan-Smith H, Chang AB. Adenovirus C is associated with chronic suppurative lung diseases in children. Clinical Infectious Diseases: an Official Publication of the Infectious Diseases Society of America 2014 May 9. doi:10.1093/cid/ciu225 epub ahead
- Wurzel DF, Masters IB, Isles AF. A Case For Early Bronchoscopic Airway Assessment After Disc Battery Ingestion. Pediatric Pulmonology 2014 Mar;49(3):E72-4. doi: 10.1002/ppul.22858.
- 4. **Wurzel DF,** Marchant JM, Chang AB. *Drug treatment of childhood coughs*. Australian Prescriber 2014;37:115-9
- Wurzel DF, Marchant JM, Yerkovich ST, Upham JW, Masters IB, Chang AB. Wet cough in children: Infective and inflammatory characteristics in bronchoalveolar lavage fluid. Pediatric Pulmonology 2013, 49: 561–568. doi:10.1002/ppul.22792
- Wurzel DF, Marchant JM, Clark JE, Mackay IM, Yang CYT, Sloots TP, Yerkovich ST, Upham JW, Masters IB, Baker PJ, Anderson-James S, Chang AB. Respiratory Virus Detection In Nasopharyngeal Aspirate Versus Bronchoalveolar Lavage Is Dependent On Virus Type. Journal of Clinical Virology, 2013; Dec;58(4):683-8. doi: 10.1016/j.jcv.2013.09.016.
- Wurzel DF, Marchant JM, Yerkovich ST, Upham JW, Masters IB, Chang AB. Short courses of antibiotics for children and adults with bronchiectasis. Cochrane Database Syst Rev 2011;6:CD008695.

Manuscript submission to peer-reviewed journal (JAMA Pediatrics):

1. **Wurzel DF**, Marchant JM, Yerkovich ST, Upham JW, Petsky HL, Chang AB. Protracted Bacterial Bronchitis in Children: natural history and risk factors for

bronchiectasis.

Conference Abstracts Relevant to Thesis:

- 1. European Respiratory Society International Congress, Amsterdam, Netherlands 2015: Oral Presentation "Protracted bacterial bronchitis in children: Natural history and risk factors for bronchiectasis."
- Thoracic Society of Australia and New Zealand, Gold Coast 2015: Oral presentation "Protracted Bacterial Bronchitis in Children: Natural history and risk factors for bronchiectasis at 2 years."
- Queensland Children's Medical Research Institute, Brisbane 2013: Oral presentation, "Prospective Characterisation of Protracted Bacterial Bronchitis (PBB) in Children."
- 4. American Thoracic Society, Philadelphia 2013: Poster discussion, "Further Description of Protracted Bacterial Bronchitis in Children."
- American Thoracic Society, Philadelphia 2013: Poster discussion "Nasopharyngeal Aspirate (NPA) Is The Ideal Viral Sampling Method In Children With Chronic Respiratory Symptoms."
- Thoracic Society of Australian and New Zealand, Darwin 2013: Oral presentation "Wet cough in children: Infective and Inflammatory Characteristics in Bronchoalveolar lavage fluid."
- Thoracic Society of Australian and New Zealand, Darwin 2013: Poster presentation "Protracted Bacterial Bronchitis in children: results of a large prospective cohort study."
- Asia Pacific Society Respirology, Hong Kong 2012: Oral presentation "Wet cough in children: Infective and Inflammatory Characteristics in Broncho-alveolar lavage fluid."
- Queensland Children's Medical Research Institute, Brisbane 2012: Poster presentation "Wet cough in children: Infective and Inflammatory Characteristics in Broncho-alveolar lavage fluid."
- 10. Thoracic Society of Australian and New Zealand, Canberra 2012: Poster presentation "Cochrane review Short courses of antibiotics for children and adults with bronchiectasis."
- 11. Thoracic Society of Australian and New Zealand, Perth 2011: Poster presentation

"Risk for development of bronchiectasis in patients with Protracted Bacterial Bronchitis."

Publications included in this thesis

 Wurzel DF, Marchant JM, Clarke JE, Yerkovich ST, Upham JW, Masters IB, Chang AB. Wet cough in children: Infective and inflammatory characteristics in bronchoalveolar lavage fluid. Pediatric Pulmonology 2014, 49: 561–568. doi:10.1002/ppul.22792 - incorporated into Chapter 2 as section 2.1.

Contributor	Statement of contribution
Danielle F Wurzel (Candidate)	Co-conceptualised the study (40%)
	Data analysis (90%)
	Wrote and edited the paper (85%)
	Data collection (10%)
Julie M Marchant	Co-conceptualised the study (20%)
	Critical review of the manuscript
Julia E Clarke	Critical review of the manuscript
Stephanie T Yerkovich	Data analysis (10%)
	Critical review of manuscript
John W Upham	Critical review of manuscript
Ian B Masters	Critical review of manuscript
Anne B Chang	Co-conceptualised the study (40%)
	Edited the paper (15%)
	Critical review of the manuscript

2. Wurzel DF, Marchant JM, Clark JE, Mackay IM, Wang CYT, Sloots TP, Yerkovich

ST, Upham JW, Masters IB, Baker PJ, Anderson-James S, Chang AB. *Respiratory Virus Detection In Nasopharyngeal Aspirate Versus Bronchoalveolar Lavage Is Dependent On Virus Type.* Journal of Clinical Virology, 2013 Dec;58(4):683-8. doi: 10.1016/j.jcv.2013.09.016 - incorporated into Chapter 3 as section 3.1.

Contributor	Statement of contribution
Danielle F Wurzel (Candidate)	Co-conceptualised the study (40%)
	Data analysis (80%)
	Wrote and edited the paper (80%)
	Data collection (10%)
Julie M Marchant	Co-conceptualised the study (20%)
	Data analysis (10%)
	Critical review of manuscript
Julia E Clarke	Critical review of the manuscript
Ian M Mackay	Laboratory viral processing (30%)
	Critical review of the manuscript
Claire YT Wang	Laboratory viral processing (60%)
	Critical review of the manuscript
Theo P Sloots	Laboratory viral processing (10%)
	Critical review of the manuscript
Stephanie T Yerkovich	Critical review of manuscript
John W Upham	Critical review of manuscript
Ian B Masters	Data collection (10%)
	Critical review of manuscript
Peter J Baker	Data analysis (10%)

S Anderson-James	Data collection (80%) and study coordination (80%) Critical review of manuscript
Anne B Chang	Co-conceptualised the study (40%)
	Wrote and edited the paper (20%)
	Critical review of the manuscript

3. **Wurzel DF,** Mackay IM, Marchant JM, Upham JW, Yerkovich ST, Petsky H, Vaughan-Smith H, Chang AB. *Adenovirus C is associated with chronic suppurative lung diseases in children.* Clinical Infectious Diseases: an Official Publication of the Infectious Diseases Society of America 2014 May 9. doi:10.1093/cid/ciu225 incorporated into Chapter 4 as section 4.1.

Contributor	Statement of contribution
Danielle F Wurzel (Candidate)	Co-conceptualised the study (40%)
	Data analysis (90%)
	Wrote and edited the paper (90%)
	Sample tracking/transfer adenovirus positive BAL samples (between clinical and research laboratories)
	Data collection (20%)
Ian M Mackay	Laboratory viral processing – research component (100%)
	Critical review of the manuscript
Julie M Marchant	Co-conceptualised the study (10%)
	Critical review of manuscript

John W Upham	Co-conceptualised the study (10%) Critical review of manuscript
Stephanie T Yerkovich	Data analysis (10%) Critical review of manuscript
Helen Petsky	Study coordination (80%) Data collection (80%) Critical review of the manuscript
Heidi Vaughan-Smith	Critical review of the manuscript Laboratory component (molecular testing of <i>H. influenzae</i> isolates – not included in final version)
Anne B Chang	Co-conceptualised the study (40%) Wrote and edited the paper (10%) Critical review of the manuscript

 Wurzel DF, Marchant JM, Yerkovich ST, Upham JW, Masters IB, Chang AB. Prospective Characterisation of Protracted Bacterial Bronchitis (PBB) in Children. Chest 2014 Jun 1;145(6):1271-8. doi: 10.1378/chest.13-2442 - incorporated into Chapter 5 as section 5.1.

Contributor	Statement of contribution

Danielle F Wurzel (Candidate)	Co-conceptualised the study (40%)
	Data analysis (90%)
	Wrote and edited the paper (80%)
	Data collection (10%)
Julie M Marchant	Co-conceptualised the study (20%)
	Edited the paper (10%)
	Data analysis (10%)
Stephanie T Yerkovich	Critical review of manuscript
John W Upham	Critical review of manuscript
Ian M Mackay	Laboratory viral processing – research component (100%)
	Critical review of the manuscript
Ian B Masters	Critical review of manuscript
Anne B Chang	Co-conceptualised the study (40%)
	Edited the paper (10%)
	Critical review of manuscript

5. **Wurzel DF**, Marchant JM, Yerkovich ST, Upham JW, Petsky HL, Chang AB. *Protracted Bacterial Bronchitis in Children: natural history and risk factors for bronchiectasis.* In progress - incorporated into Chapter 6 as section 6.1.

Contributor Statement of cont	ntribution
-------------------------------	------------

Danielle F Wurzel (Candidate)	Co-conceptualised the study (40%)
	Data analysis (80%)
	Wrote and edited the paper (80%)
	2-year follow-up data collection (90%) and data entry (100%)
Julie M Marchant	Co-conceptualised the study (20%)
	Critical review of manuscript
Stephanie T Yerkovich	Critical review of manuscript
	Data analysis (20%)
John W Upham	Critical review of manuscript
Helen Petsky	Study coordination (including data collection – baseline and cough diaries) 80%
	Critical review of manuscript
Anne B Chang	Co-conceptualised the study (40%)
	Edited the paper (20%)
	Critical review of the manuscript

6. **Wurzel DF**, Marchant JM, Yerkovich ST, Upham JW, Masters IB, Chang AB. *Short courses of antibiotics for children and adults with bronchiectasis.* Cochrane Database Syst Rev 2011;6:CD008695 - incorporated into Chapter 7 as section 7.1.

Contributor	Statement of contribution

Danielle F Wurzel (Candidate)	Constructed the search strategy
	Reviewed all abstracts
	Extracted the data
	Wrote and edited the manuscript (85%)
Julie M Marchant	Assisted with data interpretation
	Wrote and edited the manuscript (15%)
	Critical review of manuscript
Stephanie T Yerkovich	Critical review of manuscript
John W Upham	Critical review of manuscript
Ian B Masters	Critical review of manuscript
Anne B Chang	Reviewed all abstracts
	Critical review of manuscript

7. **Wurzel DF**, Marchant JM, Chang AB. *Drug treatment of childhood coughs*. Australian Prescriber 2014;37:115-9 - incorporated into Chapter 7 as section 7.2.

Contributor	Statement of contribution
Danielle F Wurzel (Candidate)	Wrote and edited the manuscript (80%)
Julie M Marchant	Critical review of manuscript
Anne B Chang	Wrote and edited the manuscript (20%)

Contributions by others to the thesis

In addition to the invaluable contributions of my supervisors and co-authors, as outlined above, a number of others have contributed significantly to this thesis. Dr Helen Petsky and Ms Sophie Anderson-James coordinated the overall "cough" study including recruitment of patients, data collection, data entry and follow-up of cough diaries (with the exception of the 2-year follow-up component). Ms Carol Willis was primarily responsible for data entry and database management. Associate Professor Ian Mackay and Ms Claire Wang were responsible for viral processing of respiratory samples that included PCR for the extended viral respiratory panel and molecular typing of adenovirus positive samples. Dr Heidi Vaughan-Smith and Jemima Beissmarth undertook H. influenzae characterisation.

Statement of parts of the thesis submitted to qualify for the award of another degree

None

Acknowledgements

First and foremost, I wish to acknowledge the dedication, patience and support provided to me by Anne Chang (my primary supervisor) and Julie Marchant over the duration of my PhD. Over the past 4 years, Anne and Julie have provided me with tremendous support, both intellectually and personally. Anne has been a major source of wisdom and insight for me. She has taught me many important lessons, both within and beyond the realms of research. These lessons will remain with me for many years to come. Julie has also been an invaluable mentor. Whilst also providing me with practical advice on balancing a PhD with a young family, Julie has helped to ensure that I stay on track with my PhD studies and provided me with incredible support and encouragement for which I am extremely grateful.

I would also like to acknowledge the invaluable support and mentorship of my cosupervisors. John Upham and Stephanie Yerkovich have provided significant intellectual support and guidance in my research and have brought additional perspective to the immunological aspects of my PhD. Steph has also assisted me greatly with statistical components of my papers. I also wish to acknowledge the intellectual input of Brent Masters and his careful teaching of bronchoscopic technique and critical analysis of the manuscripts and final thesis document.

I wish to acknowledge scholarship support from the Thoracic Society of Australia and New Zealand/Allen and Hanbury's, Queensland Children's Medical Research Institute and National Health and Medical Research Council. I also acknowledge the research infrastructure provided by Queensland Children's Medical Research Institute.

Lastly, I wish to acknowledge the support of my wonderful family. Firstly, my husband Ben who has given me endless encouragement, advice and support (both personal and technical!) throughout my PhD. Ben has competently juggled his own work and PhD commitments with family life. Thank you also to our beautiful children, Eli and Jacob (our two sons), who continue to be a source of immense joy and inspiration to Ben and I. Finally, thank you to my dear parents. Mum and Dad have been incredible in supporting me over the past few years and helping us to negotiate the busy process of working, studying and managing a young family. I thank them dearly for all of their help and understanding.

<u>Keywords</u>

Adenovirus, airway inflammation, bronchiectasis, children, coinfection, *Haemophilus influenzae,* nasopharyngeal aspirate, protracted bacterial bronchitis, respiratory bacteria, respiratory viruses.

Australian and New Zealand Standard Research Classifications (ANZSRC)

ANZSRC code: 111403, Paediatrics	40%
ANZSRC code: 110203, Respiratory Diseases	40%
ANZSRC code: 110804, Medical Virology	20%

Fields of Research (FoR) Classification

FoR code:	1114, Paediatrics & Reproductive Medicine	40%
FoR code:	1102, Cardiorespiratory Medicine & Haematology	40%
FoR code:	1108, Medical Microbiology	20%

TABLE OF CONTENTS

LIST OF FIGURES	XXII
LIST OF TABLES	XXIV
LIST OF ABBREVIATIONS	XXVI
CHAPTER 1: INTRODUCTION AND LITERATURE REVIEW	1
1.1 CHRONIC COUGH	1
1.1.1 DEFINITION OF CHRONIC COUGH	1
1.1.2 'WET' VERSUS 'DRY' COUGH	2
1.1.3 THE BURDEN AND COST OF CHRONIC COUGH IN CHILDREN	3
1.2 ANATOMY AND PATHOPHYSIOLOGY OF COUGH	4
1.2.1 WHAT IS COUGH?	5
1.2.2 IN-UTERO AND EX-UTERO LUNG AND CHEST WALL DEVELOPMENT	6
1.2.3 PHYSIOLOGY OF COUGH	7
1.3 CHRONIC WET COUGH	10
1.3.1 CAUSES OF CHRONIC WET COUGH	11
1.3.2 POPULATIONS AT RISK	12
1.3.3 BRONCHOSCOPY, BRONCHOALVEOLAR LAVAGE AND WET COUGH	14
1.4 PROTRACTED BACTERIAL BRONCHITIS (PBB)	14
1.4.1 LOWER AIRWAY FINDINGS IN PBB	16
1.4.2 NOMENCLATURE	23
1.5 THE IMMUNE SYSTEM AND PBB	25
1.5.1 INNATE IMMUNITY	25
1.5.2 Adaptive immunity	27
1.6 POTENTIAL SIGNIFICANCE OF PBB	28
1.7. CURRENT TREATMENT APPROACH TO CHRONIC WET COUGH AND PBB	29
1.8 WHY THIS THESIS IS ABOUT PBB AND CHRONIC WET COUGH	30
1.9 ADDRESSING CLINICAL RESEARCH GAPS	30
1.10 HYPOTHESIS AND AIMS	31
1.11 THESIS DESIGN	32
CHAPTER 2	34
2.1 INTRODUCTION TO CHAPTER 2	34

2.2 Wet cough in Children: Infective and Inflammatory Characteristics in Bronchoalveolar

Lavage Fluid	35
CHAPTER 3	44
3.1 INTRODUCTION TO CHAPTER 3	44
3.2 Respiratory Virus Detection in Nasopharyngeal aspirate versus Bronchoalveolar	· Lavage
is Dependent on Virus Type in Children with Chronic Respiratory Symptoms	45
CHAPTER 4	52
4.1 INTRODUCTION TO CHAPTER 4	52
4.2 Adenovirus Species C Is Associated With Chronic Suppurative Lung Diseases in Cl	hildren 53
CHAPTER 5	61
5.1 INTRODUCTION TO CHAPTER 5	61
5.2 Prospective Characterization of Protracted Bacterial Bronchitis in Children	62
CHAPTER 6	71
6.1 INTRODUCTION TO CHAPTER 6	71
6.2 Protracted Bacterial Bronchitis in Children: Natural History and Risk Factors for	-
Bronchiectasis at 2 years	72
CHAPTER 7	102
7.1 INTRODUCTION TO CHAPTER 7	102
7.2 Cochrane systematic review: short courses of antibiotics for children and adults	with
bronchiectasis	103
7.3 Drug treatments of childhood cough	124
CHAPTER 8	130
8.1 DISCUSSION	130
8.2 LIMITATIONS OF THIS THESIS	133
8.3 FUTURE PROJECTS AND FURTHER RESEARCH	134

8.4 CONCLUSION

REFERENCES APPENDIX 1 SUPPLEMENTARY MATERIAL FOR CHAPTERS 5 AND 6

XX

PUBLICATIONS NOT INCLUDED IN THIS THESIS

LIST OF FIGURES

1.1 Chang's proposed clinical spectrum of PBB and chronic suppurative lung disease28
2.1 Venn diagram of relative numbers of children with lower airway infection
2.2 Boxplots of medium lower airway neutrophil% and macrophage% according to infection
4.1 Phylogenetic analysis of human adenovirus (HAdV) genotypes and their assigned species
5.1 Bar graph showing primary indication for bronchoscopy in control participants65
6.1 CONSORT diagram101
7.1 Comparison 1 - Change in general medical condition
7.2 Comparison 2 - Respiratory system adverse events
7.3 Comparison 3 - Microbiological response119
7.4 Comparison 4 - Development of pseudomonas resistance (when initially sensitive).119
7.5 Analysis 1.1, Comparison 1 – Change in general medical condition, outcome 1– improvement in general medical condition
7.6 Analysis 2.1, Comparison 2 – Respiratory system adverse events, Outcome 1 – respiratory system adverse events
7.7 Analysis 2.2, Comparison 2 – Respiratory system adverse events, Outcome 2 – hospitalised for respiratory events
7.7 Analysis 3.1, Comparison 3 – Microbiological response, Outcome 1 Microbiological response

7.8 Analysis 4.1, Comparison 4 – Development of pseudomonas resistance (when initially

sensitive), Outcome 1 Pseudomonas resistance						122			
7.9 Sin	nplified pa	ediatric ch	ron	ic cough al	gorithm				.127
8.1 So	chematic	diagram	of	proposed	pathophysiological	process	uniting	PBB	and
bronch	iectasis								.132

LIST OF TABLES

2.1 Background characteristics of study participants according to cough group
2.2 Patterns of viral-bacterial co-infection and number with wet-cough
2.3 Cough categories versus lower airway infection ("dry cough" and "no cough" groups combined
2.4 Differential BAL cell count according to cough category40
3.1 Absolute number of virus detections by PCR from paired NPA/BAL samples (n=75) with diagnostic test parameters (BAL as the arbitrary gold standard)
3.2 Lower airway cellular differential versus virus status on NPA and BAL48
3.3 Multiple linear regression of mean difference in lower airway %neutrophils according to infective status on NPA and BAL
4.1 Characteristics of the study population57
4.2 Univariate logistic regression showing relationships between human adenovirus status and bacterial infection on bronchoalveolar lavage
5.1 Background characteristics of children in the cohort
5.2 Basic immune function tests
5.3 Univariate logistic regression for lower airway bacteria and viruses
6.1 Baseline demographic and clinical characteristics of study participants95
6.2 Clinical characteristics of study participants at 2-year follow-up
6.3 Uni- and multi-variate analysis of risk factors for bronchiectasis in children with PBB.97

6.4 Survival analysis - PBB to bronchiectasis (Months) (n=161)
6.5 Cellularity from BAL of children with and without bronchiectasis
7.1 Characteristics of included studies116
7.2 Characteristics of excluded studies117
7.3 Australian rates of uncomplicated acute upper respiratory tract infections in childrer and young adults
7.4 Common indications for specialist referral in chronic childhood cough

LIST OF ABBREVIATIONS

BAL	bronchoalveolar lavage
BALF	bronchoalveolar lavage fluid
BE	Bronchiectasis
BLP	beta-lactamase positive
BLN	beta-lactamase negative
CF	cystic fibrosis
CFU	colony forming units
CGRP	calcitonin gene-related peptide
hMPV	human metapneumovirus
IL-8	interleukin-8
MMP-9	matrix metalloproteinase-9
NGF	nerve growth factor
NPA	nasopharyngeal aspirate
OSA	obstructive sleep apnoea
PBB	protracted bacterial bronchitis
PCR	polymerase chain reaction
RSV	respiratory syncytial virus
TAC1	tachykinin, precursor 1
TAC3	tachykinin, precursor 2
тсс	total cell count
TLR	toll-like receptor
NTHi	Non-typeable Haemophilus influenzae

CHAPTER 1: INTRODUCTION AND LITERATURE REVIEW 1.1 CHRONIC COUGH

Chronic cough is a common reason for referral to paediatric respiratory physicians and is associated with significant impairment in quality of life for children and families [1]. Several inter-related factors influence the presence, nature and duration of cough in an individual child. The median age of children included in paediatric cough studies is 2-3 years, hence, young age is likely to be an important risk factor for chronic cough in the paediatric population [2-4]. Cough quality i.e. whether it is wet or dry, is known to be the most important single predictor of a specific cause for cough in children [5] and is a valuable tool in clinical medicine and research.

1.1.1 DEFINITION OF CHRONIC COUGH

The definitions of chronic cough in children vary. In Australian [6] and the US [7] cough guidelines, chronic cough in children is defined as cough lasting >4 weeks. Acute and protracted acute cough are defined as cough lasting <2 and 2-4 weeks, respectively [8]. In contrast, British guidelines define chronic cough in children as cough lasting >8 weeks [9]. The rationale for the Australian and US approach is several-fold. Prospective studies on young children with acute cough suggest that in approximately 90%, cough has resolved by 3 weeks [10, 11] and morbidity is reduced if intervention for chronic cough occurs within 30 days e.g. in the case the foreign body aspiration [12].

In addition to chronicity as a means of classifying cough, cough can be divided according its quality i.e. 'wet' versus 'dry'.

1.1.2 'WET' VERSUS 'DRY' COUGH

The sound produced during cough is caused by turbulent flow of expired air causing vibration of large airways and laryngeal structures [13] with shearing of secretions from within the airways. In contrast, laminar flow through smaller airways (ie. <2mm in diameter) is inaudible to the ear. The rheological properties and the cilial transportability of mucous also influence the sound of cough. [14]. The exact amount of secretions per airway calibre needed to produce wet cough in children is unknown. Animal studies on cats have shown that as little as 0.5ml of mucus instilled into the trachea is sufficient to alter cough sound [15]. Similar studies in humans have not been performed.

In young children, 'wet' (or 'moist') cough is the preferred terminology used to describe cough associated with a presumed increase in lower airway secretions. The equivalent term used in adults is 'productive cough.' The terms evolved differently as, unlike adults, young children rarely expectorate sputum. 'Dry cough' implies no, or minimal, lower airway secretions.

In the evaluation of children with chronic cough, presence of 'wet' vs 'dry' cough is a key clinical sign and symptom used by respiratory specialists. It has utility in predicting a specific cause of cough and the likelihood of response to treatment [16, 17]. The validity of wet cough has also been shown in research studies [18-20]. When cough is wet, this implies the presence of increased airway mucous, which is usually infected with bacteria [18]. There is good agreement between parents' versus clinicians' assessment of wet cough (K=0.75, 95% CI, 0.58-0.93) with good sensitivity and specificity with lower airway findings of increased secretions on bronchoscopy (0.75 and 0.79 respectively) [19].

In contrast to the validity of wet cough as a symptom and sign, parent-reported wheeze poorly reflects true wheeze and asthma [21, 22] and has a low level of agreement with doctor assessment of wheeze [23]. A possible link between wheeze and bacterial infection of the lower airways in children has been shown [24]. Further exploration of the symptom of wheeze and its co-existence with wet cough is needed and hence addressed in this thesis.

The mechanisms responsible for the development of wet cough are further discussed in Section 1.3.

1.1.3 THE BURDEN AND COST OF CHRONIC COUGH IN CHILDREN

The burden of chronic cough in children is significant. An Australian study examining the prevalence of persistent cough (defined in this study as >3 weeks) showed that 5-10% of school-aged children between 6-12 years reported its presence [25]. It is likely that prevalence is even higher in pre-school aged children. This assumption is based on the indirect observation that the majority of children with chronic cough seen in respiratory specialist clinics are young (median age 2-3 years) [2, 3].

The frequency of doctor consultations for cough is substantial and cough constitutes the most common reason for new medical visits [26]. Previous research has shown that children have multiple doctor visits for cough before referral is made to a respiratory specialist [1]. An Australian study by Marchant et al found that over a half (53%) of children had >10 medical consultations (for cough) and >80% had 5 visits in the 12 months preceding specialist evaluation [1].

Chronic cough has been shown to be associated with significant morbidity [25, 27], financial burden and reduction in quality of life for parents and children [28] [1]. Parents worry about the impact of their child's cough. An Australian study by Marchant et al [1] involving 190 families, showed that parents were most concerned about the cause of their child's cough and whether it indicated a serious illness. Other major concerns included: their child not sleeping well and cough causing damage to their child's chest. Such parental worries largely resolved after resolution of the cough with treatment [1]. Other studies have had concordant findings with parental concerns largely relating to sleep disturbance [29], discomfort and fear that the cough may cause permanent damage to their child's chest or that their child may choke or die [28].

Studies on adults with chronic report increased rates of anxiety and depression in those with cough when compared to healthy controls [30, 31]. This is not the case for parents of children with chronic cough however, whose worries and concerns resolve after the successful treatment of their child's cough [1].

In addition to the impact on quality of life for families, cough confers significant financial burden and substantial resource utilisation. The cost of medications (prescribed and over the counter), doctor visits, medical investigations and work absenteeism, although difficult to quantify, is likely to be in the order of several millions of dollars per year in Australia.

1.2 ANATOMY AND PATHOPHYSIOLOGY OF COUGH

In order to understand the pathophysiology of cough in the context of the paediatric patient, we will briefly summarise the basic principles of cough and respiratory physiology.

1.2.1 WHAT IS COUGH?

Cough is a forced expulsive manoeuvre against a closed glottis that produces a characteristic sound [32]. Simplistically, cough is a protective reflex that aims to clear the host respiratory tract from secretions or potential irritants e.g. aspirated upper airway secretions or gastric contents, inhaled particulate matter or infectious organisms [33, 34]. For example, when cough is suppressed e.g. during general anaesthesia, secretions accumulate and atelectasis and pneumonia can result. However, cough may also play a detrimental role. Forceful or persistent coughing can result in rib fracture, vomiting, syncope and reduction in quality of life [1]. Further, airway mucosal trauma may cause, or result from, repetitive coughing and lead to development of chronic cough [33]. Further, cough pathway sensitisation can perpetuate cough when its original cause has resolved. Cough also facilitates respiratory droplet spread and transmission of organisms, such as influenza or *Bordetella pertussis*, between individuals [1, 33]. Hence, effective treatment of cough requires consideration of the multitude of factors involved in its aetio-pathogenesis.

The cough reflex is a complex pathway involving cough receptors, afferent and efferent sensory limbs, cough centre and effector muscles. Cough has three distinct phases: 1) an initial deep inspiration, 2) a brief, strong expiration against a closed glottis and 3) glottic opening, with nasopharynx closure, and forceful expulsion of air through the mouth producing a characteristic sound [35].

Cough receptors are located throughout the respiratory tract and are particularly abundant in the larynx, carina and at proximal bronchial bifurcations. When cough receptors are activated, the vagus nerve provides afferent sensory pathways that lead to cough centres in the brainstem, whilst efferent pathways supply the respiratory tract muscles responsible for effecting cough [36]. A functional MRI study on adults, using capsaicin inhalation to

stimulate cough, demonstrated activation of several areas of the cerebral cortex during cough [37]. This suggests that the cerebral cortex may also influence cough production. While cough receptors are also present outside the respiratory tract in some, further discussion of cough receptors is beyond the scope of my thesis. A recent review of the physiology of cough is available [34].

1.2.2 IN-UTERO AND EX-UTERO LUNG AND CHEST WALL DEVELOPMENT

In utero lung development is complex and begins just prior to day 28 of gestation [38]. Although most of the lung structure is present at birth, development continues until the postnatal period. In utero, four processes occur in lung development to enable the lung to perform its function of gas exchange. The first involves *branching* of the airways to form conducting and respiratory airways. *Septation* then divides the airpaces to enable *alveolisation* and lung *vascularisation*. This latter process accompanies the development of bronchi [39]. Alveolar multiplication begins at around 27 weeks gestation, increasing rapidly until 2-3 years of age then continuing at a slower rate until 8-10 years, before reaching adult levels [38].

Other post-natal changes include an increase in the diameter of the large airways and a relative increase in peripheral airways size, compared to central airways, as well as an increase in lung volumes with age. Additionally, the rib cage changes to a more vertical orientation. This change in rib orientation is influenced by the transition to an upright posture. The ribs are pulled in a caudal direction, elongating the thoracic cavity [39]. During this time, rib ossification also occurs with both factors acting to improve overall lung biomechanics resulting in an increase in thoracic volume and functional residual capacity (FRC) [38]. Prior to 2 years of age, chest wall compliance relative to lung compliance is

high as rib cage ossification is incomplete and respiratory muscles are underdeveloped. By age 2, chest wall and lung compliance are similar, as seen in adults [39].

An additional factor impacting on respiratory muscle efficiency in young children is the angle of insertion of the diaphragm [40]. The horizontal (as opposed to oblique in adults) angle of insertion of the diaphragm into the chest wall results in the chest wall being drawn inward rather than downwards, making drawing air into the chest cavity less efficient in young children than in older children and adults [39]. Together with the high compliance of the infant rib cage this significantly influences susceptibility to respiratory muscle fatigue [41]. This is particularly relevant during REM sleep when the splinting effect of the intercostal muscles on the chest wall is inhibited and paradoxical inward movement of the rib cage occurs during inspiration [42]. REM sleep comprises over 50% of total sleep time in term infants, and a larger percentage in preterm infants [43]. A recent study on 22 infants showed that these paradoxical movements resolved by 3.3 years of age [44].

The characteristic biomechanics and smaller airways caliber of young children represent a respiratory disadvantage. In addition to their impact on breathing efficiency, these factors are likely to interfere with the adequacy of secretion clearance, particularly from smaller airways. This latter factor may be relevant to the pathogenesis of cough and indeed to the development of chronic suppurative lung diseases in young children, though this remains speculative.

1.2.3 PHYSIOLOGY OF COUGH

Cough is generally considered a respiratory reflex whose aim is to protect the lungs. Its presence usually signifies an underlying condition or irritant exposure [45]. The pathophysiology of cough is complex. A detailed description of the anatomy and

neurophysiology of cough is beyond the scope of this thesis and is covered comprehensively in a recent review [34]. In this chapter, I will provide a brief summary of the aspects of cough physiology relevant to the thesis topic.

Age influences several key physiological domains relevant to cough in children. A schematic overview of these age-dependent physiological domains is presented and explained in detail in a paper by Chang et al [46]. To summarise, these domains include physiology relating to 1) cough 2) the respiratory system as a whole 3) other body systems eg. the immune system and 4) general physiology [46]. Although presented as separate entitites, Chang explains that these domains are in fact overlapping and inter-related. These domains influence cough aetiology, interpretation of investigation results and response to treatment [47].

Individual factors governing the timing, nature, frequency and effectiveness of cough are protean. These include, but are not limited to, anatomic, neurologic, metabolic, immunologic and hormonal contributors. As mentioned, many of these influences are age-dependant, but gender is also important. For example, gender does not influence cough sensitivity in childhood [48], however, post-pubertal females have greater cough sensitivity than males [49], suggesting a gender or hormonal influence.

With regards to neurological influences, the cough reflex is immature and pharyngeal coordination under-developed at birth. It is weak in infants, and more so in premature infants, particularly those who have required intubation and mechanical ventilation for a prolonged period who may have abnormal development of suck and swallow rhythms [50]. Children with co-existent neurological impairment are at further risk of swallowing dysfunction.

As young children are more prone to aspiration events as compared to older children, close attention is given to the consistencies of food provided to them to ensure that these are appropriate for the child's stage of development. For example, parents are advised that children should not consume whole nuts until the age of 5 years, because of risk of aspiration.

For many reasons, young children are generally unable to expectorate sputum, even when secretions are abundant [19]. Hence, the term 'wet' or 'moist' cough is used in children, as opposed to 'productive' cough that is used in adults.

From an anatomical perspective, children differ from adults in a number of ways. Understanding these differences helps to explain disease pathogenesis. Factors such as small airway calibre, lack of collateral ventilation [51], tracheal cartilage plasticity [52] and rib orientation disadvantage airway physiology in infants and young children. Overall, these factors may lead to an increased propensity to small airways obstruction/closure and atelectasis, and, secondarily, to mucociliary clearance impairment. This is of particular relevance to conditions associated with airway mucous hyper-secretion e.g. chronic suppurative lung diseases such as PBB and bronchiectasis and aspiration lung diseases that cause additional, intra-luminal airway obstruction [47].

The anatomical and physiological differences between adults and children may contribute to certain acute paediatric conditions, such as a bronchiolitis and croup, entities that do not usually occur in adults. Similarly, adult conditions such as chronic obstructive pulmonary disease are not recognised entities in children. Protracted bacterial bronchitis is a paediatric diagnosis not described in adult literature on cough. This thesis further explores

this common paediatric condition and provides further insight into aetiological factors that contribute to its origin.

The literature review will now focus on chronic wet cough and the clinical entity of protracted bacterial bronchitis.

1.3 CHRONIC WET COUGH

As mentioned in section 1.1.2, chronic wet cough in children implies cough that is associated with increased lower airways secretions and present for >4 weeks. In this section, I will briefly review the mechanisms by which mucous over production occurs and its role in airway clearance.

Three histopathological mechanisms can potentially cause mucous overproduction leading to wet cough. These include: 1) overexpression of mucin genes leading to increased secretion; 2) hyperplasia, hypertrophy or metaplasia of secretory cells (i.e. goblet cells, submucosal glands and possibly Clara cells); 3) oversecretion of stored mucin from secretory cells [45].

From a functional perspective, cough, together with mucociliary clearance, encourages the flow of mucous in the airways in order to facilitate its removal. Based upon a physical model by Scherer and Burtz, involving turbulent airflow through a straight tube with a mucous-like fluid, cough improved mucociliary clearance of mucus down to the level of the 12th airway generation [53].
In general, voluntary cough manoeuvres are considered ineffective in the absence of mucous hypersecretion [45]. Indeed, Yeates et al showed no effect of voluntary coughing on mucous flow in healthy adults [54]. This is believed to be due to the fact that the mucous layer is substantially thinner in health (<20µm) compared to in bronchitis (>200µm) [55, 56]. Hence, cough, and more importantly wet cough, usually signifies presence of disease [5] and is the most sensitive pointer indicating a specific cause of cough [LR 26.2 (95% CI, 3.8-181.5)] [20]. Further, a study by Goyal et al showed that children whose chronic wet cough is non-responsive to 4 weeks' appropriate antibiotic therapy are more likely, than those whose cough responds to antibiotics, to have bronchiectasis on CT chest [57]. However, to date, no aetiological studies have specifically used chronic wet cough as an inclusion criteria, to determine the relative frequency of individual diagnoses.

We will briefly describe the range of conditions known to be associated with mucous hypersecretion and hence presenting with chronic wet cough. We will then discuss current knowledge on the clinical condition of protracted bacterial bronchitis.

1.3.1 CAUSES OF CHRONIC WET COUGH

The major causes of chronic cough in children differ to those in adults, hence, paediatric specific cough algorithms should be utilised. In a study involving children with chronic cough (both wet and dry), Marchant et al evaluated the utility of an adult-based algorithm for the diagnosis [2]. This study found that the 3 major causes of chronic cough in adults (asthma, gastro-oesophageal reflux disease and upper airways cough syndrome) differed from those in children (Protracted bacterial bronchitis and natural resolution accounted for 62% of cases)[2]. Hence, empirical therapies for cough, based upon adult algorithms, are inappropriate for use in children.

Chronic wet cough in children has a number of possible causes, ranging from aspiration lung disease and bronchiectasis to atypical infections e.g. tuberculosis. Clinical conditions predisposing to the development of chronic wet cough include: immune-deficiencies (acquired or congenital), cilial motility disorders and CFTR gene mutations. Large airway obstruction e.g. resulting from foreign body inhalation may present with acute onset dry cough followed several davs later bv wet cough. Clinical management pathways/algorithms, for the assessment and management of chronic cough in children, have recently been developed. These are well validated and have been shown to improve clinical outcomes [16].

1.3.2 POPULATIONS AT RISK

The median age of referral of children to respiratory specialists for the evaluation of chronic cough ranges from 2.8 - 4.5 years in Australian studies with male children being over-represented in these cohorts [5, 16]. The longer-term significance of chronic wet cough with regards to the risk of bronchiectasis has not been investigated in a prospectively designed study.

Racial and ethnic factors may be important to chronic cough. Although prevalence data is limited, rates of chronic wet cough are higher in Indigenous compared to non-Indigenous populations; for example, in a cross-sectional survey of middle-school children, chronic productive cough was reported in 26% of Alaskan native children compared to only 7% of Seattle children [58].

Bronchiectasis is an important cause of chronic wet cough and diagnoses are increasing both within Australia and the US [59, 60]. Indigenous populations living within resource-

rich countries appear to be at increased risk, compared to non-Indigenous populations. Examples of populations with high rates of bronchiectasis include: Aboriginal and Torres Strait Islander children in Australia, Maori and Pacific Islanders in New Zealand and Alaskan children in the United States [61-63].

The prevalence of non-CF bronchiectasis is estimated at 14.7 per 1,000 Aboriginal children aged <15 years [64], based on data collected from children in Central Australia [65]. This is equivalent to approximately 1 in 68 children, which is 40 times higher than that of cystic fibrosis among non-Indigenous Australian children according to data published in 2000 [66]. It is likely that this is an under-estimate of the true rate of non-CF bronchiectasis for two reasons. First, cough is often under-reported by those from remote Indigenous communities, hence bronchiectasis would be missed in a proportion [67]. Second, recent data on the point prevalence of CF births has shown a 17% reduction in the live-birth prevalence of CF since 1989, coinciding with the introduction of newborn screening in Australia [68]. Hence, the disparity in prevalence of non-CF and CF bronchiectasis is likely higher than estimated but is also projected to continue to rise. Research into risk factors for non-CF bronchiectasis in Indigenous populations suggests that recurrent episodes of lower respiratory tract infection [61], together with environmental irritants e.g. smoke exposure [69], are likely contributors.

Air pollution (both indoor and outdoor) has been shown to be associated with cough [70]. Environmental irritants, similar to tobacco smoke exposure, are believed to have additive effects in terms of contribution to respiratory disease, rather than being the sole aetiology [35]. This is supported by research studies that have shown that cough resolution is achievable in those with high exposure rates [71]. However, it is beyond the scope of my PhD work to examine the effects of airway pollution other than accounting for tobacco

smoke exposure in the cohorts of children examined. Where relevant, age and ethnicity were also taken into account in my PhD studies.

1.3.3 BRONCHOSCOPY, BRONCHOALVEOLAR LAVAGE AND WET COUGH

Examining the lower airways is a key element in furthering our understanding of chronic wet cough in children. There is limited data on the lower airways of children with chronic cough. As many of these studies have included children with PBB, these are described together in section 1.4. My PhD work (Chapter 2) involves examining the BAL of children with chronic wet cough.

In 2006, Marchant et al conducted a cohort study involving 108 children with cough >3 weeks to examine the most common aetiologies [2]. The authors described similar, characteristic lower airway findings in 40% of children presenting with chronic cough (of which 89% had chronic wet cough). These findings included: lower airway bacterial infection (with *H. influenzae*, *M. catarrhalis* and/or *S. pneumoniae*) and intense neutrophilic inflammation. Cough in these children responded to 2-weeks of appropriate antibiotic therapy, resulting in resolution of their cough [2]. This clinical entity was named protracted bacterial bronchitis or PBB and resembled the previous, poorly-defined condition known as chronic bronchitis of childhood [72]. The following section 1.4 provides an up-to-date review of current knowledge on PBB.

1.4 PROTRACTED BACTERIAL BRONCHITIS (PBB)

Protracted Bacterial Bronchitis (PBB), first defined by our research group in 2006 [2], is now a well-recognised clinical entity [6, 9, 38, 73, 74]. It is characterised by chronic wet cough (lasting >4 weeks), response to 2 weeks' of appropriate antibiotics and absence of indicators to suggest an alternative cause of cough [6]. PBB is the commonest cause of chronic wet cough in Australian children as published in single and multicentre studies on 108 [2] and 346 children with chronic cough, respectively [16, 27]. In the latter study, the 3 most common causes of primary of cough were: PBB (41.0%), asthma/reactive airways disease (15.9%), and bronchiectasis (9.0%).

PBB is also a major cause of chronic wet cough in the UK and US and has been the subject of research by several groups [75-77]. However, these studies were limited by retrospective design, absence of a control group or small number of study participants. For example, a retrospective study by Zgherea et al showed that 46% of children with chronic wet cough had positive bacterial cultures on bronchoalveolar lavage, however, no comparator/control group was included [77]. This raised the need for further research on this topic. Further, the potential role of viruses in PBB is unknown. These research gaps were addressed in this thesis.

Although bronchoscopic assessment provides definitive evidence for lower airway bacterial infection and facilitates a diagnosis of PBB, bronchoscopy is not always feasible in the clinical context, particularly in primary care settings. Hence, clinical criteria for PBB diagnosis in children were devised to obviate the need for bronchoscopy [6]. Bacterial infection of the lower airways is inferred by the observation of a clinical response to appropriate antibiotics. 'Appropriate antibiotics' refers to antimicrobials sufficiently broad-spectrum to cover common respiratory pathogens, including beta-lactamase producing *H. influenzae* strains, *M. catarrhalis* (inherently beta-lactamase positive) and *S. pneumoniae*. Amoxycillin combined with clavulanic acid is the usual first choice in PBB.

The definition of PBB, and its iterations over time, is detailed in Chapter 1.4.2.1.

1.4.1 LOWER AIRWAY FINDINGS IN PBB

Bacterial Infection

Early studies on wet cough and chronic bronchitis in adults showed that bacterial infection was a common finding. In a study in adults published in the Lancet in 1957, 25 of 61 (41%) sputum samples from adults with chronic bronchitis showed infection with respiratory bacteria. The most common bacterial species isolated were *H. influenzae* and *S. pneumoniae*, with the same organisms predominating in the sputum of adults with bronchiectasis [78].

An earlier study by Boyd, published in 1931, described 56 children with bronchiectasis and reported recurrent or chronic bronchitis to be the second most common cause of bronchiectasis in children, after bronchopneumonia [79]. Boyd showed that *S. pneumoniae and H. influenzae* were amongst the commonest organisms identified on lung aspiration or suction. The study concluded that the 2 most important factors in the development of bronchial dilatation were: 1) obstruction of a bronchus, especially if partial; and 2) infection [79].

Studies on the lower airways of children are limited as children are generally unable to expectorate sputum, and hence we rely on lower airway sampling techniques. Recent research into the lower airways of children with chronic cough has been limited to retrospective studies and small cohorts. Fitch et al published a prospective study on 23 children with chronic unexplained cough and found that those with chronic cough had significantly higher neutrophils in their BAL compared to controls and postulated that this

was due to underlying persistent airways infection [80]. Subsequent research further supported this finding. In 2007, in a retrospective chart review of 81 patients with PBB, in which bronchoscopy was performed in 19 patients, Donnelly et al observed that neutrophils were the major cell line in the BAL samples and that *H. influenzae* and *S pneumoniae* were again the most common organisms seen. The airways of children with PBB appeared oedematous and had presence of secretions [75]. These findings were consistent with Marchant et al's observations in the original studies in which PBB was first defined [2].

In 2012, Zgherea et al published a retrospective chart review of 197 children with chronic wet cough undergoing evaluation. The majority (55%) were 0-3 years of age and most (56%) had purulent bronchitis (as opposed to non-purulent bronchitis) on bronchoscopy. Positive bacterial cultures were also a major feature, being present in almost half (46%). Lower airway neutrophilia was seen, with the severity of the neutrophilia being more pronounced in those with purulent bronchitis, compared to non-purulent bronchitis [77].

Kompare et al's retrospective study of 70 children, published in the same year had concordant findings. The link between bacterial infection and neutrophilia of the lower airways was once again observed, with an additional finding of high rates of airway malacias (74% had tracheo- and/or bronchomalacia) on bronchoscopy [76].

Hence, the triad of chronic wet cough (that responds to antibiotics), bacterial infection and neutrophilic inflammation of the lower airways had been supported by studies both within Australia and internationally. There is also notable similarity in these key features of PBB and those of non-CF bronchiectasis. Lower airway bacterial infection is widely accepted as detrimental to the lung and, together with neutrophilic airway inflammation and impairment

of mucociliary clearance, constitute major components of Cole's "vicious cycle" hypothesis on the pathogenesis of bronchiectasis [81].

Growth of one or more bacterial species, at a density consistent with infection (e.g. $\geq 10^4$ cfu/ml), is present in the lower airway secretions of most children with PBB. However, this is not an absolute criteria for a diagnosis of PBB. The reasons for this are explained later in this section. The 3 bacterial species most commonly identified in the BAL of children with PBB include: *Haemophilus influenzae, Streptococcus pneumoniae* and *Moraxella catarrhalis* [2]. Research into *S. pneumoniae* serotypes isolated from the airways of children with PBB has shown that both vaccine and non-vaccine serotypes are present, suggesting that serotype replacement is unlikely to be the sole explanation for the specific strains identified [82]. The majority of *H. influenzae* isolates from the nasopharynx and lower airways of children with PBB, although prior to this thesis, this had not been studied.

The effectiveness of antibiotic therapy in the treatment of chronic wet cough has been shown in retrospective studies [2, 75]. Further, a Cochrane review published in 2005 [84] evaluated the utility of antibiotics in the treatment of chronic cough (>75% had moist cough). In the 2 included studies with a total of 140 participants, antibiotics reduced the proportion of children not cured at follow-up, with number needed to treat (NNT) of 3 (95% CI 2 to 4). More recently, a double-blinded randomised controlled trial by Marchant et al [85] showed that chronic wet cough in children responded to amoxycillin-clavulanate antibiotics, with a NNT of 3 (95%CI 2 to 4). This study also showed that requirement for further antibiotics was significantly lower in those receiving antibiotics, with NNT of 4 (95% CI 3-5) [85]. Hence, the effectiveness of antibiotics in the treatment of wet cough has been demonstrated.

It is nevertheless important to note that a proportion of children with PBB have negative bacterial cultures of their BAL [2]. There are a number of factors that may account for this. First, antibiotic exposure at the time of sampling may suppress colony counts to below the threshold to be classified as 'infection'; second, inadequate bronchoalveolar lavage technique with poor return or very localised sampling may miss the infected portion of the lung [86]; and third, sub-optimal sample handling e.g. a delay in processing, inadequate storage, plating etc may result in failure to culture an organism that may be present. This may be of particular relevance in the case of fastidious organisms e.g. nontypeable *H. influenzae* (NTHi) [87].

In earlier studies on PBB [2], children with respiratory viruses in BAL were excluded and hence, prior to this thesis, the role of viruses in PBB had not been explored. Chapter 2.2 explores the role of viral co-infections in the BAL of children with chronic wet cough.

Viral Infection

Prior to this thesis, most studies on PBB have focused on bacterial infection of the airways [2, 4, 75-77]. Respiratory viruses, such as respiratory syncytial virus (RSV), influenza viruses and rhinoviruses are commonly associated with acute cough and self-limiting respiratory illnesses in children. They are also associated with acute exacerbations of non-CF bronchiectasis in children [88], Adenovirus can persist in the airway and has been linked to chronic respiratory diseases such as bronchiectasis and bronchiolitis obliterans [89]. The role of respiratory viruses in the pathogenesis of PBB, and indeed chronic cough, are a focus of this PhD thesis. I also focus on and review the potential role of adenoviruses in this context.

Adenovirus is a common cause of upper and lower respiratory tract infections in children. Although most infections are mild or asymptomatic [90], studies reporting on lower respiratory tract infections in hospitalised infants and young children have found adenovirus to be responsible for 7-15% of cases with high rates of viral-viral co-infection (60-80%) [91-94].

The role of adenoviruses in the pathogenesis of chronic respiratory diseases, such as bronchiectasis and bronchiolitis obliterans, is well-described [89, 95-98]. A meta-analysis published by Edmond et al, examined the long-term sequelae from childhood pneumonia, and showed that adenovirus pneumonia was associated with the highest risk of sequelae (54.8% [95%CI 39.2%-70.5%]). These sequelae included: restrictive lung disease, obstructive lung disease, bronchiectasis, chronic bronchitis, asthma or a combination of conditions [99].

Adenovirus can cause chronic or persistent infection in the airways. Longitudinal studies where children were followed over a 1-year period, have shown that 81% of children with repeated isolation of adenovirus in their upper airway carry the same strain, suggesting chronic rather than repeated infection [100]. Macek and colleagues demonstrated persistence of HAdV in the lower airways of 9 of 11 children with persistent asthma who underwent repeated lower airway sampling [101]. HAdVs can cause latent infection within lymphocytes [102], tonsillar tissue [103] and the lung [104]. These findings support the hypothesis that adenovirus may play a role in the pathogenesis of certain chronic respiratory disorders [102] and this is explored further, in my PhD, in relation in PBB.

I will briefly discuss viral-bacterial co-infection in relation to respiratory tract disease.

Viral bacterial co-infection

Viral bacterial co-infection of the lung is associated with increased severity of lower respiratory tract symptoms [105, 106]. Further, numerous studies in animals [107-110] and several studies in humans support the concept of viral potentiation of bacterial infection [111-113]. There are several mechanisms by which viral infections increase the likelihood of secondary bacterial infection.

In the upper respiratory tract, viral infection predisposes to secondary bacterial infection via impairment of the epiglottal-cough reflex, interruption of the muco-ciliary function and disruption of ciliary ultrastructure with loss of cilia and ciliated cells [114]. Changes to cilial beat frequency also occur impairing the ability to remove fluid, particles and bacteria from the middle ear, sinuses and lower respiratory tract [108-110]. In addition to structural and consequently functional impairment, certain viruses e.g. adenoviruses and herpes viruses, modulate the host immune response to enable persistence of virus within the airways [115]. Prolonged faecal excretion of adenovirus 1 and 2 (for up to 515 days) has been shown in children suggesting persistence within human enteral tissues as well as the respiratory tract [116].

Enhanced adherence of bacterial pathogens is seen in virus infected or damaged cells. Viral proteins on the cell surface can act as bacterial receptors. Also, loss of airway epithelial cells exposes the basement membrane, thereby enabling more sites for bacterial binding [117-119]. Clinical studies correlating viral bacterial co-infection and bronchitis in children are lacking. Potential synergy between viruses and bacteria has been documented in other acute respiratory diseases such as pneumonia [120] and RSV

bronchiolitis [121]. We postulate that viral-bacterial co-infection and microbial synergy is also a key feature of PBB.

To date, the most convincing evidence for microbial synergy between common respiratory pathogens has arisen from animal models of otitis media. Dramatic increases in bacteria positive otitis media were found with co-infection between adenovirus and *S. pneumonia* (67%), as compared to each pathogen alone (4% and 21% respectively) [107]. Similar synergy has been shown in animal studies of otitis media between adenovirus and non-typeable *H. influenzae* [108]. This study also showed that timing of infection was important. Adenovirus prior to *H. influenzae* infection resulted in the greatest mucosal inflammation and tympanic membrane dysfunction [108]. The relevance of these findings with regards to the respiratory tract of humans warrants further study.

Neutrophilic Inflammation

Neutrophilic lower airway inflammation is an important feature of a number of chronic respiratory diseases. Lower airway neutrophilic inflammation is a common and almost universal finding in children with PBB. This has been shown in numerous studies of the lower airway secretions of children with PBB [2, 75-77]. With regards to cellularity of the lower airway, no significant findings in relation to lymphocytes or eosinophils have been seen in children with PBB [2]. A study by Zimmerman et al, examining eosinophil counts in induced sputum of children with and without asthma with post-infectious cough, showed eosinophil levels in sputum to be significantly higher in those with untreated allergic asthma [122]. Similar findings have not been seen in children with PBB. The influence of infection on lower airway cellularity and presence of neutrophilic airway inflammation requires further study and was explored in Chapter 2.

Less commonly studied markers of lower airway inflammation, including: toll-like receptor (TLR 2 and 4) mRNA and inflammatory cytokines (interleukin-8 and matrix metalloprotease-9) have been shown to be increased in BAL fluid of children with PBB [4], suggesting innate immune system activation in children with this condition, however, these studies were preliminary. A further study on the innate immune system and PBB is included in appendix 2.

1.4.2 NOMENCLATURE

'Original' versus 'revised' definition of PBB

The original definition of PBB was first published in 2006 [2] as described in chapter 1.3.3. This study included 108 young children referred to respiratory specialists for assessment of chronic cough. The aims of the study were to identify the major causes of chronic cough in children and to evaluate the utility of an adult-based cough algorithm in the diagnosis of cough in children. This study was first to identify the clinical triad of PBB. The definition of PBB has evolved over time [2].

The original definition of PBB included the following criteria: 1) prolonged wet cough lasting >3 weeks 2) presence of bacterial infection of the lower airways as seen on BAL and 3) with resolution of cough with 2 weeks of appropriate antibiotic therapy [2]. In 2010, the definition was adapted for increased feasibility, to include: 1) presence of isolated chronic wet cough lasting >4 weeks 2) resolution of cough with appropriate antibiotic treatment and 3) absence of pointers suggestive of an alternative specific cause of cough [6]. For increased feasibility and applicability to both primary and tertiary care settings, the BAL component was omitted in the revised definition of PBB [123]. This was further

supported by a randomised controlled trial on children with chronic cough (n=50) that showed significantly higher cough-resolution rates in children receiving 2-weeks of amoxycillin clavulanate antibiotics (48%), compared to placebo (16%) [85].

Recurrent versus non-recurrent PBB

The importance of differentiating recurrent versus non-recurrent PBB in an individual child is based upon the theory that those with more frequent episodes of protracted wet cough are at greater risk of bronchiectasis. This is substantiated by studies on adults, newly diagnosed with bronchiectasis, that show that most (60-80%) have a history of chronic wet cough since childhood [124, 125]. Chronic bacterial infection and concomitant neutrophilic inflammation of the lower airways, key features of PBB, are also the major features described in Cole's vicious circle hypothesis on the pathogenesis of bronchiectasis. Damage to airway epithelium (e.g. secondary to acute or chronic respiratory infection) impairs mucociliary clearance and predisposes to chronic bacterial infection thus initiating an inflammatory response that further inhibits mucociliary clearance, resulting in a vicious circle phenomenon [81].

Young children presenting with PBB frequently experience repeated episodes. A recent retrospective review of 44 children (median age 2.7 years) of which 33 had PBB, 25 (76%) had further relapse/s over an unspecified median follow-up period [126]. An earlier review of 81 children with clinical PBB found that 41 (51%) were symptom free after 2 courses of antibiotics, however 11 (13%) needed 6 or more courses of antibiotics [75]. No studies have examined the relationship between recurrent episodes of PBB and later diagnosis of bronchiectasis. Hence, this important question comprised a major aim of my PhD thesis.

In this PhD, we have defined recurrent PBB as >3 episodes of PBB per year. This cut-off was chosen, based upon the observation in clinical practice that children with more frequent episodes of PBB are less likely to "grow-out" of their wet cough episodes. It is unknown, however, whether these clinical observations were indeed important and hence, the postulated link between recurrent PBB and bronchiectasis is examined in chapter 5.

1.5 THE IMMUNE SYSTEM AND PBB

Studies on children undergoing evaluation for chronic cough have included basic immune function testing and children with major immune-deficits were excluded from earlier studies on PBB. The rationale for this approach was that children with major immune-deficiencies are closely monitored for bronchiectasis, a well-known complication of these conditions. The focus of PBB research is therefore on immune-competent, otherwise healthy children, in whom the index of suspicion for chronic suppurative lung diseases is low.

It is postulated that a defect in the innate immune system, or possibly immaturity of the adaptive immune system, predisposes to the development of PBB in children [4]. Prior to this thesis, studies examining the immune system in PBB focused on aspects of innate immunity, such as toll-like receptors and cytokines [4]. Another study explored the relationship between bacterial infection, neuropeptides and innate immune mediators in BAL fluid [127]. Both studies concluded that further studies in this area were warranted.

1.5.1 INNATE IMMUNITY

The innate immune system comprises a complex network of molecules, receptors and inflammatory mediators. In health, the innate immune system fulfills two major functions: 1) to provide rapid recognition and primary defense against invading pathogens and 2) to

communicate with (and shape) the subsequent adaptive immune system response [128]. The innate immune system differentiates between 'self' and 'non-self' to protect the host from invading organisms. In disease, any of the aforementioned processes may be altered. The innate immune system is pre-programmed and, unlike, the adaptive immune system has no specific memory. Major cell lines involved in the innate immune response include neutrophils, natural killer cells, mast cells, basophils and the complement system [128].

The important and complex role of the innate immune system, in lung health and disease, is recognised [129]. Few studies, however, have examined the innate immune system of children with PBB. In a small study, Marchant et al [4] found increased median cytokine levels (IL-8 and active MMP-9) and TLR-2 and TLR-4 mRNA expression in the BAL of children with PBB when compared to controls who were cough-free or had natural resolution of cough. TLR-2 and TLR-4 are known for their role in responding to bacterial infection. This finding was in association with the presence of positive bacterial cultures from lower airway secretions and increased total cell count and percentage neutrophils. IL-8 and MMP-9 were also increased in BAL and correlated with neutrophil levels. IL-8 is known to potentiate MMP-9. Hence, this study's findings, of significant innate immune system up-regulation in children with PBB, suggested enhanced, rather than deficient, innate immune response in these children. Further, a positive feedback loop, initiated by the presence of bacterial infection, may exist between the 2 cytokines studied as well as recruited airway neutrophils, leading to a state of chronic inflammation in the airway [4].

In another study, Grissell et al [127], examined the relationship between bacterial infection of the lower airways in children with gene expression for TLRs 2, 3, 4; chemokine receptors (CCR3, CCR5, CXCR1); neurotrophins and neurokinin genes (TAC1, TAC3,

CGRP, NGF). This study found that, in children with bacterial colonization of their lower airways, there was an airway inflammatory response, with increased BAL neutrophils, IL-8 protein, and CXCR1 expression. Further, Substance P (TAC1) and TLR4 RNA expression were reduced in the presence of bacterial infection. Overall, these findings suggested possible co-regulation of neuropeptides with the innate immune system, a relationship that may suggest impaired clearance of bacteria from the airways of children with PBB [127].

The above provides clues to the role of the host immune response in diseases such as PBB. While this importance is acknowledged, it is beyond the scope of my PhD to further examine these aspects.

1.5.2 ADAPTIVE IMMUNITY

In defending against invading pathogens, the role of the adaptive immune system is to respond to signals from the innate immune system to initiate antibody (B cell) and cellular (helper and cytotoxic T cell) mechanisms of immune defense [128]. Complex processes occur: including antigen presentation and T or B cell recognition; followed by cell priming; activation and differentiation resulting in release of antibodies into the blood or tissues (from activated B cells) or activated T cells leaving lymphoid tissue and localisation at the disease site [128]. Unlike the innate immune system, the adaptive immune system displays immunological memory, hence, if previous exposure to a pathogen type has occurred, subsequent detection and removal is enhanced. Similar to the innate immune system, aberrations in the function of the adaptive immune system can lead to impaired immunity, with resultant disease [128].

Forerunner studies on adaptive immunity in PBB have included measurement of major antibody types and subtypes (IgG and IgG subtypes, IgA, IgM and IgE) [5] [2]. The

majority of children had intact immune function, and any child with major immune deficiency was excluded from further studies. However, these previous cohorts were small (n ranging from 38-45) and incomplete immune assessments were described [4]. Given this research gap, my PhD work addressed describing basic immune function and atopy in a much larger cohort of children with PBB (Chapter 5). Availability of such data will help clinicians to better understand the complex interplay between the developing immune system and PBB.

1.6 POTENTIAL SIGNIFICANCE OF PBB

PBB and bronchiectasis share many similarities. Indeed, their clinical presentations are alike i.e. recurrent and/or protracted episodes of wet cough responsive to antibiotics, coupled with lower airway findings of neutrophilia and bacterial infection. This observation led to a hypothesis by Chang that the 2 conditions were linked and represented a clinical spectrum (see Figure 1 below) [51].



Progression of disease process

Figure 1.1: Figure reproduced with permission from A Chang.

No studies to date however, have prospectively examined the temporal relationship between PBB and BE to provide definitive support for the proposed link. Hence, this question represented the overarching aim of my thesis.

1.7. CURRENT TREATMENT APPROACH TO CHRONIC WET COUGH AND PBB

Chronic cough pathways, for the assessment and management of chronic cough in children, assist in the accurate diagnosis of chronic cough, resulting in earlier resolution of cough [16, 27].

If a child presents with chronic wet cough lasting >4 weeks, a thorough history and examination is undertaken. Specific cough pointers, suggestive of an underlying diagnosis, are sought. These include a history of developmental delay, feeding difficulties, recurrent lower respiratory tract infection, wheeze, previous treatment response etc. Examination includes assessment for chest wall deformity, clubbing, abnormality on chest auscultation and growth parameters. Chest X-ray (in all) and lung function tests (if school-aged or above) are performed. In the absence of indicators to suggest an alternative specific cause for cough (e.g. asthma, viral bronchitis) a 2-week trial of antibiotic therapy with an appropriate antibiotic (e.g. amoxycillin-clavulanate) is indicated. A randomised controlled trial published by Marchant et al in 2012 supports this approach. This showed that antibiotics (amoxycillin-clavulanate) were associated with higher cough resolution rates compared to placebo. In 50 children with wet cough lasting >3 weeks, antibiotics were associated with a cough resolution rate of 48% versus 16% in those receiving placebo (p=0.016) [85].

Follow-up is then undertaken at 2-3 weeks to ensure cough resolution and confirm a diagnosis of PBB. If a child has a partial response to antibiotics, a further 2-4 weeks' course is usually given and further aspects of the history are elicited e.g. adherence to

treatment and appropriateness of dose, with reassessment of the original diagnosis. Subsequent investigations and follow-up are based upon individual clinician practice.

1.8 WHY THIS THESIS IS ABOUT PBB AND CHRONIC WET COUGH

Chronic wet cough commonly affects children and places substantial burden on families and the healthcare system. Hence, accurate identification of the aetiology of cough is important to ensure that the most effective treatment is instituted. Using clear-cut diagnostic criteria, my supervisors were first to define the clinical entity of PBB providing the platform for further research on this important topic.

This thesis aims to further expand on current knowledge in relation to wet cough and PBB with a focus on the role of viral and bacterial infection. The medium-term significance of PBB with respect to development of bronchiectasis is also evaluated. We further describe the clinical profile of PBB and explore potential aetiological factors from the perspective of the environment, infection and the immune system. The overarching aim of this research was to delineate the natural history of PBB in children to evaluate the postulated link to bronchiectasis.

1.9 ADDRESSING CLINICAL RESEARCH GAPS

Several research groups have explored PBB in children however many questions remained. Chronic wet cough is the principle clinical feature of PBB. Further exploration of this "grass roots" symptom of PBB, and its relationship to lower airway neutrophilic inflammation and infection, warranted further exploration. Second, bacterial infection is a

key feature of PBB however the role of viral infection (and viral-bacterial co-infection) had not been investigated. Third, to date, descriptions of the clinical profile of children with PBB were based on small cohorts and further, in-depth characterisation was needed. Lastly, and arguably the most significant question, was whether PBB represent a risk for future bronchiectasis. This question warranted evaluation in a large prospective cohort study. Addressing these research gaps formed the basis of this PhD manuscript.

1.10 HYPOTHESIS AND AIMS

The over-arching aim of my thesis was to determine whether PBB is antecedent to a diagnosis of bronchiectasis, and if so, whether identifiable risk factors exist for a diagnosis of bronchiectasis. The principle hypothesis of this thesis was that PBB and bronchiectasis represent a clinical spectrum.

In order to address the aforementioned research gaps, the specific aims of the studies were to:

- Examine and contrast the lower airway findings of children with different types of cough (wet and dry) compared to no cough, with respect to infection and inflammation.
- 2. Evaluate the sensitivity, specificity and predictive values of NPA for detecting viruses in bronchoalveolar lavage (BAL); and determine the relationship between infection and lower airway neutrophilic inflammation in children with non-acute respiratory illness.
- 3. Evaluate the major human adenovirus genotypes/species, and relationships to bacterial co-infection, in children with PBB and BE.
- Delineate the clinical and laboratory characteristics of children with PBB with a specific focus on viral detection.

- 5. Determine the 2-year outcomes and risk factors for BE and recurrent PBB, in children with PBB.
- Undertake a systematic review on the efficacy of short-course antibiotic therapy for bronchiectasis in children and review specific treatment options for cough in children in general practice.

1.11 THESIS DESIGN

The studies included within this PhD thesis were designed to address key research gaps, as already outlined. Chapter 2 explores the principle clinical feature of PBB i.e. wet cough. It evaluates the significance of wet cough, compared to dry and no cough, in relation to presence of lower airway infection (bacterial and/or viral) and neutrophilic airway inflammation. This study explores the significance of viral-bacterial co-infection and provides the foundation for further studies.

Viral infections are an important cause and contributor to respiratory disease pathogenesis. The relationship between viruses in the upper, versus the lower, airways, is poorly studied. Chapter 3 evaluates the relative yield and agreement between viruses detected on NPA compared to BAL. It relates these findings to presence of neutrophilic inflammation of the lower airways.

The link between adenovirus and bronchiectasis is established. Chapter 4 investigates the adenovirus genotypes present in the lower airways of children with PBB and bronchiectasis. It also examines the potential link between adenovirus and bacterial infection of the lower airways.

Previous studies on PBB have been small and/or retrospective. Chapter 5 further describes the clinical profile of children with PBB to assist clinicians in the more accurate diagnosis of this condition. The natural history of PBB, with respect to the development of bronchiectasis, is unknown. Chapter 6 explores the medium-term outcome of children with PBB and investigates potential risk factors for a diagnosis of bronchiectasis at 2-years.

Chapter 7 discusses the current literature relating to cough treatments. Chapter 7.1 is a Cochrane review that examines the efficacy of short-course antibiotics in the treatment of bronchiectasis in children and adults. Chapter 7.2 is a review of available cough remedies in children, with relevance to general practice.

The final chapter (Chapter 8) summarises the thesis' major research findings in the context of current literature. Potential areas for further research are also explored.

CHAPTER 2

2.1 INTRODUCTION TO CHAPTER 2

In Chapter 1.3.1 we have outlined the major diagnostic entities causing chronic wet cough in children. In clinical medicine, accurate diagnosis of these conditions can be challenging. Pre-school aged children, who comprise the majority of the cohort presenting with chronic cough, are usually unable to perform lung function testing, expectorate sputum for analysis or provide a detailed and accurate history. Hence, diagnosis is frequently based upon indirect findings. Further, data specifically examining the relationship between cough nature i.e. wet versus dry cough, and lower airway infection and inflammation is lacking.

This study included a large cohort of young children presenting to a tertiary respiratory specialist clinic for investigation of chronic respiratory symptoms, the majority of whom had chronic cough. The remaining children were undergoing evaluation for indications such as persistent wheeze, stridor or noisy breathing. Children were divided according to a cough-nature, symptom-based approach into 3 groups ('wet cough,' 'dry cough' and 'no cough.') We compared rates of lower airway infection (with bacteria, viruses and viral-bacterial co-infection) in the 3 groups and explored the relationship with lower airway neutrophilic inflammation.

2.2 Wet cough in Children: Infective and Inflammatory Characteristics in Bronchoalveolar Lavage Fluid

Wet Cough in Children: Infective and Inflammatory Characteristics in Broncho-Alveolar Lavage Fluid

Danielle F. Wurzel, FRACP,^{1,2}* Julie M. Marchant, PhD,^{1,2} Julia E. Clark, FRCPCH,³ I. Brent Masters, PhD,^{1,2} Stephanie T. Yerkovich, PhD,^{4,5} John W. Upham, PhD,⁴ and Anne B. Chang, PhD^{1,2,6}

Summary. Wet cough is a common feature of many disease processes affecting children. Our aim was to examine the relationships between cough nature, lower airway infection (bacterial, viral, and viral-bacterial) and severity of neutrophilic airway inflammation. We hypothesized that viral-bacterial co-infection of the lower airway would be associated with wet cough and heightened neutrophilic airway inflammation. We prospectively recruited 232 children undergoing elective flexible bronchoscopy. Participants were grouped using a cough nature symptombased approach, into wet, dry or no cough groups. Broncho-alveolar lavage (BAL) and clinical data, including presence, nature, and duration of cough and key demographic factors, were collected. Children with wet cough (n = 143) were more likely to have lower airway bacterial infection (OR 2.6, P = 0.001), viral infection (OR 2.04, P = 0.045) and viral-bacterial co-infection (OR 2.65, P = 0.042) compared to those without wet cough. Wet cough was associated with heightened airway neutrophilia (median 19%) as compared to dry or no cough. Viral-bacterial co-infection was associated with the highest median %neutrophils (33.5%) compared to bacteria only, virus/es only and no infection (20%, 18%, and 6%, respectively, P < 0.0001). Children with wet cough had higher rates of lower airway infection with bacteria and viruses. Maximal neutrophilic airway inflammation was seen in those with viral-bacterial co-infection. Cough nature may be a useful indicator of infection and inflammation of the lower airways in children. Pediatr Pulmonol. © 2013 Wiley Periodicals, Inc.

Key words: airway inflammation; bacterial infections; bronchitis; respiratory tract; viral infections; co-infection.

Funding source: Financial Markets Foundation for Children, Number: 2010-005, Thoracic Society of Australia and New Zealand/Allen and Hanbury, Queensland Children's Medical Research Institute, National Health and Medical Research Council (NHMRC), Number: GNT1039688, NHMRC fellowships, Numbers: 545216, 511019.

¹Queensland Children's Medical Research Institute, The University of Queensland, Royal Children's Hospital, Brisbane, Queensland, Australia.

²Queensland Children's Respiratory Centre, Royal Children's Hospital, Brisbane, Queensland, Australia.

³Department of Infectious Diseases, Royal Children's Hospital, Brisbane, Queensland, Australia.

⁴School of Medicine, The University of Queensland, Brisbane, Queensland, Australia.

⁵Queensland Lung Transplant Service, The Prince Charles Hospital, Brisbane, Queensland, Australia.

⁶Child Health Division, Menzies School of Health Research, Charles Darwin University, Darwin, Northern Territory, Australia.

Conflict of interest: None.

Contributor's Statement Page:

- Danielle F. Wurzel: Dr. Wurzel was responsible for data analysis and manuscript preparation and was involved in data collection.
- Julie M. Marchant: Dr. Marchant co-conceptualized the study and assisted in data analysis and manuscript preparation.

- Julia E. Clark: Dr. Clark critically reviewed the manuscript and assisted in interpretation of findings.
- Stephanie T. Yerkovich: Dr. Yerkovich assisted in data analysis and manuscript preparation.
- John W. Upham: Professor Upham assisted in manuscript preparation.
- I. Brent Masters: Associate Professor Masters co-conceptualized the study and assisted in data collection and manuscript preparation.
- Anne B. Chang: Professor Chang conceptualized the study and was involved in all aspects of the study design, data collection, data analysis and manuscript preparation.
- All authors read and approved the final manuscript as submitted.

*Correspondence to: Danielle F. Wurzel, FRACP, Queensland Children's Medical Research Institute, Royal Children's Hospital, Brisbane, Qld 4006, Australia. E-mail: danielle.wurzel@uqconnect.edu.au

Received 14 November 2012; Accepted 11 February 2013.

DOI 10.1002/ppul.22792 Published online in Wiley Online Library (wileyonlinelibrary.com).

© 2013 Wiley Periodicals, Inc.

2 Wurzel et al.

INTRODUCTION

Chronic wet cough is the presenting feature of several different disease processes in children, ranging from viral respiratory tract infections to bronchiectasis.¹ Wet cough is an important discriminating sign in children presenting for evaluation of respiratory symptoms.² Unlike the symptom of wheeze,³ wet cough is reliably reported by parents when compared to clinician's assessment and bronchoscopic findings.⁴

The association between chronic wet cough, bacterial infection and extent of lower airway suppuration has been described in prospective^{2,5} and retrospective studies.^{6–8} However, these studies have been limited by lack of a comparative group or small cohorts. Furthermore, there are no studies to date that report on common respiratory viral pathogens associated with chronic wet cough, nor on the rates of lower airway infection in children without cough. Beyond the field of cystic fibrosis (CF), research into the role of viral–bacterial co-infection and cough in children is scarce.

Neutrophilic inflammation of the lower airways is common in children with chronic wet cough.⁹ It is also a key factor in Cole's "vicious circle" hypothesis on the pathogenesis of bronchiectasis whereby infection, inflammation, and impairment of mucociliary clearance mechanisms are responsible for airway destruction.¹⁰ The effectiveness of antimicrobials in the treatment of chronic wet cough in children has been shown in cohort studies^{7,8,11} and further supported by a recent double-blinded randomized controlled trial.¹²

In this study, 232 children were assigned to symptom-based cough groups and associations between cough presence and nature (i.e., wet or dry), lower airway infection (viral and/or bacterial), and inflammation were evaluated. We hypothesized that neutrophilic inflammation of the lower airways is greater in children with wet cough, as compared to those with dry cough or no cough, and is most intense in the presence of viral–bacterial co-infection.

ABBREVIATIONS:			
BAL	Broncho-alveolar lavage		
BALF	Broncho-alveolar lavage fluid		
BLP	Beta-lactamase positive		
BLN	Beta-lactamase negative		
CF	Cystic fibrosis		
RSV	Respiratory syncytial virus		
hMPV	Human metapneumovirus		
PCR	Polymerase chain reaction		
PBB	Protracted bacterial bronchitis		
TCC	Total cell count		

MATERIALS AND METHODS

Study Design

Any child referred by a pediatric respiratory physician for elective flexible bronchoscopy (between March 2008 and November 2011) was eligible for inclusion. The single exclusion criterion was CF, as determined by sweat chloride level and/or CF gene mutation analysis. Two hundred fifty-four children were enrolled and BAL data was available for 232 children. Cough category, that is, "wet," "dry," or "no cough" was determined by parental assessment (on the day of bronchoscopy) as previously done^{2,11} and validated.⁴

The majority of children enrolled in the study had recurrent or chronic (>4 weeks) cough (71.1%). Other common indications for bronchoscopy included: stridor or recurrent croup (7.8%), wheeze or noisy breathing (3.4%) and chronic X-ray changes or recurrent pneumonias (3.1%). Thirty-six (15.5%) participants had radiologically proven bronchiectasis. All children underwent pre-anesthetic assessment by an anesthetist who did not allow performance of bronchoscopy in children with an acute respiratory illness. The Queensland Children's Health Services Ethics Committee approved the study and all parents provided written informed consent.

A standardized clinical questionnaire was undertaken on the day of the bronchoscopy. This included current respiratory symptoms (with detailed questions regarding cough presence or absence and cough nature), past medical history and relevant demographic factors including household cigarette smoking.

Flexible bronchoscopy (Olympus; Tokyo, Japan) was performed under general anesthesia. BAL was obtained as per European Respiratory Society guidelines.¹³ Sterile normal saline solution, in three aliquots of 1 ml/kg (maximum 20 ml), was instilled into the most affected area or into the right middle lobe in patients with generalized disease. Gentle suction was applied and the sample was collected in a mucus trap. To reduce upper airway contamination, suction through the bronchoscope was avoided until the tip was wedged in the appropriate segment.

The first aliquot was used for microbiological assessment; the second and third were pooled for cellularity studies. Microbiological assessment included quantitative aerobic bacterial cultures, specific testing for mycobacterial species using inoculation of Lowenstein– Jensen media, and PCR techniques for mycoplasma.

Positive bacterial culture was defined as growth of $\geq 10^5$ colony-forming units (cfu)/ml.⁹ Bacteria considered as significant respiratory pathogens included: Streptococcus pneumoniae, Moraxella catarrhalis, Haemophilus influenzae, Pseudomonas aeruginosa, Staphylococcus aureus, Enterobacteriaceae, Mycobacteria, Aspergillus species, and Mycoplasma pneumonia.¹⁴ Detection of respiratory viruses was undertaken using polymerase chain reaction (PCR), as previously described,¹⁵ to detect respiratory syncytial virus (RSV), adenovirus, parainfluenza virus (Types 1–3), influenza and human metapneumovirus (hMPV). A total cell count (TCC) was performed and cell differential profile (minimum 400 cells counted—modified Wright's stain, Diff Quik, Lab Aids, Narrabeen, NSW, Australia) as described in prior studies.⁹

Statistical Analyses

Statistical analysis was carried out using IBM SPSS Statistics (Version 20.0, IBM Corp., Armonk, NY). Medians and inter-quartile ranges were reported as data were non-normally distributed. Mann–Whitney's *U*-test was used for two group comparisons and Kruskal–Wallis test for >2 groups comparisons of continuous variables. Pearson's chi-square test was used for categorical variables, with continuity correction used for 2×2 analyses. Children were divided into three groups for the purposes of analysis: "wet cough," "dry cough," and "no cough." For calculation of odds ratios and confidence intervals, "dry cough" and "no cough" group. For all analyses, we used two-sided tests, with *P*-values < 0.05 denoting statistical significance.

RESULTS

Participants

The median age of the 232 children (151 male, 81 female) recruited to the study was 26.5 months (IQR 15, 62). Children with dry cough were significantly older than those with wet or no cough and were more likely to be receiving corticosteroid medication. The median duration of cough overall was 28 weeks (8, 58); that in the wet cough group was 28 weeks (11.5, 56.3) and dry cough was 52 weeks (6, 108). There were no other significant differences in baseline characteristics between groups (Table 1).

Of the 232 BAL results, 115 (49.6%) had positive bacterial cultures ($\geq 10^5$ cfu/ml) and/or one or more virus detected. Of the total group, 94 (40.5%) cultured one or more bacterial species, 50 (21.6%) were PCR positive for one or more virus and 29 (12.5%) had co-infection with both bacteria and virus/es (Fig. 1).

Bacteria

The proportion of children that had positive culture $(\geq 10^5 \text{ cfu/ml})$ of one or more bacterial species varied significantly between groups; 70/143 (49%) in the wet cough group, 5/18 (27.8%) in the dry cough group, and 19/71 (26.8%) in the no cough group (P = 0.004). *H. influenzae* was the predominant pathogen in both the wet and dry cough groups, being present in 49 of 143 (34.2%) of the wet cough group and 4 of 18 (22%) of the dry cough group and 7 of 71 (9.9%) of the no cough group (P = 0.001). A large proportion of *H. influenzae* strains isolated were B-lactamase positive, with 17/49 (34.7%) in the wet cough group, only 1/7 (14.3%) in the no cough group, and none in the dry cough group (P = 0.012).

M. catarrhalis was the second most common organism in the wet cough group, present in 26/143 (18.2%), followed by *S. pneumoniae* in 22/143 (15.4%) and *S. aureus* in 8/143 (5.6%). In the dry cough group, *S. pneumoniae* was present in 1 of 18 (5.6%). In the no cough group, *M. catarrhalis* was isolated in 7 of 71 (9.9%), *S. aureus* in 3 of 71 (4.2%), *S. pneumoniae* 4 of 71 (5.6%) and *Escherichia coli* 1 of 71 (1.4%). Of the 209 children where data for *M. pneumoniae* PCR were available, 4 were positive (3 with wet cough, 1 with no cough). None in the "dry cough" group had *M. pneumoniae*.

Antibiotic use in the 24 hr prior to the bronchoscopy appeared to be associated with lower rates of bacterial infection, however, this was not statistically significant (P = 0.062). Only nine of 36 (25%) children who received antibiotics were positive for bacteria compared to 84/194 (43.3%) who did not receive antibiotics being positive (two missing data-points for antibiotic usage).

TABLE 1-	-Backaround	Characteristics	of Study	Participants	According t	o Cough Group

Characteristic	Wet cough, $n = 143$, $n (\%)$	Dry cough, $n = 18$, $n (\%)$	No cough, $n = 71$, $n (\%)$	<i>P</i> -value ^a
Sex, M:F (% male)	94:49 (66)	11:7 (61)	46:25 (65)	0.926
Median age, mo (IQR)	26 (15, 60)	66 (31, 159)	25 (11, 59)	0.001
Household tobacco smoke exposure	55 (39)	2 (12)	22 (32)	0.069
Siblings—1+	115 (80)	16 (89)	55 (78)	0.551
Indigenous status ^b	9 (6)	1 (6)	7 (10)	0.837
Bronchiectasis (radiologically proven)	24 (17)	2 (11)	10 (14)	0.759
Antibiotics prior 24 hr	28 (20)	2 (12)	6 (9)	0.092
Corticosteroid use (oral or inhaled)	n = 118, 22 (19)	n = 11, 5 (46)	n = 64, 8 (13)	0.031

^a*P*-value tests whether all three groups have the same percentage, median, or frequency. *P*-values < 0.05 considered significant and denoted in bold. ^bThree in wet cough group and one in no cough group had unknown indigenous status.



Fig. 1. Venn diagram illustrating relative number of participants demonstrating lower airway infection (bacterial, viral, and viral-bacterial) in BALF.

Viruses

One or more viruses were detected in 37/143 (25.9%) of the wet cough group, 1/18 (5.6%) of the dry cough group, and 12/71 (16.9%) of the no cough group (P = 0.074). Adenovirus was the most common virus isolated in the BAL samples across cough categories. In those with wet cough, 21 of 143 (14.7%) were positive for adenovirus, with 1/18 (5.6%) and 5/71 (7%) positive in the dry and no cough groups (P = 0.183). Parainfluenza was the second most common virus, present in 9/143 (6.3%), none and 3/71 (4.2%) in the wet, dry, and no cough groups, respectively (P = 0.478). RSV was present in 6 (4.2%), 0 and 2 (2.8%) participants in each of the wet cough, dry cough and no cough groups respectively (P = 0.616). Influenza was present uncommonly, with 2 (1.4%) in the wet cough and 2 (4.2%) in the no cough groups (P = 0.636). hMPV was isolated in a single participant in the wet cough group (0.7%).

To account for seasonal influences on viral carriage, the season in which the bronchoscopy was performed was compared to each individual virus. Parainfluenza was the only virus whose presence was associated with a particular season, being detected significantly more frequently in winter (P = 0.038).

Viral–Bacterial Co-Infection

Viral-bacterial co-infection was identified in 23/143 (16.1%) of children with wet cough, 1/18 (5.6%) with dry cough, and 5/71 (7%) with no cough (P = 0.110). The most common viral-bacterial combination was adenovirus and *H. influenzae*, with 13 of 29 (44.8%) having this combination, of which the majority, 10 of 13 (76.9%) had wet cough (Table 2). Due to small participant numbers in the no cough group, dry and no cough categories were combined into a single category in Table 3.

TABLE 2— Patterns of Viral–Bacterial Co-Infection and Number With Wet Cough

Co-infection (viral-bacterial)	No. of children	No. with wet cough
Adenovirus and H. influenzae ^a	13	10
RSV and H. influenza ^b	4	4
Parainfluenza and <i>H. influenzae</i> ^c	3	2
Adenovirus and S. pneumoniae	2	2
Parainfluenza and M. catarrhalis ^d	2	2
Adenovirus and M. catarrhalis ^e	1	0
Parainfluenza and S. pneumoniae	1	0
Parainfluenza and <i>M. pneumonia</i>	1	1
Influenza and M. catarrhalis	1	1
Influenza and S. aureus	1	1
Total	29	23

^aFour also had *M. catarrhalis*, 3 had *S. pneumonia*, 1 *S. aureus*. ^bTwo also had *S. pneumoniae*, 2 had *M. catarrhalis*. ^cOne also had *S. aureus* and *S. pneumonia*. ^dOne also had adenovirus. ^eOne also had *E. coli*.

Airway Cellularity

Cellularity data were available for 212 participants. Median TCC differed significantly across cough groups with wet cough having the highest median TCC, followed by the dry cough and no cough groups (Table 4). Wet cough also had the highest median %neutrophils, and the lowest median %macrophages. Only 30 (14.2%) participants had any eosinophils identified (range: 1–16%). Of these children, 14 (6.6%) had clinically significant eosinophilia defined as %eosinophils on BAL >2.5%.¹⁶ Nine had wet cough, of which three were receiving inhaled corticosteroids and/or beta-agonists. There was no significant difference in eosinophil count across cough groups (P = 0.759).

When grouped in accordance with infection category (Fig. 2), median %neutrophils were highest in the viralbacterial co-infection group followed by the bacterial, viral and no infection groups (P < 0.0001). Conversely, median %macrophages were highest in the no infection group and lowest in the viral-bacterial co-infection group (P = 0.002). There was no significant difference in median %lymphocytes across groups (P = 0.349). No significant differences in %eosinophils were seen across infection groups (P = 0.111), there were insufficient numbers to generate a boxplot.

Children with no cough and bacterial infection had similar airway %neutrophils and TCC to those with no cough without bacterial infection. Airway %neutrophils (median 4% "no bacteria" vs. 4.5% "bacteria present," P = 0.976) and TCCs (median $137 \times 10^6/L$ vs. $155 \times 10^6/L$, P = 0.37) showed no significant difference.

Corticosteroid use was identified in 35 patients at the time of BAL but did not influence airway cellularity (results not shown).

	Wet cough			
	Yes, n = 143, n (%)	No, n = 89, n (%)	Odds ratio (95% CI)	<i>P</i> -value ^a
No infection	59 (41.3)	58 (65.2)	0.38 (0.22, 0.65)	<0.0001
Any bacterial species	70 (49)	24 (27)	2.60 (1.47, 4.60)	0.001
H. influenzae	49 (34.3)	11 (12.4)	3.65 (1.78, 7.50)	<0.0001
H. influenzae (BLP)	17 (11.9)	1 (1.1)	11.87 (1.55, 90.86)	0.017
H. influenzae (BLN)	32 (22.4)	10 (11.2)	2.28 (1.06, 4.90)	0.035
M. catarrhalis	26 (18.2)	7 (7.9)	2.60 (1.08, 6.28)	0.033
S. pneumoniae	22 (15.4)	5 (5.6)	3.06 (1.11, 8.39)	0.030
S. aureus	8 (5.6)	3 (3.4)	1.70 (0.44, 6.58)	0.443
Any viral species	37 (25.9)	13 (14.6)	2.04 (1.02, 4.10)	0.045
Adenovirus	21 (14.7)	6 (6.7)	2.38 (0.92, 6.15)	0.073
Parainfluenza	9 (6.3)	3 (3.4)	1.93 (0.51,7.31)	0.336
RSV	2 (1.4)	6 (6.7)	0.53 (0.10, 2.66)	0.436
Influenza	2 (1.4)	2 (2.2)	0.62 (0.09,4.46)	0.632
Viral-bacterial co-infection	23 (16.1)	6 (6.7)	2.65 (1.04, 6.80)	0.042
Adenovirus and H. influenza	10 (7)	3 (3.4)	2.16 (0.58, 8.06)	0.383

TABLE 3— Cough Categories Versus Lower Airway Infection ("Dry Cough" and "No Cough" Groups Combined)

BLP, beta-lactamase positive; BLN, beta-lactamase negative.

^a*P*-values < 0.05 denoted in bold.

DISCUSSION

Findings from this large prospective cohort study indicate that rates of lower airway infection with bacteria and/or viruses are significantly higher in children with wet cough compared to those with dry or no cough. In addition, airway neutrophilia is higher in children with wet cough and most heightened in the presence of viral–bacterial co-infection. The use of a symptom-based approach increases the relevance and utility of our findings to clinical practice.

Our prior studies reporting BAL findings in children with wet cough have largely focused on protracted bacterial bronchitis (PBB), one of the most common causes of chronic wet cough in children,¹¹ findings which have subsequently been supported by retrospective studies.^{7,8,17} This study's cohort of children was recruited after our previously studied cohorts and included those with a range of clinical indications. One of the strengths of this study is the use of cough groupings, that is, "wet cough," "dry cough," and "no cough," in place of diagnostic categories. It is well known that significant symptom overlap exists between common pediatric respiratory conditions causing cough.^{7,11} Our rationale was that symptom-based categories are likely to be more objective and repeatable than diagnostic categories. Children with CF were excluded due to their exaggerated airway inflammatory responses and unique microbiota.^{18,19} We included children with non-CF bronchiectasis, the majority of whom had idiopathic bronchiectasis, as their airway cellularity and microbiota are similar to children with PBB.²⁰ Our justification for including these children was to reflect the diverse spectrum of disease presentation with wet cough and increase the value and general applicability of our findings.

Persistent lower airway neutrophilic inflammation is known to be important in the pathogenesis of many

	Wet cough, $n = 132$	Dry cough, $n = 15$	No cough, $n = 65$	<i>P</i> -value ^a
Total cell count ($\times 10^6$ /L)	200 (108, 381)	185 (101, 414)	140 (82, 268)	0.016
Neutrophil % ^b	19 (6, 53)	11 (8, 36)	4 (2,11)	<0.0001
Macrophage %	63 (39, 83)	74 (44, 88)	83 (69, 92)	<0.0001
Lymphocyte %	9 (5, 17)	8 (3, 15)	7 (5, 15)	0.612
Eosinophil % ^c	0 (0, 0)	0 (0, 0)	0 (0, 0)	

TABLE 4—Differential BAL Cell Count According to Cough Category

All values within cough categories given as median (IQR).

Median cell counts/% rounded to nearest whole number.

Due to a lack of standardization of BAL sampling and processing methods, there is currently no international consensus on normal BAL cell counts in children.¹³ We have quoted cutoffs used in our center.

^aP-values calculated using Kruskal–Wallis test to compare medians across three groups. P-values < 0.05 denoted in bold.

^bNeutrophilia defined as >6.5% neutrophils in BAL fluid.¹¹

^cEosinophilia defined as >2.5% eosinophils in BAL fluid.¹⁶



Fig. 2. A and B: Boxplots illustrating median (A) %neutrophils and (B) %macrophages according to infection presence and type on BAL; no infection, virus, bacteria and both virus and bacteria. Of the outliers (represented by asterisks/open circles in (A), most had wet cough (11 of 16). Five had elevated (>2.5%) eosinophils on BAL. Cough improved or resolved in 9 who received antibiotics. It is likely that some of these children had undetected bacterial infection (i.e., <10⁵ cfu/ml), asthma or other conditions such as aspiration, to explain these elevated neutrophil counts.

respiratory diseases.^{21,22} Neutrophilic inflammation is a common pathological feature of diseases such as COPD, bronchiectasis, and severe asthma.²² Its effect on the growing lung may be particularly relevant. It is postulated that recurrent or prolonged episodes of wet cough are antecedent to bronchiectasis,^{20,23} with neutrophilic inflammation being the presumed mechanism. Therefore, our findings of neutrophilic inflammation in symptom-based "wet cough" groupings may be clinically relevant and important.

Research on viral–bacterial co-infection in the lower airway epithelium suggests a role for increased pathogen adherence and delayed clearance,²⁴ with impairment of mucociliary clearance being an important contributor.²⁵ Clinically, viral–bacterial co-infection is thought to play an important role in the pathogenesis of respiratory diseases such as severe RSV bronchiolitis,²⁶ pneumonia,²⁷ otitis media,^{28,29} and sinusitis.^{30,31} We have shown that viral–bacterial co-infection is more common in children with (compared to those without) wet cough, and that it causes a heightened neutrophilic airway response. It is possible that viral–bacterial coinfection may also play a role in the pathogenesis of bacterial bronchitis and further research is needed.

Interestingly, in this study, the same organisms were identified in children with and without wet cough (Table 3) albeit with varying frequencies. We observed higher rates of beta-lactamase positive *H. influenzae* in the wet cough compared to the no wet cough group. This finding may be explained by antibiotic selection pressure (these children may have had greater

specifically examine for this. Lower airway infection rates in the cough-free group were also interesting. Over a quarter (27%) of cough-free children had bacteria detected with no significant increase in their median %neutrophils in BAL (median neutrophils: 4% in those without bacteria vs. 4.5% in those with bacteria). This indicates that bacteria were present without signs of active infection, that is, cough or neutrophilia. Upper airway contamination of the bronchoscope may explain this finding, however, the higher threshold of $>10^5$ cfu/ ml for diagnosis of bacterial infection used in this study should minimize such interference. Viruses were detected in 16.9% of the cough-free children despite having no acute viral lower respiratory tract infection symptoms. Together, these findings support recent literature suggesting that the lung has a characteristic microbiota, both in health and disease,^{32,33} and is unlikely to be sterile, as previously believed.

cumulative antibiotic exposure), however, we did not

The high prevalence of virus in the BAL has to be interpreted with caution in the context that viruses detected by PCR can be found in up to 33% of asymptomatic children.³⁴ Nevertheless, the role of viruses in the pathogenesis of chronic cough in children is the focus of further research by our group at present. Preliminary results show that viral infection probably plays an important role in the initiation of conditions such as PBB (results not yet published), the most common cause of chronic wet cough in children.^{11,17}

There were a number of relevant negative findings from our study. Firstly, we did not find a statistically

significant inverse relationship between antibiotic use (in the 24 hr prior to bronchoscopy) and significant bacterial growth ($>10^5$ cfu/ml on BAL). However, our study was not powered to detect this difference, and we would expect a significant association with larger participant numbers. Secondly, there was no difference in airway lymphocyte nor eosinophil counts across infection groups, even in the presence of viral infection, nor was there any association between corticosteroid use and cellularity. However, 9 of the 14 children with airway eosinophilia on BAL had wet cough; 3 of these were receiving inhaled corticosteroids and/or beta-agonist at time of bronchoscopy. Although not specifically examined in this paper, bacterial infection has recently been found to be associated with wheeze (and possibly later development of asthma). $^{35-37}$ We believe that asthma and PBB co-exist in some children, and the two conditions are not mutually exclusive.

There are some important limitations of this study. Firstly, due to the ethical issues in obtaining control BAL data in children, we used a comparator rather than a control group. This comparator group was children without cough having bronchoscopy for investigation of other respiratory symptomatology, for example, stridor. It is difficult to predict the effect of these "other respiratory symptomatology" on the infective and inflammatory properties of the lower airway but we would expect a reduction in the strength of the observed associations. This factor may explain the large number of cough-free children having bacteria and viruses detected. Secondly, we did not undertake PCR for human rhinovirus (HRV), a very common upper airway virus. The importance of HRV in the lower airways of children remains unknown. Thirdly, as bronchoscopies were performed for clinical indications only, the majority of participants had only a single BAL performed. Without multiple lower airway specimens over time, we were unable to attribute causality. Lastly, we did not look specifically at biomarkers of neutrophilic inflammation, such as IL-8.

In conclusion, our study has several important and novel findings. Foremost, wet cough is associated with higher rates of lower airway infection, with bacteria and viruses, compared to dry or no cough, with similar organisms present across cough groups. Bacteria and/or viruses are commonly present in the lower airway of children without symptoms of viral respiratory tract infection and who are cough-free. Viral–bacterial co-infection of the lower airways is associated with a significantly heightened neutrophilic airway response. Lastly, a direct correlation exists between cough nature and neutrophilic airway inflammation; children with wet cough have greater neutrophilic inflammation compared to children with dry cough.

Wet Cough in Children—Characteristics in BALF 7

Our symptom-based approach shows that cough nature in children correlates with severity of underlying airway inflammation and neutrophilic response as found in BAL, highlighting the importance of accurate characterization of cough during pediatric respiratory consultations. The finding of high rates of lower airway infection associated with wet cough may assist clinicians in making management decisions and inform future research in this area.

ACKNOWLEDGMENTS

We would like to thank all of the families that participated in this study. We would like to thank the following research assistants who enrolled participants and assisted in data collection: Sophie Anderson-James, Helen Petsky, Carol Willis, Emily Bailey, and Joanna Tuppin. We also acknowledge the contribution of Dr Peter Baker (statistician) and Kerry-Anne O'Grady (epidemiologist).

D. Wurzel received funding from Thoracic Society of Australia and New Zealand/Allen and Hanbury's Paediatric Respiratory grant-in-aid; National Health and Medical Research Council (NHMRC) post-graduate scholarship (GNT1039688) and Queensland Children's Medical Research Institute. A. Chang (545216) and J. Upham (511019) are supported by NHMRC fellowships.

REFERENCES

- 1. Shields MD, Bush A, Everard ML, McKenzie S, Primhak R, Group BTSCG. BTS guidelines: recommendations for the assessment and management of cough in children. Thorax 2008;63:iii1-iii15.
- Marchant JM, Masters IB, Taylor SM, Chang AB. Utility of signs and symptoms of chronic cough in predicting specific cause in children. Thorax 2006;61:694–698.
- Turner SW, Craig LC, Harbour PJ, Forbes SH, McNeill G, Seaton A, Devereux G, Russell G, Helms PJ. Early rattles, purrs and whistles as predictors of later wheeze. Arch Dis Child 2008;93:701–704 [Epub 2008/03/14].
- Chang AB, Gaffney JT, Eastburn MM, Faoagali J, Cox NC, Masters IB. Cough quality in children: a comparison of subjective vs. bronchoscopic findings. Respir Res 2005;6:3.
- Chang AB, Faoagali J, Cox NC, Marchant JM, Dean B, Petsky HL, Masters IB. A bronchoscopic scoring system for airway secretions–airway cellularity and microbiological validation. Pediatr Pulmonol 2006;41:887–892.
- Zgherea D, Pagala S, Mendiratta M, Marcus MG, Shelov SP, Kazachkov M. Bronchoscopic findings in children with chronic wet cough. Pediatrics 2012;129:e364–369.
- Donnelly D, Critchlow A, Everard ML. Outcomes in children treated for persistent bacterial bronchitis. Thorax 2007;62:80–84.
- Kompare M, Weinberger M. Protracted bacterial bronchitis in young children: association with Airway Malacia. J Pediatr 2012;160:88–92.
- Marchant JM, Gibson PG, Grissell TV, Timmins NL, Masters IB, Chang AB. Prospective assessment of protracted bacterial bronchitis: airway inflammation and innate immune activation. Pediatr Pulmonol 2008;43:1092–1099.

8 Wurzel et al.

- Cole PJ, Inflammation: a two-edged sword—the model of bronchiectasis. Eur J Respir Dis Suppl 1986;147:6–15 [Epub 1986/ 01/01].
- 11. Marchant JM, Masters IB, Taylor SM, Cox NC, Seymour GJ, Chang AB. Evaluation and outcome of young children with chronic cough. Chest 2006;129:1132–1141.
- Marchant JM, Masters IB, Champion AA, Petsky HL, Chang A. Randomised controlled trial of amoxycillin-clavulanate in children with chronic wet cough. Thorax 2012;67:689–693.
- de Blic J, Midulla F, Barbato A, Clement A, Dab I, Eber E, Green C, Grigg J, Kotecha S, Kurland G, Pohunek P, Ratjen F, Rossi G. Bronchoalveolar lavage in children. ERS Task Force on bronchoalveolar lavage in children. European Respiratory Society. Eur Respir J 2000;15:217–231.
- King PT, Holdsworth SR, Freezer NJ, Villanueva E, Holmes PW. Microbiologic follow-up study in adult bronchiectasis. Respir Med 2007;101:1633–1638 [Epub 2007/05/01].
- Syrmis MW, Whiley DM, Thomas M, Mackay IM, Williamson J, Siebert DJ, Nissen MD, Sloots TP. A sensitive, specific, and cost-effective multiplex reverse transcriptase-PCR assay for the detection of seven common respiratory viruses in respiratory samples. J Mol Diagn 2004;6:125–131 [Epub 2004/04/21].
- Pizzutto SJ, Grimwood K, Bauert P, Schutz K, Yerkovich ST, Upham JW, Chang A. Bronchoscopy contributes to the clinical management of Indigenous children newly diagnosed with bronchiectasis. Pediatr Pulmonol 2013;48:67–73.
- Zgherea D, Pagala S, Mendiratta M, Marcus MG, Shelov SP, Kazachkov M. Bronchoscopic findings in children with chronic wet cough. Pediatrics 2012;129:e364–e369 [Epub 2012/01/11].
- Muhlebach MS, Stewart PW, Leigh MW, Noah TL. Quantitation of inflammatory responses to bacteria in young cystic fibrosis and control patients. Am J Respir Crit Care Med 1999;160:186– 191 [Epub 1999/07/03].
- Muhlebach MS, Noah TL. Endotoxin activity and inflammatory markers in the airways of young patients with cystic fibrosis. Am J Respir Critic Care Med 2002;165:911–915 [Epub 2002/04/06].
- Chang AB, Redding GJ, Everard ML. Chronic wet cough: protracted bronchitis, chronic suppurative lung disease and bronchiectasis. Pediatr Pulmonol 2008;43:519–531.
- Gernez Y, Tirouvanziam R, Chanez P. Neutrophils in chronic inflammatory airway diseases: can we target them how? Eur Respir J 2010;35:467–469 [Epub 2010/03/02].
- Cowburn AS, Condliffe AM, Farahi N, Summers C, Chilvers ER. Advances in neutrophil biology: clinical implications. Chest 2008;134:606–612 [Epub 2008/09/10].
- Field CE. Bronchiectasis in childhood; prophylaxis, treatment and progress with a follow-up study of 202 cases of established bronchiectasis. Pediatrics 1949;4:355–372 [Epub 1949/09/01].
- de Vrankrijker AM, Wolfs TF, Ciofu O, Hoiby N, van der Ent CK, Poulsen SS, Johansen HK. Respiratory syncytial virus

infection facilitates acute colonization of Pseudomonas aeruginosa in mice. J Med Virol 2009;81:2096–2103 [Epub 2009/10/ 27].

- Sethi S. Coinfection in exacerbations of COPD: a new frontier. Chest 2006;129:223–224 [Epub 2006/02/16].
- Thorburn K, Harigopal S, Reddy V, Taylor N, van Saene HKF. High incidence of pulmonary bacterial co-infection in children with severe respiratory syncytial virus (RSV) bronchiolitis. Thorax 2006;61:611–615.
- Korppi M, Leinonen M, Makela PH, Launiala K. Mixed infection is common in children with respiratory adenovirus infection. Acta Paediatr Scand 1991;80:413–417 [Epub 1991/04/01].
- Chonmaitree T, Owen MJ, Patel JA, Hedgpeth D, Horlick D, Howie VM. Effect of viral respiratory tract infection on outcome of acute otitis media. J Pediatr 1992;120:856–862 [Epub 1992/06/01].
- Harsten G, Prellner K, Lofgren B, Kalm O. Serum antibodies against respiratory tract viruses in episodes of acute otitis media. J Laryngol Otol 1991;105:337–340 [Epub 1991/05/01].
- 30. Wald ER. Sinusitis in children. Isr J Med Sci 1994;30:403–407 [Epub 1994/05/01].
- Yonkers AJ. Sinusitis—inspecting the causes treatment. Ear Nose Throat J 1992;71:258–262 [Epub 1992/06/01].
- 32. Erb-Downward JR, Thompson DL, Han MK, Freeman CM, McCloskey L, Schmidt LA, Young VB, Toews GB, Curtis JL, Sundaram B, Martinez FJ, Huffnagle GB. Analysis of the lung microbiome in the "healthy" smoker and in COPD. PLoS ONE 2011;6:e16384 [Epub 2011/03/03].
- Hilty M, Burke C, Pedro H, Cardenas P, Bush A, Bossley C, Davies J, Ervine A, Poulter L, Pachter L, Moffatt MF, Cookson WO. Disordered microbial communities in asthmatic airways. PLoS ONE 2010;5:e8578 [Epub 2010/01/07].
- 34. Jansen RR, Wieringa J, Koekkoek SM, Visser CE, Pajkrt D, Molenkamp R, de Jong MD, Schinkel J. Frequent detection of respiratory viruses without symptoms: toward defining clinically relevant cutoff values. J of clin microbio 2011;49:2631–2636.
- 35. Bisgaard H, Hermansen MN, Bonnelykke K, Stokholm J, Baty F, Skytt NL, Aniscenko J, Kebadze T, Johnston SL. Association of bacteria and viruses with wheezy episodes in young children: prospective birth cohort study. BMJ (Clinical Research ed) 2010;341:c4978 [Epub 2010/10/06].
- 36. De Schutter I, Dreesman A, Soetens O, De Waele M, Crokaert F, Verhaegen J, Piérard D, Malfroot A. In young children, persistent wheezing is associated with bronchial bacterial infection: a retrospective analysis. BMC Pediatr 2012;12:83.
- 37. Chen J, Deng Y, Zhao J, Luo Z, Peng W, Yang J, Ren L, Wang L, Fu Z, Yang X, Liu E. The polymorphism of IL-17 G-152A was associated with childhood asthma and bacterial colonization of the hypopharynx in bronchiolitis. J Clin Immunol 2010;30: 539–545.

CHAPTER 3

3.1 INTRODUCTION TO CHAPTER 3

In Chapter 2 we demonstrated that children with wet cough have higher rates of lower airway infection with bacteria, viruses and viral-bacterial co-infection. We have also shown that children with viral-bacterial co-infection have the greatest neutrophilic lower airway inflammation, raising the possibility of microbial synergy. To date, research into chronic wet cough in children has focused on bacterial pathogens. The role of viral infections therefore needed further evaluation.

Nasopharyngeal aspirate (NPA) is a common sampling method for detection of respiratory tract viruses in children. Unlike bronchoalveolar lavage (BAL), a positive NPA does not necessarily reflect presence of virus in the lower respiratory tract. However, BAL is significantly more invasive than NPA. The level of agreement between NPA and BAL for the detection of respiratory viruses in children with non-acute respiratory illness had been poorly studied. Further, the relationship between virus presence in the airways (as detected by PCR) and lower airway neutrophilic inflammation needed further elucidation.

Specific viruses may exert greater influence on airway inflammation and may have relevance to the pathogenesis of certain chronic respiratory disorders in children. In this study, we tested for an extended panel of viruses in paired upper (NPA) and lower (BAL) respiratory tract samples from children undergoing bronchoscopy and BAL for assessment of chronic respiratory symptoms. The final diagnosis in the majority of the children in the study (64%) was PBB or bronchiectasis whilst those remaining (36%) had either no specific diagnosis or an airway abnormality. Of those with no final diagnosis, most (55%) were being investigated for chronic cough (73% had wet cough, 27% had dry cough) whilst the remainder had complaints of stridor, persistent wheeze, noisy breathing, haemoptysis or persistent auscultatory findings.

3.2 Respiratory Virus Detection in Nasopharyngeal aspirate versus Bronchoalveolar Lavage is Dependent on Virus Type in Children with Chronic Respiratory Symptoms ELSEVIER

Contents lists available at ScienceDirect

Journal of Clinical Virology

journal homepage: www.elsevier.com/locate/jcv

Respiratory virus detection in nasopharyngeal aspirate versus bronchoalveolar lavage is dependent on virus type in children with chronic respiratory symptoms

CrossMark

IROLOGY

Danielle F. Wurzel^{a,b,*}, Julie M. Marchant^{a,b}, Julia E. Clark^e, Ian M. Mackay^f, Claire Y.T. Wang^f, Theo P. Sloots^f, John W. Upham^c, Stephanie T. Yerkovich^{c,g}, I. Brent Masters^{a,b}, Peter J. Baker^d, Sophie Anderson-James^{a,b}, Anne B. Chang^{a,b,h}

^a Queensland Children's Medical Research Institute, The University of Queensland, Brisbane, Australia

^b Queensland Children's Respiratory Centre, Royal Children's Hospital, Brisbane, Australia

^c School of Medicine, The University of Queensland, Herston, Australia

^d School of Population Health, The University of Queensland, Herston, Australia

^e Department of Infectious Diseases, Royal Children's Hospital, Brisbane, Australia

^f Queensland Paediatric Infectious Diseases Laboratory, Queensland Children's Medical Research Institute, Sir Albert Sakzewski Virus Research Centre,

Children's Health Queensland Hospital and Health Service, The University of Queensland, Herston, Australia

^g Queensland Lung Transplant Service, Prince Charles Hospital, Brisbane, Australia

^h Child Health Division, Menzies School of Health Research, Darwin

ARTICLE INFO

Article history: Received 2 July 2013 Received in revised form 12 September 2013 Accepted 13 September 2013

Keywords: Bacteria Human adenovirus Human rhinovirus Neutrophilic airway inflammation Polymerase chain reaction Co-infection

ABSTRACT

Background: The comparative yield of respiratory virus detection from nasopharyngeal aspirate (NPA) versus bronchoalveolar lavage (BAL) is uncertain. Furthermore, the significance of virus detection and its relationship to lower airway neutrophilic inflammation is poorly studied.

Objectives: To evaluate the sensitivity, specificity and predictive values of NPA for detecting respiratory viruses in BAL; and to determine the relationship between viruses and lower airway neutrophilia in children with non-acute respiratory illness.

Study design: 150 paired NPA and BAL samples were obtained from 75 children aged <18 years undergoing flexible bronchoscopy for investigation of chronic respiratory symptoms. Viral studies were performed using polymerase chain reaction (PCR). Cellularity studies were performed on BALs. Diagnostic parameters of NPA compared to BAL and associations between viruses and lower airway %neutrophils were evaluated.

Results: NPA had a higher yield than BAL for detection of any respiratory virus (52 versus 38, respectively). NPA had a high sensitivity (92%) and low specificity (57%) for detecting HRV in BAL with poor kappa agreement value of 0.398 (95% CI 0.218–0.578, p < 0.001). NPA had a fair sensitivity (69%) and good specificity (90.3%) for detecting HAdV on BAL, kappa agreement was 0.561 (95% CI 0.321–0.801, p < 0.001). HAdV positivity on NPA, compared to negativity, was independently associated with heightened airway neutrophilia [mean difference (95% CI): 18 (1,35); p = 0.042].

Conclusions: NPA has a higher yield for respiratory virus detection than BAL, however its diagnostic accuracy is dependent on viral species. Adenovirus positivity is associated with significantly heightened lower airway neutrophilia in children with chronic respiratory symptoms.

Crown Copyright © 2013 Published by Elsevier B.V. All rights reserved.

1. Background

* Corresponding author at: Queensland Children's Medical Research Institute, Royal Children's Hospital, Brisbane, Queensland 4006, Australia.

Tel.: +61 7 3636 1280; fax: +61 7 3636 1958.

Abbreviations: NPA, nasopharyngeal aspirate; BAL, bronchoalveolar lavage; PCR, polymerase chain reaction; RVI, respiratory virus infection; HRV, human rhinovirus; HAdV, human adenovirus; RSV, respiratory syncytial virus; IFAV, human influenza A virus; IFBV, human influenza B virus; HPIV1–3, human para-influenza virus 1–3; hMPV, human metapneumovirus; HBoV, human bocavirus; HCoV – NL63, OC43, 229E, HKU1, human coronaviruses; WUPyV, WU polyomavirus; KIPyV, KI polyomavirus.

E-mail address: daniellewurzel@netspace.net.au (D.F. Wurzel).

Accurate identification of respiratory pathogens is important in clinical medicine and research. In bacteriology, upper airway sampling can misrepresent the lower airway microbiota [1,2]. In contrast, there is little such comparative data for virology. Nasopharyngeal aspirates (NPA) or nasal washes are regarded as the specimen of choice for detection of upper respiratory tract viruses by polymerase chain reaction (PCR) [3,4]. Upper airway sampling is relatively simple, can be performed at the bedside and is minimally invasive. In contrast, lower airway sampling

^{1386-6532/\$ –} see front matter Crown Copyright © 2013 Published by Elsevier B.V. All rights reserved. http://dx.doi.org/10.1016/j.jcv.2013.09.016
in young children is invasive, requiring a bronchoalveolar lavage (BAL).

Traditionally, BAL has been considered the 'gold-standard' for the microbiological diagnosis (viral and bacterial) of lower respiratory tract infection in children. However, as BAL requires specialist input and is often performed under anesthesia, it is reserved for patients with complex or severe disease, e.g., in the setting of intensive care; immune-compromise or in children with chronic or recurrent respiratory tract symptoms. The disparities in ease of collection of NPA versus BAL, raise two important questions. Firstly, what is the comparative yield of NPA compared to BAL for respiratory virus detection in children? Secondly, what is the significance of a positive NPA with respect to lower airway inflammation? Given the lack of studies addressing these important questions [5,6], we evaluated paired NPA and BAL specimens in 75 children.

2. Objectives

To compare the yield and diagnostic parameters of respiratory virus detection on NPA versus BAL using molecular methods, and to evaluate whether virus positivity on NPA correlates with lower airway neutrophilic inflammation and by inference, active viral pulmonary disease.

3. Study design

The Queensland Children's Health Services Ethics Committee approved the study and written informed consent was obtained from each parent or guardian. Children undergoing flexible bronchoscopy, for any clinical indication, as arranged by their treating physician, were eligible. Samples evaluated in this article were obtained from children recruited to a larger cohort study on chronic cough in children [7]. Caregivers completed a standardized clinical questionnaire on the day of bronchoscopy including current respiratory symptoms and relevant demographics. Children with symptoms of significant acute lower respiratory tract infection, e.g., high fever, tachypnea/shortness of breath, wheeze or rattly chest were deemed, by an anesthetist, to be unfit for anesthesia and excluded from the study.

Contemporaneous NPA and BAL sampling was performed under general anesthesia. NPA was collected first, using a disposable catheter connected to a mucous trap. Dry nasopharyngeal suction, via both nares, was performed, followed by suction of 2–3 ml of sterile normal saline directly into the suction catheter to rinse through remaining contents.

Bronchoscopic BAL was then performed using standardized methods as per European Respiratory Society guidelines [8]. Sterile normal saline, in three aliquots of 1 ml/kg (maximum 20 ml), was instilled into the most affected area, or right middle lobe in patients with generalized disease. To minimize upper airway contamination, suction through the bronchoscope was avoided until the tip had entered the distal airways. The first aliquot was used for microbiological testing, the second and third were pooled for cellularity studies.

Viral studies were undertaken using real-time PCR techniques, as described previously [9–13], to detect 16 respiratory virus types and subtypes. These included human rhinoviruses (HRV), human adenoviruses (HAdV), respiratory syncytial virus (RSV), human influenza A virus (IFAV), human influenza B virus (IFBV), human para-influenza virus 1–3 (HPIV1–3), human metapneumovirus (hMPV), human bocavirus (HBoV), human coronaviruses (HCoV – NL63, OC43, 229E, HKU1), WU polyomavirus (WUPyV) and KI polyomavirus (KIPyV). BAL specimens were refrigerated immediately and processed within 24 h. NPA specimens were stored at -70 °C prior to processing.

We defined positive bacterial culture as growth of $\geq 10^5$ colonyforming units (cfu)/ml [10] of any of the following pathogens on BAL: *Streptococcus pneumoniae*, *Moraxella catarrhalis*, *Haemophilus influenzae*, *Pseudomonas aeruginosa*, *Staphylococcus aureus* or Enterobacteriaceae. Total cell count and differential profile (minimum 400 cells counted) were performed on BAL as described previously [14].

3.1. Statistical analyses

Statistical analyses were carried out using IBM SPSS (v20, San Francisco, USA). Medians and inter-quartile ranges were reported as data were non-normally distributed. Kappa analyses were used to assess levels of agreement between sampling techniques. Comparisons of categorical variables were performed using Pearson's χ^2 test or Fisher exact test if any category had an expected value of <5. Mann–Whitney *U* test was used for two group comparisons and Kruskal–Wallis test for >2 groups comparisons of continuous variables. Multiple linear regression was employed to adjust for virus co-detection and bacterial infection. A two-tailed *p*-value <0.05 was considered statistically significant. Sensitivity, specificity, positive and negative predictive values and likelihood ratios were manually calculated.

4. Results

Between February 2010 and May 2012, paired NPA and BAL samples were obtained from 75 children (median age = 33 months, IQR 16, 69; male n = 50, 66.7%). One-third of children (n = 25, 33.3%) were exposed to household cigarette smoke. Chronic cough (\geq 4 weeks) was reported in 45 participants (60%) at time of bronchoscopy. The major diagnoses within the cohort included: protracted bacterial bronchitis (n = 31, 41%) and bronchiectasis (n = 17, 23%); the remaining participants had either no specific diagnosis (n = 22, 29%) or a congenital airway abnormality (n = 5, 7%). The most common viruses detected on NPA were HRV (n = 44) and HAdV (n = 15). In regard to seasonality, HAdV positivity on BAL was more likely to occur in spring or summer than winter or autumn (p = 0.004) using chi-square analysis. HRV detections followed a bimodal distribution with peak detections occurring in spring and autumn.

4.1. Comparison of yield on paired NPA and BAL samples

Of the 75 paired NPA and BAL samples, 52 (69.3%) children had one or more virus detected from NPA and 38 (50.7%) had one or more virus from BAL. When considering detection of any virus, 32 (42.7%) NPA-BAL pairs were concordant PCR-positive, 17 (22.7%) pairs were concordant PCR-negative and 26 (34.7%) pairs had discordant PCR results. Of the discordant pairs, most (n = 20, 76.9%) were PCR positive on NPA but negative on BAL. Detection rates varied among the different viruses (Table 1). Most marked was the difference between NPA and BAL for HRV. Using positive PCR on BAL as an arbitrary gold standard, we calculated sensitivity, specificity, positive and negative predictive values and likelihood ratios (Table 1). HRV and HAdV were the most common viruses detected from NPA and BAL. Levels of agreement between NPA and BAL varied among the different viruses. The agreement between NPA and BAL for HRV was poor, kappa = 0.398 (95% CI 0.218, 0.578, p < 0.001)but good for HAdV at 0.561 (95%CI 0.321, 0.801, *p* < 0.001).

4.2. Relationship between virus positivity and BAL neutrophilia

The median lower airway % neutrophils were significantly higher in children who were NPA-positive for any virus compared to

Table 1

Absolute number of virus detections by PCR from paired NPA/BAL samples (n = 75) with diagnostic test parameters (BAL as the arbitrary gold-standard).

NPA/BAL	HRV	HAdV	Other ^a	HBoV	HCoV	WUPyV/KIPyV	Total ^b
NPA+	44	15	5	7	4 ^c	4	52
BAL+	24	13	4	4	3 ^c	1	38
NPA+/BAL-	22	6	1 ^d	5	3	4	20
NPA-/BAL+	2	4	0	2	2	1	6
NPA+/BAL+	22	9	4	2	1	0	32
NPA-/BAL-	29	56	70	66	69	70	17
Sensitivity	92%	69%	-	50%	33%	-	84%
Specificity	57%	90%	99%	93%	96%	-	46%
PPV	0.5	0.6	0.80	0.29	0.25	-	0.62
NPV	0.94	0.93	-	0.97	0.97	-	0.74
LR+	2.12	7.15	-	7.10	7.99	-	1.56
LR-	0.19	0.34	-	0.54	0.70	-	0.35
Карра	0.40	0.56	0.88	0.32	0.25	0.02	0.30

Abbreviations: PPV, positive predictive value; NPV, negative predictive value; LR+, positive likelihood ratio; LR-, negative likelihood ratio.

^a RSV. IFA/B. HPIV1-3. hMPV.

^b Total no. of children with one or more viruses detected.

^c HCoV-OC43.

^d HPIV-1.

Table 2

Lower airway cellular differential versus virus status on NPA and BAL.

Lower airway cellular profile	NPA HRV+ $(n = 44)$	NPA HRV– $(n=31)$	p-Value	NPA AdV+ $(n = 15)$	NPA AdV $-(n=60)$	p-Value
TCCs (×10 ⁶ /L)	215(106,285)	240 (94, 360)	0.894	260(168,473)	200 (90, 290)	0.132
Neutr%	16(6,55)	13(4,32)	0.514	51(15,68)	12(5,36)	0.017
Macr%	60(37,82)	71 (29, 87)	0.605	46(23,74)	71(41,84)	0.075
Lymph%	6(4,13)	6(3,15)	0.812	5(2,11)	6(4,15)	0.183
Eosin%	0(0,0)	0(0,0)	0.262	0(0,2)	0(0,0)	0.314
Lower airway cellular profile	BAL HRV+ $(n=24)$	BAL HRV– $(n=51)$	<i>p</i> -Value	BAL AdV+ $(n = 13)$	BAL AdV $-(n=62)$	p-Value
TCCs (×10 ⁶ /L)	210(108,413)	230(95,350)	0.645	240(112,485)	215(94,285)	0.534
Neutr%	16(9,63)	15(5,40)	0.457	44(14,70)	12(5,37)	0.077
Macr%	65(27,82)	68(39,84)	0.544	54(25,83)	70(38,84)	0.497
Lymph%	8(5,20)	5(3,14)	0.318	3(2,7)	7(5,15)	0.017
Eosin%	0(0,2)	0(0,0)	0.162	0(0,1)	0(0,0)	0.825

Abbreviations: Neutr%, percentage neutrophils; Macro%, percentage macrophages; Lymph%, percentage lymphocytes; Eosin%, percentage eosinophils. Cell counts expressed as median (interquartile range) and rounded to nearest whole number.

NPA-negative children (19 [IQR 7, 65] vs. 11 [IQR 3, 21.5]; p = 0.045). Total cell counts (TCCs), %macrophages, %lymphocytes and %eosinophils were similar across groups (data not shown). When evaluating the two most frequent viruses, HRV and HAdV (Table 2), presence of HAdV in the NPA was significantly associated with increased BAL neutrophilia (p = 0.017) whereas HRV status on NPA showed no association with %neutrophils (p = 0.514).

Similar to the NPA findings, the presence of any virus in the BAL was associated with higher median %neutrophils compared to the virus-negative state (23 [IQR 10, 67] vs. 11 [5,28], respectively; p = 0.04). Median lower airway TCC was higher in BAL virus-positive (235 × 10⁶/L [IQR 115, 448]) versus virus-negative (160 × 10⁶/L [58, 255]) groups (p = 0.068). No differences in lymphocyte counts was observed when considering total viruses, but for individual viruses, %lymphocytes were significantly lower in HAdV BAL positive compared to HAdV BAL negative participants (p = 0.017). Children who were HAdV positive on BAL had higher %neutrophils than those who were HAdV negative on BAL (p = 0.077). In comparison, %neutrophils were similar across HRV positive and negative groups on BAL (p = 0.457).

4.3. Regression analysis adjusting for virus co-detection and bacterial infection

On multi-variable analysis, HRV in either NPA or BAL showed no association with lower airway %neutrophils. In contrast, HAdV in NPA and/or BAL was significantly associated with increased %neutrophils on BAL (Table 3). Bacterial infection was detected in over half (32; 61.5%) of NPA virus positive participants with much lower rates (6; 26.1%) in virus negative participants (p = 0.005). Rates of bacterial infection in BAL virus positive (22; 57.9%) children were also greater than BAL virus negative children (16; 43.2%); however, this difference was not significant (p = 0.204). The predominant bacterial organisms were: *H. influenzae, M. catarrhalis* and *S. pneumoniae*. Presence of bacterial infection as seen on BAL had the strongest independent influence on lower airway neutrophil counts (beta = 0.256 and 0.311 for NPA and BAL, respectively) as shown in Table 3.

5. Discussion

In this study, we systematically compared upper airway (NPA) to lower airways (BAL) sampling with regards to diagnostic yield for respiratory viruses in children with non-acute respiratory illness. We found that concordance of NPA with BAL is largely dependent on the virus being investigated. NPA has good concordance with BAL for detection of HAdV, but poor concordance for HRV. In comparison to BAL, sensitivity, specificity, PPV and NPV of the most frequently detected viruses on NPA differed according to the virus being investigated. Specificity was high for all viruses, except for HRV. The reverse was observed with regards to sensitivity. Lastly, presence of HAdV on NPA and/or BAL was associated with significantly heightened %neutrophils in the lower airway.

To our knowledge, three major studies have compared upper and lower airway specimens for respiratory virus detection. All were limited by unpaired specimens or focused on specific Multiple linear regression of mean difference in lower airway %neutrophils according to infective status on NPA and BAL.

	Unadjusted ^a mean difference in neutr% (IQR)	<i>p</i> -Value	Adjusted ^b mean difference in neutr% (IQR)	<i>p</i> -Value ^c	Beta ^d
NPA					
HAdV	20(3,37)	0.020	18(1,35)	0.042	0.245
HRV	5(-9,19)	0.486	-2(-16,12)	0.782	-0.033
Bacteria $\ge 10^5$ cfu/ml (on BAL)	16(3,30)	0.016	15(1,28)	0.035	0.256
BAL					
HAdV	15(-3,32)	0.097	18(0.3, 35)	0.046	0.235
HRV	4(-10,19)	0.550	0(-14,15)	0.950	0.007
$Bacteria \geq 10^5 \ cfu/ml \ (on \ BAL)$	16(3,30)	0.016	18(4,31)	0.01	0.311

^a Unadjusted mean difference in %neutrophils between HAdV, HRV and bacteria positive and negative groups calculated using independent samples t-test.

^b Adjusted mean difference in %neutrophils using multiple linear regression (adjusted for viral co-detection and bacterial infection).

^c *p*-Values remain significant after logarithmic transformation of %neutrophils therefore mean differences reported.

^d Standardized coefficient indicates independent influence of variable on %neutrophils.

diseases or viruses. The first compared NPA to BAL in lung transplant recipients using unpaired, non-contemporaneous NPA and BAL specimens obtained from 72 and 56 participants for NPA and BAL, respectively. A higher yield of hMPV was seen in NPA compared to BAL, with comparable rates of IFAV, HPIV, RSV, HCoV and HRV [6]. Diagnostic accuracy of NPA for BAL was not assessed in this study [6]. The second study used paired specimens, obtained from 21 participants on 92 occasions, and compared induced sputum to NPA for detection of respiratory viruses in children with cystic fibrosis. HRV was the most common virus identified and was present in the same frequency (21.7%) in NPA and sputum. Concordance rates were higher than in the present study (87% for HRV and 92% for other viruses), with higher sensitivity and specificity (70% and 91.7%, respectively) [15]. The third study compared paired nasal wash/tracheal suction specimens to BAL in terms of RSV detection in 6 participants, and found BAL was superior for detection of RSV using antigen detection and culture methods (insensitive methods when compared to PCR) [5].

Studies directly comparing NPA to BAL have focused on atypical pneumonia. The first study evaluated *Pneumocystis jiroveci* pneumonia and the second compared detection methods for mycoplasma pneumonia. Both studies found high concordance between NPA and BAL. However, in contrast to our findings, in both studies, BAL had a higher yield of positive detections than NPA. Interestingly, in the study on mycoplasma pneumonia, BAL positivity was associated with increased lower airway %neutrophils, however, NPA positivity was not. These differing findings may reflect the different modes of transmission and mechanisms of disease progression between viral and bacterial infections of the respiratory tract.

Our study showed that NPA/BAL discordance was most marked for HRV, where 22 (29.3%) children were NPA-positive but BALnegative. In contrast, HAdV was more likely to also be found in the lower airway, with only 2 NPA-positive/BAL-negative children. Previous studies have found HRV to have greater propensity to infect the nasopharynx and proximal lower airways than the distal airways or alveoli [16]. This may be related to HRV's ability to replicate optimally at cooler temperatures of the upper airway [17], although this finding is inconsistent across HRV types [18]. In vivo studies demonstrate that HRV induces minimal epithelial cytotoxicity with only a small subset of nasal and bronchial mucosal cells becoming infected [16,19,20]. These factors may explain why HRV is an uncommon cause of pneumonia, except in the immune-compromised host [21]. Whilst HRV and HAdV can both cause uncomplicated upper respiratory tract infection [22], disease manifestation differs at the severe end of the spectrum. HRV has been shown to play an important role in asthma exacerbations [23]; HAdV has been implicated in the pathogenesis of obliterative bronchiolitis and bronchiectasis [24]. Thus, differences

between HAdV and HRV in their mechanisms of propagation and respiratory disease pathogenesis are likely to explain our observations.

Our cross-sectional findings of elevated rates of HRV (59%) in children who were essentially free of acute RVI symptoms is higher than prior studies showing asymptomatic HRV nasopharyngeal detection ranging from 12% to 33% [25–27]. HRV is frequently identified in patients with viral:bacterial co-infection [28], a fact that may have relevance to our cohort of children with high rates of bacterial infection and chronic cough.

To further understand the implications of our findings in relation to lower airway neutrophilic inflammation and virus detection, the distinction between positive virus detection and active viral disease merits discussion. A PCR-positive NPA in a child without overt symptoms of acute RVI is often believed to represent viral genome shedding from a resolving infection [29] or preceding a new symptomatic episode. However, certain HAdV types can cause prolonged host shedding [30] with intermittent viral excretion over months to years [31,32]. In contrast, HRVs usually do not persist. Garnett et al. postulate that "smoldering HAdV at the site of lung inflammation" may contribute to the pathogenesis of lung diseases such as chronic obstructive pulmonary disease (COPD) in adults [33]. Our finding of low BAL %lymphocytes, together with neutrophilia, in children who were HAdV DNA positive on BAL, supports the notion that active lower airway HAdV replication was likely to be occurring (as opposed to latent HAdV infection of lymphocytes). Whether such persistence of HAdV viral DNA within the respiratory tract of children is benign or contributes to the development of chronic lung disease/s is unknown.

The major strengths of this study are its prospective design, the use of highly sensitive PCR techniques to detect a wide range of respiratory viruses, contemporaneous detection of lower airway bacteria and inclusion of BAL cellularity findings. However, there are a number of limitations. Firstly, this study involved children with chronic respiratory symptoms, predominantly cough. Hence, these findings may not be readily extrapolated to healthy children or to those with acute lower respiratory tract illness. This is particularly pertinent to influenza and RSV, as there were very few children in whom these viruses were detected. Secondly, we used neutrophilic lower airway inflammation as a surrogate marker for active pulmonary disease, as used by other authors [34,35]. We did not, however, test for additional indirect markers of viral infection such as Toll-like receptor 3 (TLR3), interferon gamma induced protein 10 (IP-10) or virus-specific antibodies. Thirdly, we only obtained BAL samples at a single time-point, precluding the ability to make inferences regarding causality. Lastly, although we were fastidious in our BAL technique, contamination of the bronchoscope in its route through the upper airway is possible. The significant number of BAL positive/NPA negative cases, however, negates the

argument that contamination could account for all lower airway detections.

In conclusion, we have shown that while upper airway sampling is the most practical and frequently used method of assessing virus presence, it does not necessarily reflect virus in the lower airways in children with non-acute respiratory illness. However, certain viruses (e.g., HAdV), when detected on NPA, may provide clues to the existence of lower airway neutrophilic inflammation and further research is needed. Studies addressing acute viral respiratory tract illnesses in children are required to ascertain the applicability of our findings to the acute setting.

Author contributions

Dr. Wurzel co-conceptualized the study, was responsible for data analysis and manuscript preparation and was involved in data collection.

Dr. Marchant co-conceptualized the study and assisted in data analysis and manuscript preparation.

Dr. Clark provided intellectual input into the study and critically reviewed the manuscript.

Dr. Mackay and Ms. Wang were responsible for viral processing of specimens and critically reviewed the manuscript.

Prof. Sloots was responsible for viral processing of specimens and critically reviewed the manuscript.

Prof. Upham and Dr. Yerkovich assisted in preparation and critical review of the manuscript.

Assoc. Prof. Masters assisted in data collection and critical review of the manuscript.

Dr. Baker assisted in statistical aspects of the study.

Ms. Anderson-James assisted with data collection and study coordination.

Prof. Chang conceptualized the study and was involved in all aspects of the study.

All authors read and approved the final manuscript as submitted.

Funding

This work was supported by Financial Markets Foundation for Children (project grant 2010-005); Royal Children's Hospital Foundation; Thoracic Society of Australia and New Zealand/Allen and Hanbury's and Queensland Children's Medical Research Institute [DW]; National Health and Medical Research Council [545216 to AC, 511019 to JU and GNT1039688 to DW].

Competing interests

IM has received payment for consultancy services from Firebrick pharma. JU has served on the advisory board for Novartis and has previously received payments for lectures or served on speakers' bureaus for AstraZeneca, GSK, Novartis and Boehringer.

All other authors report no potential conflicts.

Ethical approval

Ethical approval was given by Queensland Children's Health Services (RCH) Human Research Ethics Committee (HREC). Reference number: HREC/03/QRCH/17.

Acknowledgements

We wish to thank the families that participated in this study. We also thank Helen Petsky, Carol Willis, Stacey Spencer and Joanna Tuppin for their assistance with patient recruitment, data entry and study coordination.

References

- [1] Goddard AF, Staudinger BJ, Dowd SE, Joshi-Datar A, Wolcott RD, Aitken ML, et al. Direct sampling of cystic fibrosis lungs indicates that DNAbased analyses of upper-airway specimens can misrepresent lung microbiota. Proceedings of the National Academy of Sciences of the United States of America 2012;109:13769–74.
- [2] Armstrong DS, Grimwood K, Carlin JB, Carzino R, Olinsky A, Phelan PD. Bronchoalveolar lavage or oropharyngeal cultures to identify lower respiratory pathogens in infants with cystic fibrosis. Pediatric Pulmonology 1996;21:267–75.
- [3] Hyypia T, Puhakka T, Ruuskanen O, Makela M, Arola A, Arstila P. Molecular diagnosis of human rhinovirus infections: comparison with virus isolation. Journal of Clinical Microbiology 1998;36:2081–3.
- [4] Ngaosuwankul N, Noisumdaeng P, Komolsiri P, Pooruk P, Chokephaibulkit K, Chotpitayasunondh T, et al. Influenza A viral loads in respiratory samples collected from patients infected with pandemic H1N1, seasonal H1N1 and H3N2 viruses. Virology Journal 2010;7:75.
- [5] Englund JA, Piedra PA, Jewell A, Patel K, Baxter BB, Whimbey E. Rapid diagnosis of respiratory syncytial virus infections in immunocompromised adults. Journal of Clinical Microbiology 1996;34:1649–53.
- [6] Gerna G, Vitulo P, Rovida F, Lilleri D, Pellegrini C, Oggionni T, et al. Impact of human metapneumovirus and human cytomegalovirus versus other respiratory viruses on the lower respiratory tract infections of lung transplant recipients. Journal of Medical Virology 2006;78:408–16.
- [7] Chang AB, Yerkovich ST, Gibson PG, Anderson-James S, Petsky HL, Carroll ML, et al. Pulmonary innate immunity in children with protracted bacterial bronchitis. Journal of Pediatrics 2012;161:621–5, e1.
- [8] de Blic J, Midulla F, Barbato A, Clement A, Dab I, Eber E, et al. Bronchoalveolar lavage in children. ERS Task Force on bronchoalveolar lavage in children. European Respiratory Society. European Respiratory Journal 2000;15:217–31.
- [9] Syrmis MW, Whiley DM, Thomas M, Mackay IM, Williamson J, Siebert DJ, et al. A sensitive, specific, and cost-effective multiplex reverse transcriptase-PCR assay for the detection of seven common respiratory viruses in respiratory samples. Journal of Molecular Diagnostics 2004;6:125–31.
- [10] Bialasiewicz S, Whiley DM, Lambert SB, Gould A, Nissen MD, Sloots TP. Development and evaluation of real-time PCR assays for the detection of the newly identified KI and WU polyomaviruses. Journal of Clinical Virology 2007;40:9–14.
- [11] Tozer SJ, Lambert SB, Whiley DM, Bialasiewicz S, Lyon MJ, Nissen MD, et al. Detection of human bocavirus in respiratory, fecal, and blood samples by realtime PCR. Journal of Medical Virology 2009;81:488–93.
- [12] Mackay IM, Arden KE, Speicher DJ, O'Neil NT, McErlean PK, Greer RM, et al. Cocirculation of four human coronaviruses (HCoVs) in Queensland children with acute respiratory tract illnesses in 2004. Viruses 2012;4:637–53.
- [13] Mackay IM. Human rhinoviruses. In: Schuller M, Sloots TP, James GS, Halliday CL, Carter IWJ, editors. PCR for clinical microbiology: an Australian and international perspective. London: Springer; 2010.
- [14] Marchant JM, Gibson PG, Grissell TV, Timmins NL, Masters IB, Chang AB. Prospective assessment of protracted bacterial bronchitis: airway inflammation and innate immune activation. Pediatric Pulmonology 2008;43:1092–9.
- [15] Burns JL, Emerson J, Kuypers J, Campbell AP, Gibson RL, McNamara S, et al. Respiratory viruses in children with cystic fibrosis: viral detection and clinical findings. Influenza and Other Respiratory Viruses 2012;6:218–23.
- [16] Mosser AG, Vrtis R, Burchell L, Lee WM, Dick CR, Weisshaar E, et al. Quantitative and qualitative analysis of rhinovirus infection in bronchial tissues. American Journal of Respiratory and Critical Care Medicine 2005;171:645–51.
- [17] Killington RA, Stott EJ, Lee D. The effect of temperature on the synthesis of rhinovirus type 2 RNA. Journal of General Virology 1977;36:403–11.
- [18] Papadopoulos NG, Sanderson G, Hunter J, Johnston SL. Rhinoviruses replicate effectively at lower airway temperatures. Journal of Medical Virology 1999;58:100–4.
- [19] Arruda E, Boyle TR, Winther B, Pevear DC, Gwaltney Jr JM, Hayden FG. Localization of human rhinovirus replication in the upper respiratory tract by in situ hybridization. Journal of Infectious Diseases 1995;171:1329–33.
 [20] Papadopoulos NG, Bates PJ, Bardin PG, Papi A, Leir SH, Fraenkel DJ,
- [20] Papadopoulos NG, Bates PJ, Bardin PG, Papi A, Leir SH, Fraenkel DJ, et al. Rhinoviruses infect the lower airways. Journal of Infectious Diseases 2000;181:1875–84.
- [21] Ison MG, Hayden FG, Kaiser L, Corey L, Boeckh M. Rhinovirus infections in hematopoietic stem cell transplant recipients with pneumonia. Clinical Infectious Diseases 2003;36:1139–43.
- [22] Denny Jr FW. The clinical impact of human respiratory virus infections. American Journal of Respiratory and Critical Care Medicine 1995;152:S4–12.
- [23] Johnston SL, Pattemore PK, Sanderson G, Smith S, Lampe F, Josephs L, et al. Community study of role of viral infections in exacerbations of asthma in 9–11 year old children. British Medical Journal 1995;310:1225–9.
- [24] Becroft DM. Bronchiolitis obliterans, bronchiectasis, and other sequelae of adenovirus type 21 infection in young children. Journal of Clinical Pathology 1971;24:72–82.
- [25] Jansen RR, Wieringa J, Koekkoek SM, Visser CE, Pajkrt D, Molenkamp R, et al. Frequent detection of respiratory viruses without symptoms: toward

defining clinically relevant cutoff values. Journal of Clinical Microbiology 2011;49:2631-6.

- [26] Advani S, Sengupta A, Forman M, Valsamakis A, Milstone AM. Detecting respiratory viruses in asymptomatic children. Pediatric Infectious Disease Journal 2012;31:1221–6.
- [27] van der Zalm MM, Wilbrink B, van Ewijk BE, Overduin P, Wolfs TF, van der Ent CK. Highly frequent infections with human rhinovirus in healthy young children: a longitudinal cohort study. Journal of Clinical Virology 2011;52:317–20.
- [28] Franz A, Adams O, Willems R, Bonzel L, Neuhausen N, Schweizer-Krantz S, et al. Correlation of viral load of respiratory pathogens and co-infections with disease severity in children hospitalized for lower respiratory tract infection. Journal of Clinical Virology 2010;48:239–45.
- [29] Pavia AT. Viral infections of the lower respiratory tract: old viruses, new viruses, and the role of diagnosis. Clinical Infectious Diseases 2011;52:S284–9.
- [30] Brandt CD, Kim HW, Vargosko AJ, Jeffries BC, Arrobio JO, Rindge B, et al. Infections in 18,000 infants and children in a controlled study of respiratory tract disease. I. Adenovirus pathogenicity in relation to serologic type and illness syndrome. American Journal of Epidemiology 1969;90:484–500.
- [31] Fox JP, Brandt CD, Wassermann FE, Hall CE, Spigland I, Kogon A, et al. The virus watch program: a continuing surveillance of viral infections in metropolitan New York families. VI. Observations of adenovirus infections: virus excretion patterns, antibody response, efficiency of surveillance, patterns of infections, and relation to illness. American Journal of Epidemiology 1969;89: 25–50.
- [32] Fox JP, Hall CE, Cooney MK. The Seattle Virus Watch. VII. Observations of adenovirus infections. American Journal of Epidemiology 1977;105: 362–86.
- [33] Garnett CT, Erdman D, Xu W, Gooding LR. Prevalence and quantitation of species C adenovirus DNA in human mucosal lymphocytes. Journal of Virology 2002;76:10608–16.
- [34] Jarjour NN, Gern JE, Kelly EA, Swenson CA, Dick CR, Busse WW. The effect of an experimental rhinovirus 16 infection on bronchial lavage neutrophils. Journal of Allergy and Clinical Immunology 2000;105:1169–77.
- [35] Grigg J, Riedler J, Robertson CF. Bronchoalveolar lavage fluid cellularity and soluble intercellular adhesion molecule-1 in children with colds. Pediatric Pulmonology 1999;28:109–16.

CHAPTER 4

4.1 INTRODUCTION TO CHAPTER 4

Adenoviruses are known to be associated with bronchiectasis and bronchiolitis obliterans. In Chapters 2 and 3 we have identified the potential role of viruses in chronic respiratory diseases in children. We have also demonstrated a link between neutrophilic lower airways inflammation and presence of adenovirus (as detected by PCR in BAL or NPA).

In Chapter 4 we explored the possible role of adenoviruses in chronic endobronchial suppuration. Specifically, we examined the adenovirus genotypes detected in the lower airways of a large cohort of children with PBB and bronchiectasis. We also explored the potential link between adenovirus and presence of bacterial infection. In doing so, we aimed to further expand our understanding of the role of viral-bacterial co-infection in the pathogenesis of chronic suppurative lung disorders in children.

4.2 Adenovirus Species C Is Associated With Chronic Suppurative Lung Diseases in Children

Adenovirus Species C Is Associated With Chronic Suppurative Lung Diseases in Children

Danielle F. Wurzel,^{1,2} Ian M. Mackay,⁴ Julie M. Marchant,^{1,2} Claire Y. T. Wang,⁴ Stephanie T. Yerkovich,⁵ John W. Upham,⁶ Heidi C. Smith-Vaughan,⁷ Helen L. Petsky,^{2,3} and Anne B. Chang^{2,3,7}

¹Queensland Children's Medical Research Institute, The University of Queensland, ²Queensland Children's Respiratory Centre, Royal Children's Hospital, and ³Queensland Children's Medical Research Institute, Queensland University of Technology, Brisbane; ⁴Queensland Paediatric Infectious Diseases Laboratory, Queensland Children's Medical Research Institute, Sir Albert Sakzewski Virus Research Centre, Children's Health Queensland Hospital and Health Service, The University of Queensland, Herston; ⁵Queensland Lung Transplant Service, Prince Charles Hospital, and ⁶School of Medicine, The University of Queensland, Brisbane; and ⁷Child Health Division, Menzies School of Health Research, Charles Darwin University, Darwin, Australia

Background. The role of human adenoviruses (HAdVs) in chronic respiratory disease pathogenesis is recognized. However, no studies have performed molecular sequencing of HAdVs from the lower airways of children with chronic endobronchial suppuration. We thus examined the major HAdV genotypes/species, and relationships to bacterial coinfection, in children with protracted bacterial bronchitis (PBB) and mild bronchiectasis (BE).

Methods. Bronchoalveolar lavage (BAL) samples of 245 children with PBB or mild (cylindrical) BE were included in this prospective cohort study. HAdVs were genotyped (when possible) in those whose BAL had HAdV detected (HAdV⁺). Presence of bacterial infection (defined as $\geq 10^4$ colony-forming units/mL) was compared between BAL HAdV⁺ and HAdV negative (HAdV⁻) groups. Immune function tests were performed including blood lymphocyte subsets in a random subgroup.

Results. Species C HAdVs were identified in 23 of 24 (96%) HAdV⁺ children; 13 (57%) were HAdV-1 and 10 (43%) were HAdV-2. An HAdV⁺ BAL was significantly associated with bacterial coinfection with *Haemophilus influenzae, Moraxella catarrhalis,* or *Streptococcus pneumoniae* (odds ratio [OR], 3.27; 95% confidence interval, 1.38–7.75; P = .007) and negatively associated with *Staphylococcus aureus* infection (P = .03). Young age was related to increased rates of HAdV⁺. Blood CD16 and CD56 natural killer cells were significantly more likely to be elevated in those with HAdV (80%) compared with those without (56.1%) (P = .027).

Conclusions. HAdV-C is the major HAdV species detected in the lower airways of children with PBB and BE. Younger age appears to be an important risk factor for $HAdV^+$ of the lower airways and influences the likelihood of bacterial coinfection.

Keywords. bronchiectasis; protracted bacterial bronchitis; respiratory bacteria; respiratory viruses; children.

The burden (prevalence, cost, and importance) of protracted bacterial bronchitis (PBB) and bronchiectasis (BE) are increasingly appreciated [1–4]. PBB is the most common cause of chronic cough in children presenting to pediatric pulmonologists in some series [1, 5]. Over the last 2 decades, the diagnosis of BE in

Clinical Infectious Diseases 2014;59(1):34-40

children and adults has increased [6,7]. Both conditions are characterized by chronic wet cough, lower airway endobronchial suppuration with bacterial infection, intense airway neutrophilia, and upregulation of inflammatory and immune markers [5, 8–10]. It is hypothesized, but remains unproven, that PBB and BE represent a clinical continuum based upon degree of severity, sharing common triggers and/or pathophysiology [11].

To date, research into the microbiology of chronic suppurative lung diseases (PBB and BE) in children has largely focused on bacterial pathogens, with a paucity of research into viral contributors. Viral–bacterial coinfection is associated with heightened neutrophilic inflammation of the lower airways of children [12]. Human adenovirus (HAdV) detection is associated with lower

Received 12 January 2014; accepted 21 March 2014; electronically published 18 April 2014.

Correspondence: Danielle Wurzel, MBBS, FRACP, Queensland Children's Medical Research Institute, Royal Children's Hospital, Brisbane, Queensland 4006, Australia (daniellewurzel@netspace.net.au).

[©] The Author 2014. Published by Oxford University Press on behalf of the Infectious Diseases Society of America. All rights reserved. For Permissions, please e-mail: journals.permissions@oup.com. DOI: 10.1093/cid/ciu225

airway neutrophilic inflammation in children with chronic respiratory symptoms, [13] and is significantly more likely to be present in the lower airways of children with PBB, compared with controls [14]. HAdVs (particularly types 1–5, 7, 14, and 21; members of HAdV species B, C, and E) are known to be associated with future small airways dysfunction and bronchiectasis [15, 16]. However, no studies, to our knowledge, have examined the HAdV genotypes in the lower airways of children with chronic endobronchial suppuration to investigate a possible role in pathogenesis.

Thus, we studied the bronchoalveolar lavage (BAL) of 245 children with PBB and mild BE. We aimed to identify (1) the prevalence of HAdV; (2) the diversity of genotypes/species using sequence analysis, and (3) whether presence of HAdV increased the odds of bacterial coinfection. We hypothesized that, in the lower airways of children with PBB and mild BE, certain HAdV genotypes/species would predominate, and that presence of HAdV increases the risk of bacterial coinfection.

METHODS

Children included in this study were a subset of a larger prospective, longitudinal cohort study on chronic cough in children [12]. Ethical approval was obtained from The Queensland Children's Health Services Human Research Ethics Committee (HREC/03/QRCH/17). Written informed consent was obtained from the parents/guardians.

Children were recruited between March 2008 and September 2013, covering 6 winter seasons. Of 398 children undergoing flexible bronchoscopy and BAL for clinical indication, 245 were diagnosed with PBB or BE, and were therefore eligible for inclusion. Children were divided into 2 groups: BAL positive for HAdV (HAdV⁺) or BAL negative for HAdV (HAdV⁻).

Key demographics and cough characteristics were obtained via completion of a standardized clinical questionnaire. Prospective follow-up of participants was undertaken with daily cough diaries and monthly contact via email or telephone.

Basic immune function tests were performed on peripheral blood, including full blood examination, immunoglobulins (IgG, IgA, IgM, IgE), IgG subclasses, specific antibody (IgG) responses to *Haemophilus influenzae* type b and *Clostridium tetani*, and lymphocyte subsets (including CD16⁺ and CD56⁺ natural killer [NK] cells). Bronchoscopy and BAL were performed as described previously [13], in accordance with European Respiratory Society guidelines [17]. Quantitative bacteriology and cellularity on BAL specimens were undertaken as outlined in a prior study [8].

Microbiology

Standardized semiquantitative culture of BAL fluid was undertaken using routine laboratory techniques, as described previously [8]. Infection with major respiratory pathogens (*H. influenzae*, *Moraxella catarrhalis*, *Streptococcus pneumoniae*, and *Staphylococcus aureus*) was defined by bacterial load of $\geq 10^4$ colony-forming units (CFU)/mL BAL, using standard culture techniques described elsewhere [8].

Real-time polymerase chain reaction (PCR) [18–21] was used to detect a conventional panel of respiratory viruses in all, and an extended panel in a random subset of, children. Viruses included HAdVs, respiratory syncytial virus, human influenza A virus, human influenza B virus, human parainfluenza viruses 1–3 (HPIV1-3), human metapneumovirus (standard panel), human rhinoviruses, human bocavirus, human coronaviruses (NL63, OC43, 229E, HKU1), WU polyomavirus, and KI polyomavirus (extended panel).

HAdV PCR testing was undertaken on all BAL samples at a clinical laboratory (Pathology Queensland, Royal Brisbane and Women's Hospital, Brisbane, Australia), using a previously described method [22], to identify HAdV⁺ participants. A random subset of HAdV isolates, based upon sample availability, from 26 HAdV⁺ on BAL children, underwent nucleotide sequencing at a research laboratory (Queensland Paediatric Infectious Diseases Laboratory, Brisbane, Australia).

HAdV genotyping, based on the hexon gene hypervariable regions 1-6, was conducted as previously described [23]. In brief, 2 µL of nucleic acid extract was amplified in a nested PCR reaction. Round 1 contained primers (0.38 µM; AdHexF1-TICTTTGACA-TICGIGGIGTICTIGA and AdHexR12-CTGTCIACIGCCTG RTTCCACA), magnesium chloride (4.75 µM), buffer, and MyTaq HS DNA polymerase (Bioline kit, Australia) and was incubated for 1 minute at 94°C followed by 35 cycles of 94°C for 1 minute, 45°C for 1 minute, 72°C for 2 minutes, and a final incubation at 72°C for 1 minute. Round 2 used 2 µL of the Round 1 PCR product as template, the same conditions but different primers (AdHexF2- GGYCCYAGYTTYAARCCCTAYTC and Ad-HexR2- GGTTCTGTCICCCAGAGARTCIAGCA). Round 2 PCR products were checked by agarose gel electrophoresis, purified, and subjected to nucleotide sequencing (Australian Equine Genetics Research Centre, UQ) results were analyzed to determine the HAdV genotype (Geneious Pro version 6.1).

Relationships between HAdV genotypes, sequenced from 24 HAdV⁺ study children, were illustrated in a phylogenetic tree (Figure 1) constructed using the neighbor-joining method (with evolutionary history inferred) in MEGA5 [24] after sequence alignment in Geneious version 6.1 [25]. Additional sequences were included from GenBank (labeled with genotype and accession number) to define the species according to a previously described approach [23].

Clinical Definitions

BE was diagnosed based on radiological criteria [26] in children with clinical symptoms consistent with BE [27]. All had evidence of mild (cylindrical) BE on high-resolution computed tomography (CT) scan (reconstructed from a multidetector CT scan). A



Figure 1. A neighbor-joining method of phylogenetic analysis of human adenovirus (HAdV) genotypes and their assigned species. The 24 nucleotide sequences (approximately 800 nt long) from the hexon gene of HAdV-positive samples from study patients are prefixed with Royal Children's Hospital (RCH). The evolutionary distances were computed using the maximum composite likelihood method [23] and are in the units of the number of base substitutions per site. Additional sequences for speciation are included from GenBank (labeled with genotype and accession number in bold). The percentage of replicate trees in which the associated taxa clustered together in the bootstrap test (500 replicates) are shown next to the major branches.

diagnosis of PBB was made in children who fulfilled the following criteria: (1) history of chronic (\geq 4 weeks) wet cough, (2) prospective evidence (supported by cough diaries) of response to 2 weeks of amoxicillin-clavulanate antibiotics, and (3) absence of clinical pointers suggesting an alternative cause for cough [5].

Statistical Analysis

Descriptive statistics were used to summarize baseline patient characteristics. Median and interquartile range (IQR) were

reported as data were nonnormally distributed. Univariate analyses were performed using Pearson χ^2 (or Fisher exact test) for categorical variables. Mann-Whitney *U* test was used for 2group comparisons and Kruskal-Wallis test for >2-group comparisons of continuous variables. Binary logistic regression was used to calculate odds ratios and *P* values. A 2-tailed *P* value of <.05 was considered statistically significant. Statistical analyses were performed using IBM SPSS version 20.0 (IBM SPSS, Armonk, New York).

Table 1. Characteristics of the Study Population

		Age	Age, mo		Sex	
Group	No. (%)	Median	(IQR)	Μ	F	
HAdV ⁺	40	19	(13–25)	29	11	
PBB	33 (83)	17	(12–22)	24	9	
BE	7 (18)	30	(22–65)	5	2	
HAdV ⁻	205	37	(18–69)	126	79	
PBB	126 (61)	26	(15–56)	78	48	
BE	79 (39)	57	(31–90)	48	31	

Abbreviations:+/-, positive/negative detection; BE,bronchiectasis; HAdV, human adenovirus; IQR, interquartile range; PBB,protracted bacterial bronchitis.

RESULTS

Study Population

Of the 245 children with PBB or BE, 40 were HAdV⁺ and 205 HAdV⁻. The median age of the children in the study was 30 months (IQR, 17–63 months), 23 months (IQR, 14–47 months) in those with PBB, and 57 months (IQR, 30–87 months) in those with BE. The median age was similar between HAdV⁺ and HAdV⁻ groups in children with BE (P = .312), Table 1. In children with PBB, those in the HAdV⁺ group were significantly younger than the HAdV⁻ group (P = .001). There were no differences in the sex of the children according to HAdV status within the PBB or BE groups (P = .248 and P = .703). No seasonal differences in HAdV detection rates were observed (P = .506).

CD 16⁺ and CD56⁺ NK cells were elevated above the normal range in 20–25 (80%) of HAdV⁺ and 64 of 114 (56.1%) of HAdV⁻ children (P = .027). When HAdV was excluded, similar rates of NK cell elevation were seen in children with or without standard panel respiratory viruses detected in BAL (P = .601). Basic immune function tests were otherwise normal.

HAdV Genotypes

HAdV genotyping was performed on a random subset (based upon sample availability) of 26 HAdV⁺ on BAL children. Of 24 typeable HAdV isolates, 23 (96%) were identified as HAdV species C; 13 (57%) as HAdV-1, and 10 (43%) as HAdV-2. A single HAdV⁺ isolate was identified as genotype 4, species E, in a child with low-grade fever and mild upper respiratory tract symptoms at time of bronchoscopy (Figure 1).

Bacterial Codetection With HAdV

The major bacterial pathogens detected in both $HAdV^+$ and $HAdV^-$ groups were *H. influenzae*, *M. catarrhalis*, and *S. pneumoniae*. $HAdV^+$ status showed positive association with each major bacterial pathogen (Table 2). *Staphylococcus aureus* was

Table 2.Univariate Logistic Regression Showing RelationshipsBetween Human Adenovirus Status and Bacterial Infection onBronchoalveolar Lavage

Bacterial	HAdV ⁺	HAdV ⁻		Р
Infection	(n = 40)	(n = 205)	OR (95% CI)	Value
Haemophilus influenzae	27 (68%)	96 (47%)	2.36 (1.15–4.83)	.019
Moraxella catarrhalis	14 (35%)	38 (19%)	2.37 (1.13–4.96)	.022
Streptococcus pneumoniae	14 (35%)	46 (22%)	1.86 (.90–3.85)	.094
Staphylococcus aureus	0 (0%)	21 (10.2%)		.030 ^a

Abbreviations: +/–, positive/negative detection; CI, confidence interval; HAdV, human adenovirus; OR, odds ratio.

^a Calculated using Fisher exact test.

the fourth most common bacterium causing lower airway infection; however, a significant negative association with $HAdV^+$ status was observed (*P* = .03).

Lower airway infection with 1 or more of the 3 major bacteria showed significant positive association with HAdV detection (odds ratio [OR], 3.273 [95% confidence interval {CI}, 1.382–7.748]; P = .007). After adjustment for age, using multiple logistic regression, this association was no longer significant (OR, 2.383 [95% CI, .980–5.794]; P = .055), whereas the relationship between bacterial infection and age remained significant (OR, 0.987 [95% CI, .980–.994]; P < .001].

Although there were slightly more detections of HAdV-1 compared to HAdV-2, no further associations between individual genotypes and bacterial coinfection were observed (data not shown).

The HAdV⁺ group was more likely to have lower airway infection with multiple (2 or more) major bacterial species (excluding *S. aureus*), compared with the HAdV⁻ group (OR, 2.471 [95% CI, 1.228–4.969]; P = .011). However, on age adjustment, using multiple logistic regression, this association was no longer significant (OR, 1.674; [95% CI, .807–3.475], P = .166).

Other Pathogens

Mycoplasma pneumoniae detection rates (using PCR) were similar between HAdV⁻ and HAdV⁺ groups (4 in HAdV⁻ and a 1 in HAdV⁺ groups; P = .580). Similarly, standard panel respiratory viruses were detected at similar rates in HAdV⁻ and HAdV⁺ children. There were 3 (1.5%) and 0 detections of influenza virus (P = 1.00) and 10 (4.9%) and 2 (5%) detections of HPIV (P = 1.00) in the HAdV⁻ and HAdV⁺ groups, respectively. Respiratory syncytial virus was detected in 11 (5.4%) and 0 (P = .22), and human metapneumovirus was detected in 5 (2.4%) and 0 (P = .594) children in the HAdV⁻ and HAdV⁺ groups, respectively.

Of 52 (41 HAdV⁻ and 11 HADV⁺) participants who had extended viral panel testing of BAL, approximately a quarter of HAdV⁻ children were human rhinovirus positive on BAL (n = 10 [24.4%]) compared with more than half of HAdV⁺ children (n = 6 [54.5%]; P = .054). Likewise, human bocavirus was detected less commonly in HAdV⁻, compared to HAdV⁺ children (1 of 41 [2.4%] vs 3 of 11 [27.3%]; P = .026). Human coronavirus (OC43) was detected in 1 HAdV⁻ participant and no HAdV⁺ participants (P = 1.00).

DISCUSSION

This is the first study to examine the HAdV genotypes detectable in the lower airways of children with chronic endobronchial suppuration (PBB and bronchiectasis). We found HAdV species C (genotypes 1 and 2) to be the major HAdV species in the BAL of these children, irrespective of season. We have also shown that lower airway bacterial infection, with *H. influenzae*, *M. catarrhalis*, and *S. pneumoniae*, but not *S. aureus*, are increased in HAdV⁺ children. Younger age is an independent predictor of infection with HAdV and common respiratory bacteria, and age increases the odds of viral-bacterial coinfection. Our finding of elevated CD16 and CD56 NK cells in the blood of 80% of HAdV⁺ children provides indirect evidence of a systemic immune response to HAdV in the airways of these children.

The HAdV-C species, comprising genotypes 1, 2, 5, 6, and 57, is one of the most frequent HAdV species known to infect the respiratory tract of children [28]. Infection can result in a range of clinical manifestations, from uncomplicated upper respiratory infection to severe pneumonia. Most primary HAdV-C infections occur within the first 2 years of life [29]. The highest levels of HAdV DNA are detected in adenotonsillar tissue of 2-year-old children undergoing routine adenoidectomy or tonsillectomy, and the amount of HAdV DNA decreases with increasing age [30]. HAdVs demonstrate the ability to establish latent infection within lymphocytes [30], tonsillar tissue [31], and the lung [32] and are capable of evading host innate immune responses via multiple mechanisms [33].

Several studies have demonstrated HAdV's propensity to establish latent and/or persistent infection within the upper and lower respiratory tract [34, 35]. In a recent longitudinal study of children with chronic upper respiratory infections, 13 had repeated HAdV detection; 8 carried an identical HAdV genotype on successive occasions, suggesting chronic rather than repeated infection [34]. Using repeated lower airway sampling, Macek and colleagues demonstrated persistence of HAdV in the lower airways of 9 of 11 children with persistent asthma [35]. In adults, Matsuse and colleagues showed increased presence of latent HAdV DNA in the lung tissue of patients with chronic obstructive pulmonary disease (COPD), compared with those without COPD [32]. No studies to date, however, have genotyped HAdV in lower airways of children with chronic endobronchial suppuration.

A higher prevalence of HAdV in the airways of patients with chronic respiratory diseases has led to speculation that "smoldering HAdV at the site of lung inflammation" [30] contributes to the pathogenesis of chronic lung diseases, including asthma in children [35] and COPD in adults [32, 36]. Marin et al showed that HAdV was detected in 78.4% of nasopharyngeal samples of asthmatic children during symptom-free periods, vs 5% of healthy controls [37]. Furthermore, we have recently shown that HAdV is significantly more likely to be detected in BAL fluid of children with PBB (23%) compared with controls (4%) [14]. It is possible that chronic inflammatory diseases of the respiratory tract predispose to HAdV persistence in the respiratory tract, or that HAdVs induce an inflammatory response that may be a key factor in chronic respiratory disease pathogenesis. If the latter is correct, the association of HAdV with bacterial infection of the lower airways is likely to be a relevant and significant finding.

Childhood lower respiratory infections are known to cause bronchiectasis. Nontypeable *H. influenzae* (NTHi) is the most common bacterium in the lower airways of children with PBB and other chronic lung diseases [5, 38, 39]. Recently, De Schutter and colleagues showed that NTHi is also the commonest cause of (nonresponsive or recurrent) community-acquired pneumonia in children [40]. Similarly, compared with controls, elevated rates of *H. influenzae*, *M. catarrhalis*, and *S. pneumoniae* infection are detected in the lower airways of children with PBB [14].

The propensity for codetection of HAdV with bacterial pathogens (eg, NTHi), has been documented in studies on human respiratory diseases [41, 42]. However, the most convincing evidence for microbial synergy arises from an animal study on otitis media. In a chinchilla model of experimental otitis media due to HAdV-1 (species C) and NTHi, Suzuki et al demonstrated a synergistic effect of the 2 pathogens. They found that inflammation was greatest in the presence of both HAdV and NTHi, compared with HAdV or NTHi alone. Furthermore, timing of inoculation was important. HAdV inoculation prior to NTHi inoculation resulted in the greatest tympanic membrane inflammation and mucosal dysfunction [43]. We have previously shown that neutrophilic lower airway inflammation is maximal in the presence of viral-bacterial coinfection of the airways of children [12]. It is indeed plausible that HAdV-C and H. influenzae may also play a synergistic role in the initiation and/or exacerbations of chronic suppurative lung diseases in children, and further research is needed.

The negative association of HAdV with *S. aureus* in the lower airways observed in the present study may be due to the increased presence of other bacterial pathogens (eg, *H. influenzae*) in HAdV⁺ compared with HAdV⁻ children. Negative associations

between *S. pneumoniae* and *S. aureus* [44–48] and between *H. influenzae* and *S. aureus* [44, 45] are well described in studies of the upper respiratory tract. These interactions are believed to reflect natural competition between colonization with these organisms [48].

Hence, our findings in relation to HAdV in the lower airways of children with PBB and BE are likely to have clinical relevance and may represent an important clue to the underlying pathogenesis of these disease processes. Our hypothesis is based upon several key findings. These include (1) a predominance of HAdV-C in the airways of these children, with a plausible mechanism for disease; (2) the presence of elevated NK cell levels in HAdV⁺ children (however, not in those with other standard panel respiratory viruses), indicating a systemic immune response to HAdV in the lung; (3) previous research linking HAdV (in the nasopharynx and/or lung) to heightened neutrophilic inflammation and lymphopenia of the lower airways (suggestive of active viral replication) [13]; and (4) the demonstrated association between HAdV⁺ status and lower airway bacterial infection (the latter being a key feature of PBB and BE). It is therefore unlikely that HAdV is an innocent bystander, as early epidemiologic studies suggested.

Although we have described 2 novel and important findings, several limitations to our study merit discussion. First, although our cross-sectional data highlights important associations, we cannot attribute causality. Repeated lower airway sampling would be needed to establish the temporal sequence of lower airway infection and determine a cause-and-effect relationship. However, subjecting children to repeated general anesthesia for lower airway sampling, solely for research, would be unethical. Second, we did not include HAdV serology in our blood test panel. This ideally requires repeat venipuncture, which was not feasible in our context as many children in the study reside outside the Brisbane area. With regards to methods of HAdV detection, although culture detects infectious virus, PCR is more sensitive and is the current clinical standard [16]. Although a positive PCR for HAdV does not readily distinguish infectious from noninfectious virus, the identification of HAdV DNA indicates that infection has occurred at some stage. Further, HAdV's association with a systemic immune response (ie, elevated NK cells) supports the notion that HAdV may play a clinically relevant role. Third, we detected few conventional respiratory panel viruses (other than HAdV) and only performed extended viral panel analyses on a subset, limiting our ability to explore other possible viral-bacterial and viralviral interactions. Last, the strong relationship between age and HAdV status, and the increased odds of bacterial coinfection with HAdV in younger children, is likely to be significant. This finding raises the question of whether younger children are predisposed to polymicrobial infection, via immature immune system processes, rather than a true viral-bacterial interaction. Timing of HAdV acquisition, with respect to age, may therefore be important in further evaluating the link between HAdV and future development of PBB and/or BE.

We conclude that HAdV-C genotypes 1 and 2 are the dominant HAdV species infecting the lower airways of young children with chronic endobronchial suppuration. The significant association between HAdV and lower airway bacterial infection suggests a possible role of HAdV-C in the pathogenesis of chronic suppurative lung diseases in young children. Our findings may have implications for targets in the prevention of chronic suppurative lung diseases; however, further research to definitely establish causality would be required.

Notes

Acknowledgments. We thank the families that participated in this study; Sophie Anderson-James, Carol Willis, and Joanna Tuppin for their assistance with patient recruitment, data entry, and study coordination; and Xiaoyan Lu (US Centers for Disease Control and Prevention) for guidance in HAdV genotyping.

Disclaimer. The views expressed in this publication are those of the authors and do not reflect the views of the National Health and Medical Research Council (NHMRC).

Financial support. This work was supported by the NHMRC (project grant 1042601 and Centre of Research Excellence grant 1040830) and the Financial Markets Foundation for Children (project grant 2010-005). D. F. W. is supported by scholarships from the Thoracic Society of Australia and New Zealand/Allen and Hanbury's, Queensland Children's Medical Research Institute and NHMRC (1039688). A. B. C. is supported by NHMRC practitioner fellowships (grants 545216 and 1058213).

Potential conflicts of interest. I. M. M. has received payment for consultancy services from Firebrick Pharma. J. W. U. has served on the advisory board for Novartis and has previously received payments for lectures or served on speakers' bureaus for AstraZeneca, GSK, Novartis, and Boehringer. All other authors report no potential conflicts.

All authors have submitted the ICMJE Form for Disclosure of Potential Conflicts of Interest. Conflicts that the editors consider relevant to the content of the manuscript have been disclosed.

References

- Chang AB, Robertson CF, van Asperen PP, et al. A cough algorithm for chronic cough in children: a multicenter, randomized controlled study. Pediatrics 2013; 131:e1576–83.
- Kapur N, Masters IB, Newcombe P, Chang AB. The burden of disease in pediatric non-cystic fibrosis bronchiectasis. Chest 2012; 141:1018–24.
- McShane PJ, Naureckas ET, Tino G, Strek ME. Non-cystic fibrosis bronchiectasis. Am J Respir Crit Care Med 2013; 188:647–56.
- Joish VN, Spilsbury-Cantalupo M, Operschall E, Luong B, Boklage S. Economic burden of non-cystic fibrosis bronchiectasis in the first year after diagnosis from a US health plan perspective. Appl Health Econ Health Policy 2013; 11:299–304.
- Marchant JM, Masters IB, Taylor SM, Cox NC, Seymour GJ, Chang AB. Evaluation and outcome of young children with chronic cough. Chest 2006; 129:1132–41.
- 6. Australian Institute of Health and Welfare. Bronchiectasis. Australian Government, Canberra, Australia, **2013**:1–6.
- Seitz AE, Olivier KN, Adjemian J, Holland SM, Prevots R. Trends in bronchiectasis among medicare beneficiaries in the United States, 2000 to 2007. Chest 2012; 142:432–9.
- 8. Marchant JM, Gibson PG, Grissell TV, Timmins NL, Masters IB, Chang AB. Prospective assessment of protracted bacterial bronchitis: airway

- 9. Kapur N, Grimwood K, Masters IB, Morris PS, Chang AB. Lower airway microbiology and cellularity in children with newly diagnosed non-CF bronchiectasis. Pediatr Pulmonol **2012**; 47:300–7.
- 10. Mandal P, Hill AT. Bronchiectasis: breaking the cycle of inflammation and infection. Lancet Respir Med **2013**; 1:e5–6.
- Chang AB, Redding GJ, Everard ML. Chronic wet cough: protracted bronchitis, chronic suppurative lung disease and bronchiectasis. Pediatr Pulmonol 2008; 43:519–31.
- Wurzel DF, Marchant JM, Clark JE, et al. Wet cough in children: infective and inflammatory characteristics in broncho-alveolar lavage fluid [Epub ahead of print]. Pediatr Pulmonol 2013. doi:10.1002/ ppul.22792.
- 13. Wurzel DF, Marchant JM, Clark JE, et al. Respiratory virus detection in nasopharyngeal aspirate versus bronchoalveolar lavage is dependent on virus type in children with chronic respiratory symptoms. J Clin Virol **2013**; 58:683–8.
- 14. Wurzel D, Marchant J, Yerkovich ST, et al. Prospective characterisation of protracted bacterial bronchitis (PBB) in children [Epub ahead of print]. Chest **2013**. doi:10.1378/chest.13-2442.
- Becroft DM. Bronchiolitis obliterans, bronchiectasis, and other sequelae of adenovirus type 21 infection in young children. J Clin Pathol 1971; 24:72–82.
- Mandell, Douglas, and Bennett's principles and practice of infectious diseases. Edited by Gerald L. Mandell, John E. Bennett, and Raphael Dolin Philadelphia, PA: Churchill Livingstone, Elsevier, 2010.
- de Blic J, Midulla F, Barbato A, et al. Bronchoalveolar lavage in children. ERS Task Force on bronchoalveolar lavage in children. European Respiratory Society. Eur Respir J 2000; 15:217–31.
- Syrmis MW, Whiley DM, Thomas M, et al. A sensitive, specific, and cost-effective multiplex reverse transcriptase-PCR assay for the detection of seven common respiratory viruses in respiratory samples. J Mol Diagn 2004; 6:125–31.
- Bialasiewicz S, Whiley DM, Lambert SB, Gould A, Nissen MD, Sloots TP. Development and evaluation of real-time PCR assays for the detection of the newly identified KI and WU polyomaviruses. J Clin Virol 2007; 40:9–14.
- Tozer SJ, Lambert SB, Whiley DM, et al. Detection of human bocavirus in respiratory, fecal, and blood samples by real-time PCR. J Med Virol 2009; 81:488–93.
- Mackay IM, Arden KE, Speicher DJ, et al. Co-circulation of four human coronaviruses (HCoVs) in Queensland children with acute respiratory tract illnesses in 2004. Viruses 2012; 4:637–53.
- Heim A, Ebnet C, Harste G, Pring-Akerblom P. Rapid and quantitative detection of human adenovirus DNA by real-time PCR. J Med Virol 2003; 70:228–39.
- Lu X, Erdman DD. Molecular typing of human adenoviruses by PCR and sequencing of a partial region of the hexon gene. Arch Virol 2006; 151:1587–602.
- 24. Tamura K, Peterson D, Peterson N, Stecher G, Nei M, Kumar S. MEGA5: molecular evolutionary genetics analysis using maximum likelihood, evolutionary distance, and maximum parsimony methods. Mol Biol Evol 2011; 28:2731–9.
- Drummond A, Ashton B, Buxton S, et al. Geneious v5.4 [computer program], 2011. Available at: http://www.geneious.com. Accessed April 2014.
- Kapur N, Masel JP, Watson D, Masters IB, Chang AB. Bronchoarterial ratio on high-resolution CT scan of the chest in children without pulmonary pathology: need to redefine bronchial dilatation. Chest 2011; 139:1445–50.
- Chang AB, Bell SC, Byrnes CA, et al. Chronic suppurative lung disease and bronchiectasis in children and adults in Australia and New Zealand. Med J Aust 2010; 193:356–65.

- Munoz FM, Piedra PA, Demmler GJ. Disseminated adenovirus disease in immunocompromised and immunocompetent children. Clin Infect Dis 1998; 27:1194–200.
- 29. Edwards KM, Thompson J, Paolini J, Wright PF. Adenovirus infections in young children. Pediatrics **1985**; 76:420–4.
- Garnett CT, Erdman D, Xu W, Gooding LR. Prevalence and quantitation of species C adenovirus DNA in human mucosal lymphocytes. J Virol 2002; 76:10608–16.
- Neumann R, Genersch E, Eggers HJ. Detection of adenovirus nucleic acid sequences in human tonsils in the absence of infectious virus. Virus Res 1987; 7:93–7.
- Matsuse T, Hayashi S, Kuwano K, Keunecke H, Jefferies WA, Hogg JC. Latent adenoviral infection in the pathogenesis of chronic airways obstruction. Am Rev Respir Dis 1992; 146:177–84.
- Mahr JA, Gooding LR. Immune evasion by adenoviruses. Immunol Rev 1999; 168:121–30.
- Kalu SU, Loeffelholz M, Beck E, et al. Persistence of adenovirus nucleic acids in nasopharyngeal secretions: a diagnostic conundrum. Pediatr Infect Dis J 2010; 29:746–50.
- Macek V, Sorli J, Kopriva S, Marin J. Persistent adenoviral infection and chronic airway obstruction in children. Am J Respir Crit Care Med 1994; 150:7–10.
- Elliott WM, Hayashi S, Hogg JC. Immunodetection of adenoviral E1A proteins in human lung tissue. Am J Respir Cell Mol Biol 1995; 12:642–8.
- Marin J, Jeler-Kacar D, Levstek V, Macek V. Persistence of viruses in upper respiratory tract of children with asthma. J Infect 2000; 41:69–72.
- Hare KM, Grimwood K, Leach AJ, et al. Respiratory bacterial pathogens in the nasopharynx and lower airways of Australian indigenous children with bronchiectasis. J Pediatr 2010; 157:1001–5.
- King P. Haemophilus influenzae and the lung (Haemophilus and the lung). Clin Transl Med 2012; 1:10.
- De Schutter I, De Wachter E, Crokaert F, et al. Microbiology of bronchoalveolar lavage fluid in children with acute nonresponding or recurrent community-acquired pneumonia: identification of nontypeable *Haemophilus influenzae* as a major pathogen. Clin Infect Dis 2011; 52:1437–44.
- Moore HC, Jacoby P, Taylor A, et al. The interaction between respiratory viruses and pathogenic bacteria in the upper respiratory tract of asymptomatic Aboriginal and non-Aboriginal children. Pediatr Infect Dis J 2010; 29:540–5.
- 42. Korppi M. Mixed microbial aetiology of community-acquired pneumonia in children. APMIS **2002**; 110:515–22.
- Suzuki K, Bakaletz LO. Synergistic effect of adenovirus type 1 and nontypeable *Haemophilus influenzae* in a chinchilla model of experimental otitis media. Infect Immun 1994; 62:1710–8.
- Pettigrew MM, Gent JF, Revai K, Patel JA, Chonmaitree T. Microbial interactions during upper respiratory tract infections. Emerg Infect Dis 2008; 14:1584–91.
- 45. Madhi SA, Adrian P, Kuwanda L, Cutland C, Albrich WC, Klugman KP. Long-term effect of pneumococcal conjugate vaccine on nasopharyngeal colonization by *Streptococcus pneumoniae*—and associated interactions with *Staphylococcus aureus* and *Haemophilus influenzae* colonization—in HIV-infected and HIV-uninfected children. J Infect Dis 2007; 196:1662–6.
- Regev-Yochay G, Dagan R, Raz M, et al. Association between carriage of Streptococcus pneumoniae and Staphylococcus aureus in children. JAMA 2004; 292:716–20.
- Veenhoven R, Bogaert D, Uiterwaal C, et al. Effect of conjugate pneumococcal vaccine followed by polysaccharide pneumococcal vaccine on recurrent acute otitis media: a randomised study. Lancet 2003; 361:2189–95.
- Bogaert D, van Belkum A, Sluijter M, et al. Colonisation by *Streptococcus pneumoniae* and *Staphylococcus aureus* in healthy children. Lancet 2004; 363:1871–2.

CHAPTER 5

5.1 INTRODUCTION TO CHAPTER 5

Thus far, we have demonstrated a relationship between adenovirus (of which most are adenovirus species C) and bacterial infection of the lower airways of children with PBB and bronchiectasis. We have provided further evidence of the role of viral-bacterial co-infection. In Chapter 5, we refocused on PBB as a clinical entity and explored the characteristics of the typical child presenting with this condition.

In this chapter we compared a cohort of children with PBB to a control group. We systematically examined their demographic, clinical, laboratory and bronchoscopic findings. Our specific objectives were to compare rates of viral infection between groups and to further assess the frequency with which wheeze was observed in PBB. The overall aim of this study was to assist clinicians in the more accurate diagnosis of PBB, highlighting the similarities and differences between PBB and other common paediatric respiratory conditions such as asthma.

5.2 Prospective Characterization of Protracted Bacterial Bronchitis in Children



CHEST

SIGNS AND SYMPTOMS OF CHEST DISEASE

Prospective Characterization of Protracted Bacterial Bronchitis in Children

Danielle F. Wurzel, MBBS; Julie M. Marchant, PhD; Stephanie T. Yerkovich, PhD; John W. Upham, PhD; Ian M. Mackay, PhD; I. Brent Masters, PhD; and Anne B. Chang, PhD

Background: Prior studies on protracted bacterial bronchitis (PBB) in children have been retrospective or based on small cohorts. As PBB shares common features with other pediatric conditions, further characterization is needed to improve diagnostic accuracy among clinicians. In this study, we aim to further delineate the clinical and laboratory features of PBB in a larger cohort, with a specific focus on concurrent viral detection.

Methods: Children with and without PBB (control subjects) undergoing flexible bronchoscopy were prospectively recruited. Basic immune function testing and lymphocyte subset analyses were performed. BAL specimens were processed for cellularity and microbiology. Viruses were identified using polymerase chain reaction (PCR) and bacteria were identified via culture.

Results: The median age of the 104 children (69% male) with PBB was 19 months (interquartile range [IQR], 12-30 mo). Compared with control subjects, children with PBB were more likely to have attended childcare (OR, 8.43; 95% CI, 2.34-30.46). High rates of wheeze were present in both groups, and tracheobronchomalacia was common. Children with PBB had significantly elevated percentages of neutrophils in the lower airways compared with control subjects, and adenovirus was more likely to be detected in BAL specimens in those with PBB (OR, 6.69; 95% CI, 1.50-29.80). Median CD56 and CD16 natural killer (NK) cell levels in blood were elevated for age in children with PBB (0.7×10^{9} /L; IQR, 0.5-0.9 cells/L).

Conclusions: Children with PBB are, typically, very young boys with prolonged wet cough and parent-reported wheeze who have attended childcare. Coupled with elevated NK-cell levels, the association between adenovirus and PBB suggests a likely role of viruses in PBB pathogenesis. *CHEST 2014; 145(6):1271–1278*

Abbreviations: CFU = colony-forming units; HAdV = human adenovirus; HCoV = human coronavirus; IQR = interquartile range; NK = natural killer; PBB = protracted bacterial bronchitis; PCR = polymerase chain reaction; PyV = polyomavirus; RAST = radioallergosorbent test

Chronic cough in children is associated with significant morbidity¹ and a high emotional burden on parents.² Protracted bacterial bronchitis (PBB) is recognized as a major cause of chronic cough in children.³⁻⁵ In a multicenter study of 346 children newly referred for chronic cough, PBB was the most common etiologic diagnosis.⁶ Prior to 2006, when the entity of PBB was first characterized by our group,⁷ bacterial bronchitis was underappreciated as a cause of chronic cough in otherwise healthy children.⁸ Several international groups have subsequently studied PBB, with concordant findings.⁹⁻¹¹ PBB has now been incorporated into cough guidelines and education resources in many countries.^{3,4,12-14}

The original diagnostic criteria for PBB included (1) wet cough for ≥ 4 weeks, (2) identifiable lower-

airway bacterial infection on BAL culture, and (3) response to antibiotics (amoxicillin/clavulanate) with resolution of cough within 2 weeks.⁷ For clinical feasibility, criterion 2 was later substituted with "absence of specific pointers to indicate an alternative specific cause of cough,"¹⁵ resulting in criteria that were applicable to both primary and tertiary care settings.

While defining PBB has been transformative in our understanding and management of chronic wet cough in children, many questions remain. To date, studies on PBB have been small⁷ and/or retrospective⁹⁻¹¹; and some argue that PBB is poorly defined,¹⁶ with others calling for more in-depth studies.¹³ Furthermore, children with coexistent viruses detected in BAL specimens were excluded from our original PBB cohort.⁷ A large prospective study with further clinical and laboratory descriptors will assist clinicians to differentiate PBB from other causes of cough in children. We hypothesized that PBB would be associated with a well-defined set of clinical characteristics and that virus rates in BAL specimens would be increased compared with those of control subjects. In our study of 104 children with PBB, we, thus, provide more extensive clinical, laboratory, and BAL characterization of PBB.

MATERIALS AND METHODS

The Queensland Children's Health Services human research ethics committee (HREC/03/QRCH/17) approved the study, and written informed consent was obtained from each parent/guardian. Children undergoing flexible bronchoscopy and BAL for a clinical indication between March 2008 and November 2012, excluding those with cystic fibrosis, were eligible for inclusion. Of 343 potentially eligible children, 104 fulfilled clinical and BAL criteria for PBB, and 49 with chronic respiratory symptoms (without PBB) were allocated as control subjects. Any child with symptoms or signs of acute lower respiratory tract infection (eg, high fever, shortness of breath or tachypnea, recent-onset wheeze, rattle or crepitations on chest auscultation), was deemed (by an anesthetist) to be unfit for anesthesia and thereby excluded.

All caregivers completed a standardized clinical questionnaire including the following: cough descriptors (eg, cough score¹⁷ and duration), current antibiotic treatment, and demographic factors (eg, age, sex, number of siblings, prior radiograph-confirmed pneumonia, indigenous status, childcare attendance, tobacco smoke exposure). Baseline examination findings were collected. Partici-

Manuscript received October 13, 2013; revision accepted December 9, 2013; originally published Online First January 16, 2014.

Affiliations: From the Queensland Children's Medical Research Institute (Drs Wurzel, Marchant, Masters, and Chang), The University of Queensland, and Queensland Children's Respiratory Centre, Royal Children's Hospital, Brisbane, QLD; School of Medicine (Drs Yerkovich and Upham), The University of Queensland, Brisbane, QLD; Queensland Lung Transplant Service (Dr Yerkovich), The Prince Charles Hospital, Brisbane, QLD; Department of Respiratory Medicine (Dr Upham), Princess Alexandra Hospital, Brisbane, QLD; Queensland Paediatric, Infectious Diseases Laboratory (Dr Mackay), Queensland Children's Medical Research Institute, Sir Albert, Sakzewski Virus Research Centre, Children's Health Queensland Hospital and Health Service, The University of Queensland, Herston, QLD; and Child Health Division (Dr Chang), Menzies School of Health Research, Charles Darwin University, Darwin, NT, Australia.

Part of this article has been presented in abstract form (Wurzel DF, Marchant JM, Mackay IM, et al. Further description of protracted bacterial bronchitis [PBB] in children. Am J Respir Crit Care Med. 2013;187:A5904).

Funding/Support: This study received funding from the National Health and Medical Research Council [project grants 1042601 and 1040830], Financial Markets Foundation for Children [project grant 2010-005], Thoracic Society of Australia and New Zealand/Allen and Hanbury's, and the Queensland Children's Medical Research Institute.

Correspondence to: Danielle F. Wurzel, MBBS, Queensland Children's Medical Research Institute, The University of Queensland, Royal Children's Hospital, Herston, QLD 4006, Australia; e-mail: Danielle.wurzel@uqconnect.edu.au

© 2014 American College of Chest Physicians. Reproduction of this article is prohibited without written permission from the American College of Chest Physicians. See online for more details. DOI: 10.1378/chest.13-2442

1272

pants had prospective follow-up with cough diaries to confirm PBB diagnosis (via antibiotic response).

Basic immune function tests were performed on peripheral blood samples, including CBC count, immunoglobulin levels (IgG, IgA, IgM, and IgE), IgG subclasses, and specific antibody (IgG) responses to *Haemophilus influenzae* type b and *Clostridium tetani*. Lymphocyte subsets (including natural killer [NK] cells) and antigen-specific IgE (radioallergosorbent test [RAST]) for common environmental allergens (eg, cat, dog, molds and yeasts, house dust mite, common grass mix) were undertaken in a subset. Assays were performed at Queensland Health Pathology.

Bronchoscopy and BAL were performed as previously described (e-Appendix 1).⁷ Multiplex polymerase chain reaction (PCR) techniques were used to detect human adenoviruses (HAdVs), respiratory syncytial virus, metapneumovirus, influenza viruses A and B, and parainfluenza viruses in BAL specimens.¹⁸ A random subset had extended viral panel testing for rhinoviruses, human bocavirus, human coronaviruses (HCoVs) (HCoV-NL63, OC43, 229E, HKU1), and polyomaviruses (PyVs) (KIPyV and WUPyV) using PCR techniques as described elsewhere.¹⁸⁻²²

Tracheomalacia was defined as tracheal deformity at end expiration that was maintained during spontaneous respiration, but which could be altered by the passage of the bronchoscope or positive airway pressure. Bronchomalacia was defined as an appearance of deformity in the right or left main-stem bronchi and/or their respective divisions at the lobar or segmental level.²³

Statistical Analyses

Statistical analyses were carried out using IBM SPSS version 20.0 (IBM). Medians and interquartile ranges (IQRs) were reported as data that were non-normally distributed. Comparisons of categorical variables were performed using the Pearson χ^2 test or Fisher exact test (if expected value was <5). For continuous variables, the Mann-Whitney *U* test was used for two-group comparisons and Kruskal-Wallis test for comparisons of more than two groups. Univariate logistic regression was used to calculate ORs. A two-tailed *P* value < .05 was considered statistically significant.

Results

At the time of recruitment, median cough duration in the 104 children with PBB was 28 weeks (IQR, 6-57 weeks) and median cough score was 3 (IQR, 1-3), indicative of frequent coughing.¹⁷ The primary indications for bronchoscopy in the control subjects were stridor, cough, and other respiratory symptoms (Fig 1). Of the nine control subjects who reported cough on the day of bronchoscopy, the median duration was 6 weeks (IQR, 3-19 weeks). All children with PBB had a normal physical examination without evidence of chronic lung disease. Children with PBB were more likely to have attended childcare (Table 1).

Immune Function

Basic immune function test results in children with PBB were normal, except for median NK-cell levels (Table 2). All control data were within normal reported limits (not shown). Median peripheral blood eosino-phil counts were low in both groups (PBB group: 0.3×10^{9} /L [IQR, 0.1-0.6] vs control group: 0.3×10^{9} /L [IQR, 0.2-0.4]; normal range, 0.10-1.00). Median



FIGURE 1. Bar graph showing primary indication for bronchoscopy in control subjects. *Includes wheeze, exercise-induced symptoms, imaging abnormality, and hemoptysis.

total IgE levels, in 24 control subjects and 47 children with PBB, were normal (PBB group: 32 kU/L [IQR, 13-149 kU/L]; control group: 58 kU/L [IQR, 13-256]; normal range, < 100 kU/L). Of the 39 children with PBB who underwent RAST for common aeroallergens, 26% were positive for one or more allergens compared with 25% of 20 control subjects (P = .957).

Peripheral-blood lymphocyte subsets were tested in 46 children with PBB, of whom 33 (72%) had NK-cell levels above the upper limit of normal for age.²⁴ Median NK-cell levels (\times 10⁹/L) were higher in children with PBB than control subjects (PBB group: n = 46; median, 0.72 cells [IQR, 0.50-0.93 cells]; control group: n = 22; median, 0.53 cells [IQR, 0.39-0.89 cells]; P = .123). Blood NK-cell levels were more likely to be elevated above the normal for age in HAdV positive, compared with HAdV negative, participants (12 of 13 children vs 30 of 52, respectively; P = .023).

Bronchoscopic and BAL Findings

Of the 104 children with PBB, tracheo- and/or bronchomalacia were present in approximately two-thirds with similar rates in the control group (PBB group, n = 71 [68%] vs control group, n = 26 [53%]; P = .068). Laryngomalacia was more common in the control

Characteristic	PBB Group $(n = 104)$	Control Group $(n = 49)$	P Value ^a
Sex, male to female ratio, % male	72:32 (69)	33:16 (67)	.815
Age, mo (IQR), y	19 (12-30)	20 (8-63)	.967
Household tobacco smoke exposure	39 (38)	15 (33)	.565
Siblings, median (IQR)	1 (1-2)	1 (1-2)	.788
Childcare attendance	48 (91) (n = 53)	15(58)(n=26)	.001
Indigenous status ^b	10 (10)	1(2)	.106
Previous pneumonia	15(14)	8 (16)	.659
Wheeze (ever)	63 (81) (n = 78)	12 (67) (n = 18)	.192

Table 1-Background Characteristics of Children in the Cohort

Data given as No. (%) unless otherwise indicated. IQR = interquartile range.

 $^{a}P < .05$ considered statistically significant.

^bDefined as those who identified as aboriginal or Torres Strait Islander.

journal.publications.chestnet.org

Table 2—Basic	Immune	Function	Tests
---------------	--------	-----------------	-------

Immunologic Parameter	Median (IQR) $(n = 104)^a$	Normal Range
Immunoglobulins (g/L)		
IgG	7.1 (5.6-8.5)	3.0-13.0
IgA	0.5 (0.3-0.8)	0.4-1.4
IgM	0.9 (0.7-1.1)	0.5-1.6
IgG subclasses (g/L)		
IgG ₁	5.2 (4.1-6.7)	2.9-8.5
IgG ₂	0.8 (0.6-1.1)	0.45-2.6
IgG ₃	0.3 (0.2-0.4)	0.15-1.13
IgG_4	0.05 (0.02-0.2)	0.03-0.79
Lymphocyte subsets ($\times 10^{9}/L$) (n = 46)		
CD3 (T cells)	3.4 (2.7-4.4)	2.3-3.5
CD4+/CD3+ (helper T cells)	1.9 (1.4-2.6)	1.9-2.5
CD8+/CD3+ (cytotoxic T cells)	1.2(0.9-1.6)	0.35-2.5
CD 19 (total B cells)	1.4 (0.9-1.7)	0.43-3.3
CD56 and CD16 (NK cells)	$0.7 (0.5-0.9)^{b}$	0.05-0.52
Complement levels (hemolytic complement assay, CH50), U/mL	n = 70, 794 (725-874)	>520
Vaccine responses		
IgG Clostridium tetani, IU/mL	n = 86, 0.5 (0.2-1.1)	Protective level: >0.16 IU/mL
IgG Haemophilus influenza type Β, μg/mL	n = 75, 1.1 (0.3-4)	Short-term protection: $> 0.15 \ \mu g/mL$

IU = International Units; NK = natural killer. See Table 1 legend for expansion of other abbreviation.

^aWhere n < 104, patients were either recruited later in the study or insufficient specimen volume was available to perform the test. ^bParameters outside the normal range are denoted in bold.

group (PBB group, n = 10 [9.6%] vs control group, n = 15 [30.6%]; P = .001), likely explained by the fact that stridor was the major indication for bronchoscopy in control subjects (Fig 1).

Bacterial infection ($\geq 10^4$ colony-forming units [CFU]/mL) with common respiratory pathogens was present in BAL specimens of all patients with PBB (n = 104, 100%) and 19 control subjects (39%, P < .001). Nine children with PBB (9%), and four in the control group (8%) (P = .933) received antibiotics in the 24 h prior to bronchoscopy. H influenzae was the most common bacterial pathogen in both groups. A minority of strains were β -lactamase positive (PBB group: n = 17 [16%]; control group: n = 4 [8%]; P = .213]. Onehalf of the participants with PBB (n = 52, 50%) had two or more bacterial species at levels $\geq 10^4$ CFU/mL compared with four children in the control group (8%)(P < .001). Of those with PBB, most (n = 30; 56%) were found to be infected with *H* influenzae and Moraxella catarrhalis on culture. Staphylococcus aureus infection was more common in BAL specimens of control subjects than those with PBB (18% vs 11%, respectively; P = .19). Mycoplasma pneumonia was detected (on PCR) in a single child with PBB (<1%) and a single control participant (2%). Children with PBB were significantly more likely than control subjects to have coinfection with HAdV and H influenzae (PBB group: n = 22 [23%]; control group: n = 1 [2%]; P = .001). A random subset of children (n = 27 with PBB, n = 13 control subjects) had extended viral-panel analyses performed. In those with PBB, 18 of 27 samples (67%) were positive for any virus compared with five of 13 (38%) of control subjects (P = .066). Addi-

1274

tional viruses identified in children with PBB included human rhinovirus (n = 11, 41%), human bocavirus (n = 1, 4%), and HCoV (n = 1, 4%). Additional viruses identified in control subjects included human rhinovirus (n = 2, 15%), WUPyV (n = 1, 8%), and HCoV-C43 (n = 1, 8%). Table 3 presents results of univariate logistic regression for common lower-airway bacteria and viruses.

Median total cell counts were 270×10^{6} /L (IQR, 150×10^{6} /L to 520×10^{6} /L) in PBB vs 102×10^{6} /L (IQR, 47×10^{6} /L to 203×10^{6} /L) in control subjects (P < .001). The median percentage of neutrophils was markedly elevated at 42% (IQR, 19%-66%) in children with PBB compared with 4% (IQR, 2%-8%) in control subjects (P < .001) and higher in HAdV-positive participants (43%; IQR, 14%-55%) vs HAdV-negative participants (16%; IQR, 5%-55%) (P = .044). The median percentage of macrophages was correspondingly depressed at 45% (IQR, 24%-68%) in children with PBB compared with 90% (IQR, 80%-94%) in control subjects ($\tilde{P} < .001$). The percentages of lymphocytes were 9% (IQR, 5%-15%) and 6% (IQR, 3%-10%) in PBB and control groups, respectively (P = .033). The median percentage of eosinophils was 0% (IQR, 0%-0%) in both groups.

DISCUSSION

In this large prospective study on PBB in children, we advanced our understanding of PBB by further examining clinical, laboratory, and BAL descriptors in a large cohort of children. In addition to chronic

Table 3—Univariate Logistic Regression for Lower Airway Bacteria and Viruses

Infective Findings (BAL)	PBB Group, No. (%) $(n = 104)$	Control Group, No. (%) $(n\!=\!49)$	OR (95% CI)	P Value
Bacteria				
Haemophilus influenzae	75 (72)	10 (20)	10.09 (4.46-22.82)	<.001
Moraxella catarrhalis	45 (43)	5(10)	6.71 (2.46-18.30)	<.001
Streptococcus pneumoniae	41 (39)	8 (16)	3.34 (1.42-7.83)	.006
Viruses				
HAdV	22 (23)	2(4)	6.78(1.52-30.22)	.012
RSV	5 (5)	0 (0)		NS
HMPV	2 (2)	0 (0)		NS
IFV	2 (2)	2(4)	0.24 (0.02-2.71)	.248
HPIV	7 (7)	0 (0)		NS
Any virus	n = 92, 35 (38)	n = 45, 4 (9)	$6.30\ (2.08-19.09)$.001

HAdV = human adenovirus; HMPV = human metapneumovirus; HPIV = human parainfluenza virus; IFV = influenza virus; NS = not significant; RSV = respiratory syncytial virus.

wet cough in PBB, we observed high rates of parentreported wheeze, and an association between PBB and childcare attendance. Children with PBB demonstrated, within-population normal immunoglobulin levels, antibody responses (to tetanus and *H influenzae* type b vaccines) and propensity to atopy (IgE and RAST); however, many children had elevated NK-cell levels in blood specimens. Tracheobronchomalacia was common, but rates were similar in the control group. The BAL specimens from children with PBB showed a high prevalence of viruses, particularly HAdV, in addition to bacterial infection and airway neutrophilia.

Our study concurs with existing published studies on PBB regarding the demographics of these children. These factors include the young median age of children with PBB,⁷ the predominance of boys,¹⁰ and the coexistence of central airway anomalies (tracheoand/or bronchomalacia).¹⁰ Our new finding of high rates of childcare attendance is likely to have relevance to the rates of virus detection in our PBB cohort, notwithstanding the ongoing immune maturation also occurring at this age. Thus, boys with several months' history of wet cough, who attended childcare, and who were between their first and second birthdays typified our PBB cohort. Wheezing episodes were also commonly reported in children with PBB.

A high proportion of caregivers reported "wheeze ever" in their child (74%). This supports the findings of Saglani et al²⁵ of high rates of bacterial infection (43%) and neutrophilic inflammation (54%) in the BAL specimens of young children with severe recurrent wheeze, and those of Bisgaard et al²⁶ in their birth cohort study linking bacteria (in the naso- or hypopharynx) with wheeze. The similar rates of wheeze in children with PBB and the control subjects may relate to the high airway malacia rates also observed. However, it is important to acknowledge the inherent inaccuracy of parent-reported wheeze, which poorly reflects true wheeze and asthma^{27,28} and has low-level agreement with doctor assessment.^{29,30} This fact is likely to contribute to asthma overdiagnosis, particularly in primary care settings.

Coexistent asthma is unlikely to explain the high rates of parent-reported wheeze in children with PBB in the present study. This is supported by the fact that all children with PBB in this cohort had resolution of their cough with 2 weeks of antibiotic therapy (a criterion for PBB diagnosis). Our findings of low median IgE levels coupled with absence of eosinophilia (both in the lower airway and peripheral blood) provide further laboratory evidence against asthma.

Our previous descriptions of the immune status of children with PBB were limited to basic immunoglobulin levels.⁷ In the present study, we further evaluated immune function. We have demonstrated that antibody-mediated responses (to a protein-based [tetanus] and conjugate [*H influenzae* type b] vaccine) and lymphocyte subsets are normal in PBB, with the exception of elevated levels of NK cells CD16⁺ and CD56⁺. This is a novel finding, and likely to be associated with recent viral infection. Although the crosssectional nature of this study precludes making causal associations, it is probable that the high NK-cell levels seen in children with PBB were related to elevated virus detection rates in BAL specimens (38% of BAL specimens on a standard respiratory panel and 67% of BAL specimens on an extended viral panel). Moreover, the association between HAdV in BAL specimens and NK-cell elevation is consistent with published literature suggesting that NK cells play a role in innate immune defense against HAdV.³¹ Although we are unable to draw conclusions regarding the significance of HAdV in the airways of children with PBB, it deserves further research. A possible mechanism includes reactivation of HAdV in response to certain stresses, as HAdV can remain in a quiescent state in tonsillar and adenoidal tissue.32

In accordance with previous findings,³³ increased total cell counts, neutrophilia, and reduced percentage

of macrophages were observed in the BAL specimens of children with PBB. Notably, airway eosinophilia was absent. Previous research has linked viral-bacterial coinfection to significantly heightened neutrophil levels in the lower airways of children—higher than that of bacterial or viral infection alone.³⁴ This finding is likely to have relevance to our cohort with PBB, as high rates of viral-bacterial coinfection and greatly elevated percentages of neutrophils in BAL specimens (median, 42%) were apparent. Furthermore, HAdV was significantly more likely to be detected in the lower airways of children with PBB compared with control subjects and was associated with heightened levels of NK cells in blood specimens and with neutrophilia in the airways (indicating a systemic and airway immune response to the virus).

A link to chronic cough and bacterial bronchitis has been reported previously in airway malacias.¹⁰ A causal relationship (in either direction) may exist between large airway malacias and PBB to explain the high rates of tracheobronchomalacia (68%) in children with PBB. The similar tracheobronchomalacia rates in children with PBB and the control group (53%), and the high rates of bacterial infection (39%) observed in the control group are likely interrelated. These findings reflect the fact that children in the control group had chronic respiratory symptoms leading to bronchoscopy. Another possible explanation for the high rates of bacterial infection in the control group is contamination of the bronchoscope during the procedure. However, our fastidious BAL technique and use of a $\geq 10^4$ CFU/mL cutoff to indicate infection should minimize this likelihood.

A number of limitations of our study merit consideration. First, its cross-sectional nature precluded making causal links relating to the role of viruses in PBB pathogenesis. Second, due to the inherent ethical issues regarding use of control subjects for studies involving bronchoscopy, our comparator group comprised children with chronic respiratory symptoms (other than PBB). This likely reduced the strength of our observations. Third, we had missing data on childcare attendance and wheeze, as these questions were introduced later in the course of the study. Last, extended viral panel analyses were only performed on a small subset of participants, limiting our ability to make inferences regarding the significance of other viruses (eg, rhinovirus).

The major strengths of this study are its prospective design, the large cohort, and the inclusion of a control group. Our lower-airway findings confirm our previous observations in relation to PBB, with the addition of new information regarding environmental exposure (ie, childcare attendance, and virus and wheeze prevalence). Our findings illustrate the symptom overlap between PBB and asthma,^{7.9} highlighting

1276

the importance of clear-cut diagnostic criteria for PBB. The most convincing evidence for PBB in the children studied was a clear response to 2 weeks of amoxicillinclavulanate antibiotics—a key clinical factor in the diagnosis of PBB. While bronchoscopy has been undertaken in this study and included in our original description of PBB,³⁵ we do not advocate that all children with a chronic wet cough should undergo flexible bronchoscopy. In the Australian guidelines for chronic cough in children,¹⁵ the clinical definition of PBB does not include BAL criteria. The validity of this approach was documented in a recent, multicenter, randomized controlled trial on the management of chronic cough in children.³⁶

The importance of accurate PBB diagnosis and timely management extends beyond the significant financial and social repercussions of protracted cough in children.² Many authors postulate that recurrent episodes of PBB and/or untreated lower airway bacterial infection predispose to later development of bronchiectasis.³⁷⁻³⁹ Douros et al⁴⁰ described the association between duration of wet cough and abnormalities seen on CT scans. However, the question of whether recurrent PBB is antecedent to later development of bronchiectasis remains unanswered to date, and warrants elucidation in a longitudinal cohort study.

Our study's findings provide further depth to the clinical profile and our overall understanding of PBB. Children with PBB are, typically, very young boys with protracted wet cough and parent-reported wheeze, and without elevated IgE or eosinophilia, who have attended childcare. Elevated rates of viruses in BAL specimens (in particular, HAdV), and heightened levels of NK cells in blood provide both direct and indirect evidence for a role of viruses in PBB pathogenesis. In view of our findings, we postulate that interactions between the innate immune system, environmental exposures, and viruses and bacteria underlie the development of PBB in children, and further research is needed.

Acknowledgments

Author contributions: Dr Wurzel had full access to all of the data in the study and takes responsibility for the integrity of the data and the accuracy of the data analysis.

Dr Wurzel: contributed to conceptualization of the study, data analysis and collection, and manuscript preparation; read and approved the final manuscript; and served as principal author.

 \hat{Dr} Marchant: contributed to conceptualization of the study, data analysis, and manuscript preparation and read and approved the final manuscript.

Dr Yerkovich: contributed to preparation and critical review of the manuscript and read and approved the final manuscript.

Dr Upham: contributed to preparation and critical review of the manuscript and read and approved the final manuscript.

Dr Mackay: contributed to the extended viral panel analyses, critically reviewed the manuscript, and read and approved the final manuscript.

Dr Masters: contributed to data collection, critically reviewed the manuscript, and read and approved the final manuscript.

Dr Chang: contributed to conceptualization of the study, all aspects of the study, and manuscript preparation and read and approved the final manuscript.

Financial/nonfinancial disclosures: The authors have reported to *CHEST* the following conflicts of interest: Dr Wurzel is supported by Thoracic Society of Australia and New Zealand/Allen and Hanbury's, Queensland Children's Medical Research Institute and National Health and Medical Research Council (NHMRC) (scholarship 1039688). Dr Upham is a member of the Novartis Advisory Board; has received payment for lectures from AstraZeneca plc, GlaxoSmithKline plc, Novartis AG, and Boehringer Ingelheim GmbH; and is supported by NHMRC fellowship 511019. Dr Mackay has received payment for consultancy services from Firebrick Pharma Pty Ltd. Dr Chang is supported by NHMRC fellowship 1058213. Drs Marchant, Yerkovich, and Masters have reported to *CHEST* that no potential conflicts of interest exist with any companies/organizations whose products or services may be discussed in this article.

Role of sponsors: The sponsors had no role in the design of the study, the collection and analysis of the data, or the preparation of the manuscript.

Other contributions: We wish to thank the families that participated in this study. We also thank Helen Petsky, Sophie Anderson-James, Carol Willis, and Stacey Spencer for their assistance with study coordination, patient recruitment, and data entry. We also thank Claire Wang and Theo Sloots for laboratory components (extended viral panel analyses).

Additional information: The e-Appendix can be found in the "Supplemental Materials" area of the online article.

References

- 1. Faniran AO, Peat JK, Woolcock AJ. Persistent cough: is it asthma? Arch Dis Child. 1998;79(5):411-414.
- Marchant JM, Newcombe PA, Juniper EF, Sheffield JK, Stathis SL, Chang AB. What is the burden of chronic cough for families? *Chest*. 2008;134(2):303-309.
- Shields MD, Bush A, Everard ML, et al. BTS guidelines: Recommendations for the assessment and management of cough in children. *Thorax*. 2008;63(suppl 3):iii1-iii15.
- Gibson PG, Chang AB, Glasgow NJ, et al; CICADA. CICADA: Cough in children and adults: diagnosis and assessment. Australian cough guidelines summary statement. *Med J Aust.* 2010; 192(5):265-271.
- Irwin RS. Diagnosis and management of cough executive summary: ACCP evidence-based clinical practice guidelines. *Chest.* 2006;129(1_suppl):1S-23S.
- Chang AB, Robertson CF, Van Asperen PP, et al. A multicenter study on chronic cough in children: burden and etiologies based on a standardized management pathway. *Chest.* 2012;142(4):943-950.
- Marchant JM, Masters IB, Taylor SM, Cox NC, Seymour GJ, Chang AB. Evaluation and outcome of young children with chronic cough. *Chest*. 2006;129(5):1132-1141.
- de Jongste JC, Shields MD. Cough. 2: Chronic cough in children. *Thorax*. 2003;58(11):998-1003.
- Donnelly D, Critchlow A, Everard ML. Outcomes in children treated for persistent bacterial bronchitis. *Thorax*. 2007; 62(1):80-84.
- 10. Kompare M, Weinberger M. Protracted bacterial bronchitis in young children: association with airway malacia. *J Pediatr*. 2012;160(1):88-92.
- Zgherea D, Pagala S, Mendiratta M, Marcus MG, Shelov SP, Kazachkov M. Bronchoscopic findings in children with chronic wet cough. *Pediatrics*. 2012;129(2):e364-e369.
- Chang AB, Glomb WB. Guidelines for evaluating chronic cough in pediatrics: ACCP evidence-based clinical practice guidelines. *Chest.* 2006;129(1_suppl):260S-283S.

journal.publications.chestnet.org

- Goldsobel AB, Chipps BE. Cough in the pediatric population. J Pediatr. 2010;156(3):352-358.
- Leconte S, Paulus D, Degryse J. Prolonged cough in children: a summary of the Belgian primary care clinical guideline. *Prim Care Respir J.* 2008;17(4):206-211.
- Chang AB, Landau LI, Van Asperen PP, et al; Thoracic Society of Australia and New Zealand. Cough in children: definitions and clinical evaluation. *Med J Aust.* 2006;184(8): 398-403.
- Thomson M, Myer L, Zar HJ. The impact of pneumonia on development of chronic respiratory illness in childhood. *Pediatr Allergy Immunol Pulmonol*. 2010;23(4):279-290.
- Chang AB, Newman RG, Carlin JB, Phelan PD, Robertson CF. Subjective scoring of cough in children: parent-completed vs child-completed diary cards vs an objective method. *Eur Respir J.* 1998;11(2):462-466.
- Syrmis MW, Whiley DM, Thomas M, et al. A sensitive, specific, and cost-effective multiplex reverse transcriptase-PCR assay for the detection of seven common respiratory viruses in respiratory samples. J Mol Diagn. 2004;6(2):125-131.
- Bialasiewicz S, Whiley DM, Lambert SB, Gould A, Nissen MD, Sloots TP. Development and evaluation of real-time PCR assays for the detection of the newly identified KI and WU polyomaviruses. *J Clin Virol.* 2007;40(1):9-14.
- Tozer SJ, Lambert SB, Whiley DM, et al. Detection of human bocavirus in respiratory, fecal, and blood samples by real-time PCR. J Med Virol. 2009;81(3):488-493.
- Mackay IM, Arden KE, Speicher DJ, et al. Co-circulation of four human coronaviruses (HCoVs) in Queensland children with acute respiratory tract illnesses in 2004. *Viruses*. 2012; 4(4):637-653.
- Mackay IM. Human rhinoviruses. In: Schuller M, Sloots TP, James GS, Halliday CL, Carter IWJ, eds. PCR for Clinical Microbiology: An Australian and International Perspective. Dordrecht: Springer; 2010:299-301.
- Masters IB, Zimmerman PV, Chang AB. Longitudinal quantification of growth and changes in primary tracheobronchomalacia sites in children. *Pediatr Pulmonol.* 2007;42(10): 906-913.
- Kotylo PK, Fineberg NS, Freeman KS, Redmond NL, Charland C. Reference ranges for lymphocyte subsets in pediatric patients. Am J Clin Pathol. 1993;100(2):111-115.
- Saglani S, Nicholson AG, Scallan M, et al. Investigation of young children with severe recurrent wheeze: any clinical benefit? *Eur Respir J.* 2006;27(1):29-35.
- Bisgaard H, Hermansen MN, Bonnelykke K, et al. Association of bacteria and viruses with wheezy episodes in young children: prospective birth cohort study. *BMJ*. 2010;341: c4978.
- 27. Elphick HE, Sherlock P, Foxall G, et al. Survey of respiratory sounds in infants. *Arch Dis Child*. 2001;84(1):35-39.
- Mehrotra A, Lasley MV. Early rattles, purrs and whistles as predictors of later wheeze. *Pediatrics*. 2008;93(8):701-704.
- Cane RS, Ranganathan SC, McKenzie SA. What do parents of wheezy children understand by "wheeze"? Arch Dis Child. 2000;82(4):327-332.
- Cane RS, McKenzie SA. Parents' interpretations of children's respiratory symptoms on video. Arch Dis Child. 2001;84(1): 31-34.
- Tomasec P, Wang EC, Groh V, et al. Adenovirus vector delivery stimulates natural killer cell recognition. J Gen Virol. 2007;88(pt 4):1103-1108.
- Alkhalaf MA, Guiver M, Cooper RJ. Prevalence and quantitation of adenovirus DNA from human tonsil and adenoid tissues. J Med Virol. 2013;85(11):1947-1954.
- Marchant JM, Gibson PG, Grissell TV, Timmins NL, Masters IB, Chang AB. Prospective assessment of protracted bacterial

bronchitis: airway inflammation and innate immune activation. *Pediatr Pulmonol.* 2008;43(11):1092-1099.

- Wurzel DF, Marchant JM, Clark JE, et al. Wet cough in children: Infective and inflammatory characteristics in bronchoalveolar lavage fluid [published online ahead of print June 20, 2013]. *Pediatr Pulmonol*. doi:10.1002/ppul.22792.
- Marchant JM, Masters IB, Taylor SM, Chang AB. Utility of signs and symptoms of chronic cough in predicting specific cause in children. *Thorax*. 2006;61(8):694-698.
- Chang AB, Robertson CF, van Asperen PP, et al. A cough algorithm for chronic cough in children: a multicenter, randomized controlled study. *Pediatrics*. 2013;131(5):e1576-e1583.
- 37. Chang AB, Redding GJ, Everard ML. Chronic wet cough: protracted bronchitis, chronic suppurative lung disease and bronchiectasis. *Pediatr Pulmonol*. 2008;43(6):519-531.
- Field CE. Bronchiectasis in childhood; prophylaxis, treatment and progress with a follow-up study of 202 cases of established bronchiectasis. *Pediatrics*. 1949;4(3):355-372.
- Eastham KM, Fall AJ, Mitchell L, Spencer DA. The need to redefine non-cystic fibrosis bronchiectasis in childhood. *Thorax*. 2004;59(4):324-327.
- Douros K, Alexopoulou E, Nicopoulou A, et al. Bronchoscopic and high-resolution CT scan findings in children with chronic wet cough. *Chest.* 2011;140(2):317-323.

CHAPTER 6

6.1 INTRODUCTION TO CHAPTER 6

In Chapter 5, we further expanded our understanding of PBB by describing the typical child who presents with this condition. These findings will assist the clinician in the more accurate diagnosis of PBB and assist in its differentiation from other common causes of chronic cough. Further research to evaluate the longer-term significance of PBB, with respect to risk of bronchiectasis, was needed. Availability of such data is relevant when counselling parents of children who experience recurrent episodes of PBB.

Chapter 6 explored the significance of PBB with respect to a later diagnosis of bronchiectasis. It evaluated the accuracy of the hypothesis that PBB and bronchiectasis rest upon a common clinical spectrum of disease severity.

In this study, we followed a cohort of children with PBB over a two-year period to determine the proportion later diagnosed with bronchiectasis. We also examined risk factors for bronchiectasis. Findings from this study would aid the clinician in identifying those children with PBB at greatest risk of bronchiectasis. The relevance of these findings is that they would enable closer monitoring of children at risk of bronchiectasis hence leading to earlier diagnosis.

6.2 Protracted Bacterial Bronchitis in Children: Natural History and Risk Factors for Bronchiectasis at 2 years

Protracted Bacterial Bronchitis in Children: Natural History and Risk Factors for Bronchiectasis

Corresponding author: Danielle Wurzel. Royal Children's Hospital, Melbourne, Victoria 3052, Australia. Email: danielle.wurzel@rch.org.au, telephone: +61 421 646 059, fax number: +61 3 9345 9154.

Danielle F Wurzel,MBBS <u>danielle.wurzel@rch.org.au</u>, ^{1,2} Julie M Marchant,PhD <u>j.marchant@uq.edu.au</u>, ^{1,3} Stephanie T Yerkovich,PhD <u>s.yerkovich@uq.edu.au</u>, ^{5,6} John W Upham,PhD, <u>j.upham@uq.edu.au</u>, ⁶ Helen L Petsky,PhD <u>helen.petsky@qut.edu.au</u>, ⁴ Heidi Smith-Vaughan,PhD, <u>heidi.smith-vaughan@menzies.edu.au</u>, ⁷ Brent Masters,PhD, <u>brent_masters@health.qld.gov.au</u>, ^{1,3} Helen Buntain,PhD <u>buntainh@me.com</u>, ^{3,4} Anne B Chang.PhD anne.chang@menzies.edu.au ^{3,4,7}

¹Queensland Children's Medical Research Institute, The University of Queensland, ²The Royal Children's Hospital, Melbourne, ³Respiratory & Sleep Medicine, Lady Cilento Children's Hospital, Brisbane, ⁴Queensland Children's Medical Research Institute, Queensland University of Technology, ⁵Queensland Lung Transplant Service, Prince Charles Hospital, Brisbane, ⁶School of Medicine, The University of Queensland, ⁷Child Health Division, Menzies School of Health Research, Charles Darwin University, Darwin, Australia.

Running title: Risk factors for bronchiectasis in children with PBB

Key words: bacterial infection, bronchiectasis, paediatric lung disease, respiratory infection, viral infection

Word count: 2717

FUNDING

This work was supported by the National Health and Medical Research Council [project grant 1042601 and Centre of Research Excellence grant 1040830] and the Financial Markets Foundation for Children [project grant 2010-005]. DW is supported by scholarships from the Thoracic Society of Australia and New Zealand/Allen and Hanbury's, Queensland Children's Medical Research Institute and National Health and Medical Research Council [1039688]. AC is supported by a NHMRC practitioner fellowship [grant 1058213]. The views expressed in this publication are those of the authors and do not reflect the views of the NHMRC. The funders played no role in the conduct of the study or preparation of the manuscript. The authors declare no conflicts of interest in relation to this manuscript.

ABSTRACT

Importance: Protracted bacterial bronchitis (PBB) and bronchiectasis are distinct diagnostic entities sharing common clinical and laboratory features. It is postulated, but remains unproven, that PBB precedes bronchiectasis in a subset of children.

Objective: In a cohort of children with PBB, our objectives were to: (a) determine the mediumterm risk of bronchiectasis and (b) identify risk factors for bronchiectasis and recurrent episodes of PBB.

Design: Between March 2008 and October 2012, children undergoing flexible bronchoscopy and BAL for a clinical indication were recruited to this longitudinal cohort study. All had basic immune function testing. Caregivers completed clinical questionnaires at baseline and cough diaries during periods of illness. CT chest was undertaken if they developed clinical features suggestive of bronchiectasis.

Setting: The study was undertaken in a large, tertiary paediatric hospital in Brisbane, Australia.

Participants: Of 343 children undergoing bronchoscopy, 161 met criteria for PBB including: i) history of chronic (\geq 4 weeks) wet cough, (ii) response to 2 weeks of treatment with amoxicillinclavulanic acid and (iii) absence of clinical pointers suggesting an alternative cause for cough. In total, 106 children reached the 2-year follow-up time-point (13 withdrew, 42 had incomplete cough diaries).

Main outcome and measures: The primary study outcome was the proportion of children with

PBB, at study entry, diagnosed with bronchiectasis during the follow-up period.

Results: Of 161 children with PBB (66% male), 13 (8.1%) were diagnosed with bronchiectasis. Almost half (43.5%) with PBB had recurrent episodes (>3/year). Major risk factors for bronchiectasis included: *H. influenzae* lower airway infection (in BAL) (p=0.013) and recurrent episodes of PBB (p=0.003). *H. influenzae* infection conferred >7 times higher risk of bronchiectasis [HR 7.55 (95%CI 1.66 - 34.28), p=0.009] compared to absence of *H. influenzae*. The majority of isolates (82%) were non-typeable *H. influenzae*. No risk factors for recurrent PBB were identified.

Conclusion: PBB is antecedent to a diagnosis of bronchiectasis in a subgroup of children. *H. influenzae* lower airway infection and recurrent PBB are significant risk factors. Clinicians should be cognisant of the need to monitor children with PBB over time for evidence of bronchiectasis, particularly those with risk factors.

At a glance:

- Protracted bacterial bronchitis (PBB), a major cause of chronic cough in children, may be associated with later development of bronchiectasis.
- In this study, we have shown that approximately 1 in 12 children with PBB are diagnosed with bronchiectasis at 2-year follow-up.
- Risk factors for bronchiectasis include: lower airway infection with *H. influenzae* and recurrent episodes of protracted bacterial bronchitis.
- Children with PBB should be monitored over time for bronchiectasis and specialist

referral should be considered in those with risk factors.

INTRODUCTION

Protracted bacterial bronchitis (PBB), first described by our group in 2006, is a major cause of chronic cough in children who present to respiratory specialists.^{1,2} PBB is now well recognised in Australia,³ Europe ⁴ and the US.^{5,6} It is characterised by persistent wet cough, response to a 2-week course of appropriate antibiotics and absence of indicators to suggest an alternative cause for cough.^{1,7} PBB is more common in young boys and children who have attended childcare.⁸ When compared to controls, children with PBB are more likely to have lower airway infection with bacteria, such as *H. influenzae, M. catarrhalis* and *S. pneumoniae,* as well as viruses e.g. adenovirus.⁸

Currently, there are limited published data,⁹ and no prospective follow-up studies, evaluating the outcomes of children with PBB. This clinical gap limits the clinician's ability to prognosticate on the likely natural history of PBB in an individual child. Anecdotally, many otherwise healthy children experience recurrent episodes of PBB without long-term consequences. However, in a subgroup of children, recurrent PBB may be an important risk factor for, or herald the onset of, non-cystic fibrosis bronchiectasis, though this remains unproven. Certain characteristics of the lower airway microbiota may also predispose to the development of chronic lung disease.

PBB and bronchiectasis share many common features, spanning from respiratory symptoms (i.e. chronic wet cough) to intense neutrophilic lower airway inflammation and innate immune system activation.¹⁰⁻¹² The lower airway microbiota,¹³ including presence of adenovirus type C¹⁴ and predominance of nontypeable *Haemophilus influenzae* (NTHi),^{1,8} are also alike. Such similarities underpin the notion that PBB and bronchiectasis occupy distinct poles of a common clinical continuum.¹⁵ To date, the accuracy of this proposed continuum is uncertain, and warrants

4

evaluation in a prospective cohort study. If it is indeed correct, then risk factors for bronchiectasis, in children with PBB, warrant elucidation in order to further inform clinical practice and potentially determine intervention points.

Hence, in 161 children with PBB and 25 controls, we aimed to determine: (a) the 2-year outcomes with respect to development of bronchiectasis and; (b) the risk factors for diagnosis of bronchiectasis and those for recurrent episodes of PBB.

METHODS

Study participants

Participants were enrolled as part of a larger prospective cohort study aimed at evaluating the long-term outcome of children with chronic cough. Written informed consent was obtained from all parents/guardians and ethics approval granted by The Queensland Children's Health Services (RCH) Human Research Ethics Committee (HREC/03/QRCH/17).

Between March 2008 and October 2012, 343 children were enrolled. Of these, 161 fulfilled criteria for PBB and 25 were recruited as control participants (15 undergoing evaluation for respiratory symptoms other than chronic cough, and 10 healthy controls). Data from the 15 control children undergoing respiratory evaluation have previously been described ⁸ (see e-supplement). The 10 healthy controls were recruited from colleagues and friends.

All children (excluding the 10 healthy controls) underwent flexible bronchoscopy and bronchoalveolar lavage (BAL) as per clinical indication, and were recruited prior to their bronchoscopy. BALs were processed for cellularity and microbiology. Bacterial infection was defined as bacterial load of $\geq 10^4$ colony-forming units (cfu)/ml BAL.^{8,11} Laboratory tests for suppurative lung diseases were performed, as described previously.¹⁴ *H. influenzae* characterisation was undertaken at a research laboratory (Menzies School of Child Health Research, Darwin) when BAL was available (see e-supplement).

Follow up

Follow-up included monthly contacts (phone-calls or emails) to capture respiratory exacerbations (undertaken by research nurses). Parents completed cough diaries during periods of illness. The majority of children were seen by their pediatric pulmonologist, as part of routine clinical follow-up, 3-4 monthly. At 2-year follow-up, a subset of children (n=106) also underwent clinical assessment (by DW and/or AC) for bronchiectasis.

Given the ethical considerations pertaining to research-related CT chest in children,^{16,17} participants underwent CT scans only when clinical features of bronchiectasis were present.¹⁸ Clinicians had similar practices whereby CT chest was undertaken for: (a) chronic wet cough non-responsive to 4 weeks of antibiotic therapy;¹⁹ (b) persistent chest radiographic changes despite appropriate antibiotic therapy or (c) recurrent hospitalisations for acute respiratory events.

Definitions

PBB was defined as: (i) history of chronic (\geq 4 weeks) wet cough, (ii) prospective evidence (supported by cough diaries) of response to 2 weeks of treatment with amoxicillin-clavulanate and (iii) absence of clinical pointers suggesting an alternative cause for cough.¹ Bronchiectasis diagnosis was based on pediatric radiological (CT) criteria,²⁰ in the context of a child having clinical symptoms of bronchiectasis. Recurrent PBB was defined as >3 episodes of PBB in the first year after enrolment.

Statistical analyses

Descriptive statistics were utilized to summarize demographic and clinical characteristics. Median and inter-quartile ranges were reported as data were non-normally distributed. Pearson's chi-square (or Fisher's exact test) was used for categorical and Mann-Whitney U for continuous variables. Logistic regression was employed to calculate odds ratios (OR) and 95% confidence intervals (95%CI). A survival analysis (using Cox's proportional hazards regression) examined the relationship between time to diagnosis of, and risk factors for, bronchiectasis. A 2-tailed p-value of <0.05 was considered statistically significant. Statistical analyses were performed using SPSS (version 21.0, Armonk, NY: IBM Corp., USA) and STATA (version 12.1, StataCorp, Texas, USA).

RESULTS

Clinical and demographic characteristics of participants

Of the 161 children with PBB, 106 completed the 2-year follow-up period and had clinician assessment for bronchiectasis (Figure 1). The median duration of follow-up was 25 months (IQR 24, 28) in children with PBB and 27 months (IQR 26, 29) in controls undergoing bronchoscopy.

There was no difference between children who completed the 2-year follow-up and those who did not (Table 1), with regards to: length of cough at recruitment (p=0.807), *H. influenzae* status (p=0.518) or proportion with 2+ siblings (p=0.260). However, those completing the 2-year follow-up (n=106) were significantly more likely to have had recurrent PBB (p=0.003). Thus, to reduce the potential for bias, 161 was used as the denominator in subsequent analyses. The 55

additional children who did not complete 2-year follow-up were assumed to be bronchiectasisfree at the 2-year time-point. Comparison of groups at baseline showed that children with PBB had a higher burden of doctor visits in the preceding 12 months, as compared to controls (Table 1). At 2-year follow-up, children with PBB were more likely to be coughing compared to controls (p=0.005) and to have had parent-reported wheeze in the preceding 12 months (p=0.001) (Table 2). Of the 161 children with PBB, 154 had completed cough diaries up until at least the 1year time-point. Of these, 67 (43.5%) had recurrent PBB (>3 episodes/year).

Bronchiectasis diagnoses on chest CT scan

Multi-detector CT with HRCT reconstruction was performed in 25 of 161 children with PBB. Radiological evidence of bronchiectasis was present in 13 (8.1%). In those with bronchiectasis, CT was performed at median duration of 9 months (IQR 4, 19) post recruitment. The median age of these children was 38 months (IQR 27, 58). Compared to those that did not undergo CT, children with PBB who underwent CT chest, had similar rates of lower airway infection with *H. influenzae* (p=0.142) and *S. pneumoniae* (p=0.135), however, those undergoing CT scan were less likely to have *M. catarrhalis* infection, than those that did not undergo CT (34.2% vs 17%, p=0.029) (this is likely to have been a chance occurrence.) Similar rates of 'recurrent PBB' status were recorded in those children that underwent CT and those that did not (p=0.118).

No children in the control group developed clinical features of bronchiectasis. Of the control children, 3 had CT scan performed for other indications. None had radiological evidence of bronchiectasis.

Risk factors for bronchiectasis and recurrent PBB
Univariate analysis showed that bronchiectasis was significantly more likely to be diagnosed during follow up when the child had recurrent PBB, when *H. influenzae* infection was present (i.e. cultured at a clinically significant level of $\geq 10^4$ cfu/ml in the BAL) or in children with two or more siblings (table 3). There was no significant difference between groups for other factors.

Multivariate logistic regression showed that recurrent PBB status and *H. influenzae* infection were significantly associated with bronchiectasis diagnosis [OR 11.48 (95%CI 2.33-56.50) p=0.003 and OR 7.60 (95%CI 1.53-37.79), p=0.013, respectively], whereas having ≥ 2 siblings was no longer significant [OR 3.53 (0.98, 12.70), p=0.054]. Survival analysis, using Cox regression, concurred with logistic regression findings. Participants with *H. influenzae* lower airway infection were > 7 times more likely to be diagnosed with bronchiectasis per month of follow-up, compared to those without *H. influenzae*, and recurrent PBB status conferred >9 times greater risk of bronchiectasis diagnosis per month (Table 4).

The percentage of lower airway neutrophils was similar in children with and without a subsequent diagnosis of bronchiectasis (Table 5). Although the percentage of eosinophils in BAL was slightly higher in those who developed bronchiectasis, the median eosinophil count was within the normal range (i.e. <2.5%),²¹ and hence considered clinically insignificant. Further, the difference was no longer observed when the major outlier was removed.

In the examination of predictors for recurrent PBB, univariate and multivariate analysis showed no significant difference between children with and without recurrent PBB for the factors examined (sex, age, prior pneumonia, tobacco smoke exposure, maternal tobacco smoking in pregnancy, Indigenous status, number of children in household, childcare attendance and BAL cellularity and microbiology) (eTable 1). Notably, *H. influenzae* infection in the BAL was not predictive of recurrent PBB.

H. influenzae typing

Thirty-four of 55 (62%) *H. influenzae* positive samples were available for further characterisation in the research laboratory (laboratory methods in e-supplement). Of these, *H. influenzae* from the majority of samples (n=28; 82%) were identified as NTHi; encapsulated *H. influenzae* were not identified. *H. influenzae* from the remaining 6 samples were reassigned as *H. haemolyticus* following species-specific PCR.

DISCUSSION

This is the first prospective longitudinal cohort study of children with PBB. In our cohort, based in a major teaching hospital, almost half (43.5%) had recurrent episodes (>3 in the first year after recruitment) and approximately 1 in 12 were diagnosed with bronchiectasis at 2-year follow-up. We have identified 2 significant risk factors for bronchiectasis: recurrent (>3 /year) episodes of PBB and presence of *H. influenzae* infection of the lower airways. Further, *H. influenzae* in the BAL conferred >7 times higher risk of bronchiectasis diagnosis, compared to *H. influenzae* negative state.

Findings from this study, the first to investigate the 2-year outcomes of children with PBB, support the hypothesis that PBB and bronchiectasis likely represent a clinical spectrum. Children with PBB have endobronchial bacterial infection and neutrophilic airway inflammation, factors known to be injurious to the airways.²² Our findings suggest that, in a subset of children receiving close follow-up by pediatric pulmonologists, recurrent episodes (>3 per year) of PBB precede a diagnosis of bronchiectasis. Further, although a cause-effect relationship cannot be concluded, a relationship between *H. influenzae* infection and later diagnosis of bronchiectasis has been shown. This finding is in accordance with increasing recognition, amongst researchers and clinicians, of the role of *H. influenzae* in chronic respiratory disease pathogenesis.²³

H. influenzae is the most common bacterial species infecting the lower airways of children with chronic endobronchial suppuration, including PBB,⁸ recurrent or non-responsive community-acquired pneumonia ²⁴ and bronchiectasis.^{8,10,25} It is also the major bacterial pathogen associated with chronic respiratory disorders in adults e.g. bronchiectasis and COPD.^{23,26,27} The exact role of NTHi in bronchiectasis pathogenesis, however, remains unclear. It may represent the primary

insult, or alternatively, a secondary phenomenon whereby damaged and/or dilated airways act as a nidus for *H. influenzae*. Our finding that *H. influenzae* infection (predominantly NTHi on typing in this study) may be a risk factor for chronic respiratory diseases such as bronchiectasis is consistent with current literature.^{25,28-31} Nevertheless, there are likely to be other contributory factors.

Our earlier studies identified that children with PBB were significantly more likely, than controls, to have attended childcare, and to have viral infection of the lower airways, particularly with adenovirus type C.^{8,14} It is plausible, although beyond the scope of this study, that co-infection with specific microbial pathogens e.g. adenovirus C and *H. influenzae*,³² initiates the vicious cycle of lower airway infection and inflammation inherent to these conditions.³³ Factors such as young age (and concomitant immune system and respiratory tract immaturity) at initial infection, are likely to play a key role in the aetio-pathogenesis of these disorders.¹⁴

In this study, bronchiectasis was diagnosed at an early age (median 38 months), similar to our other cohorts of children with bronchiectasis.^{10,19,34} Together with findings from our previous studies on PBB,^{8,14} this observation raises the question of whether timing of airway infection (with bacteria and/or viruses), with respect to age, influences the likelihood of later diagnosis of bronchiectasis.

In support of previous studies on PBB, our present study showed that PBB is characterised by active neutrophilic inflammation of the lower airways. Although the median percentage of neutrophils in BAL was greater in those with PBB who were subsequently diagnosed with bronchiectasis, compared to those who were not (median 35 vs 25, respectively), this difference

12

was not statistically significant. This suggested that the degree of airway neutrophilia was not a predictor of subsequent bronchiectasis. Regarding the inter-group disparity in airway eosinophil levels observed between those with and without bronchiectasis, this is unlikely to be of clinical relevance for two reasons. First, eosinophil levels in both groups were within normal reported ranges, and second, when the major outlier (16% eosinophils in a child with features consistent with persistent asthma in addition to PBB) was removed from the 'bronchiectasis present' group, this difference was no longer significant.

Several limitations to this study merit discussion. Firstly, we did not universally perform CT scans on all children at study entry and exit. In view of the potential increased lifetime risk of developing cancer and the relatively common nature of PBB in otherwise well children,³⁵ we could not justify exposing all children to the radiation dose associated with 2 CT scans.¹⁷ The impact of this is two-fold. First, bronchiectasis may have been missed in some children at recruitment and/or follow-up. Hence, we cannot conclude that PBB progresses to bronchiectasis per se. Rather, our findings indicate that, in children with PBB (who are followed closely by pulmonologists over a 2-year period) a proportion (approximately 8%) will be diagnosed with bronchiectasis on CT chest. Second, we acknowledge the fact that performing CT scan only in a select subgroup introduces the possibility of selection bias in that individual clinicians may have different indications for performing CT chest. To address this, we compared children with PBB undergoing CT chest to those who did not and found no significant differences between groups with respect to bronchiectasis risk factors.

The major strengths of this study are the longitudinal nature of data collection and the inclusion of lower airway inflammatory and microbiological findings. However, as we only performed

13

bronchoscopy and BAL at a single time-point, the temporal relationship between infection with *H. influenzae* and development of bronchiectasis remains unclear. This does not, however, diminish the utility nor the relevance of our findings. Irrespective of whether *H. influenzae* is a cause (or consequence) of bronchiectasis in young children, its presence in a child with recurrent PBB should alert the pulmonologist to the increased possibility of bronchiectasis, suggesting the need for closer monitoring and/or further investigation. This is a novel finding with direct clinical relevance to those managing children with chronic cough.

Findings from this longitudinal cohort study provide further evidence for the link between PBB and bronchiectasis in young children. We have shown that at least 8% (1 in 12.4 children, median age 23 months at recruitment) with PBB are diagnosed with bronchiectasis at 2-year follow-up. Of those with PBB, at least 43.5% had recurrent episodes in their first year after recruitment. We have identified key risk factors for bronchiectasis i.e. *H. influenzae* lower airway infection and recurrent episodes of PBB.

In view of this study's findings, clinicians should be cognisant of the need to monitor children with PBB over time and consider CT chest in those with risk factors for bronchiectasis. Ongoing research into the potential role of *H. influenzae* in the pathogenesis of bronchiectasis in children is needed in order to inform future preventative and therapeutic strategies.

ACKNOWLEDGEMENTS

We wish to thank the families that participated in this study. We also thank the Cough and Asthma Airways Research Group (CAARG) research team at the Department of Respiratory and Sleep medicine, Lady Cilento Children's Hospital (formerly The Royal Children's Hospital), Brisbane for their invaluable contribution including Sophie Anderson-James BN (data collection, patient recruitment, study coordination), Carol Willis (data entry), Sandra Goodwin (data entry), Joanne Tuppin BN (data collection and patient recruitment) and Samantha Gardiner BN (data collection and patient recruitment). We also thank Jemima Beissbarth BSc (Menzies School of Health Research, Charles Darwin University, Darwin, Australia) for her laboratory work in processing samples for NTHi data. DF Wurzel had full access to all of the data in the study and takes responsibility for the integrity of the data and the accuracy of the data analysis.

CONFLICTS OF INTEREST

None

FUNDING

This work was supported by the National Health and Medical Research Council [project grant 1042601 and Centre of Research Excellence grant 1040830] and the Financial Markets Foundation for Children [project grant 2010-005]. DW is supported by scholarships from the Thoracic Society of Australia and New Zealand/Allen and Hanbury's, Queensland Children's Medical Research Institute and National Health and Medical Research Council [1039688]. AC is supported by a NHMRC practitioner fellowship [grant 1058213]. The views expressed in this publication are those of the authors and do not reflect the views of the NHMRC.

REFERENCES

- 1. Marchant JM, Masters IB, Taylor SM, Cox NC, Seymour GJ, Chang AB. Evaluation and outcome of young children with chronic cough. *Chest.* Jun 01 2006;129(5):1132-1141.
- Chang AB, Robertson CF, van Asperen PP, et al. A cough algorithm for chronic cough in children: a multicenter, randomized controlled study. *Pediatrics*. May 2013;131(5):e1576-1583.
- Gibson PG, Chang AB, Glasgow NJ, Holmes PW, Landau LI, Newcombe P. CICADA: Cough in Children and Adults: Diagnosis and Assessment. Australian cough guidelines summary statement. *The Medical journal of Australia*. Apr 01 2010;192(5):265-271.
- Dinwiddie R. Anatomy and development of the respiratory system. In: Eber E, Midulla F, eds. *ERS handbook: Paediatric Respiratory Medicine:*. Sheffield, UK: The European Respiratory Society; 2013:1-10.
- Kompare M, Weinberger M. Protracted Bacterial Bronchitis in Young Children: Association with Airway Malacia. *The Journal of pediatrics*. Feb 01 2012;160(1):88-92.
- Zgherea D, Pagala S, Mendiratta M, Marcus MG, Shelov SP, Kazachkov M. Bronchoscopic findings in children with chronic wet cough. *Pediatrics*. Feb 2012;129(2):e364-369.
- Shields MD, Bush A, Everard ML, McKenzie S, Primhak R, Group BTSCG. BTS guidelines: Recommendations for the assessment and management of cough in children. *Thorax.* Vol 63 2008:iii1-iii15.
- 8. Wurzel DF, Marchant JM, Yerkovich ST, et al. Prospective characterization of protracted bacterial bronchitis in children. *Chest.* Jun 1 2014;145(6):1271-1278.
- 9. Pritchard MG, Lenney W, Gilchrist FJ. Outcomes in children with protracted bacterial bronchitis confirmed by bronchoscopy. *Arch. Dis. Child.* Jan 2015;100(1):112.

- Kapur N, Grimwood K, Masters IB, Morris PS, Chang AB. Lower airway microbiology and cellularity in children with newly diagnosed non-CF bronchiectasis. *Pediatr. Pulmonol.* Mar 2012;47(3):300-307.
- Marchant JM, Gibson PG, Grissell TV, Timmins NL, Masters IB, Chang AB. Prospective assessment of protracted bacterial bronchitis: airway inflammation and innate immune activation. *Pediatr. Pulmonol.* Nov 2008;43(11):1092-1099.
- 12. Simpson JL, Grissell TV, Douwes J, Scott RJ, Boyle MJ, Gibson PG. Innate immune activation in neutrophilic asthma and bronchiectasis. *Thorax*. Mar 2007;62(3):211-218.
- van der Gast CJ, Cuthbertson L, Rogers GB, et al. Three clinically distinct chronic pediatric airway infections share a common core microbiota. *Annals of the American Thoracic Society.* Sep 2014;11(7):1039-1048.
- Wurzel DF, Mackay IM, Marchant JM, et al. Adenovirus Species C Is Associated With Chronic Suppurative Lung Diseases in Children. *Clin. Infect. Dis.* May 9 2014;59(1):34-40.
- 15. Chang AB, Redding GJ, Everard ML. Chronic wet cough: Protracted bronchitis, chronic suppurative lung disease and bronchiectasis. *Pediatr. Pulmonol.* Jun 2008;43(6):519-531.
- Mathews JD, Forsythe AV, Brady Z, et al. Cancer risk in 680,000 people exposed to computed tomography scans in childhood or adolescence: data linkage study of 11 million Australians. *BMJ*. 2013;346:f2360.
- Pearce MS, Salotti JA, Little MP, et al. Radiation exposure from CT scans in childhood and subsequent risk of leukaemia and brain tumours: a retrospective cohort study. *Lancet*. Aug 4 2012;380(9840):499-505.

- Chang AB, Bell SC, Byrnes CA, et al. Chronic suppurative lung disease and bronchiectasis in children and adults in Australia and New Zealand. *Med. J. Aust.* Sep 20 2010;193(6):356-365.
- Goyal V, Grimwood K, Marchant J, Masters IB, Chang AB. Does failed chronic wet cough response to antibiotics predict bronchiectasis? *Arch. Dis. Child.* Jun 2014;99(6):522-525.
- Kapur N, Masel JP, Watson D, Masters IB, Chang AB. Bronchoarterial ratio on highresolution CT scan of the chest in children without pulmonary pathology: need to redefine bronchial dilatation. *Chest.* Jul 2011;139(6):1445-1450.
- Pizzutto SJ, Grimwood K, Bauert P, et al. Bronchoscopy contributes to the clinical management of indigenous children newly diagnosed with bronchiectasis. *Pediatr. Pulmonol.* Jan 2013;48(1):67-73.
- 22. Stockley RA. Lung infections. 1. Role of bacteria in the pathogenesis and progression of acute and chronic lung infection. *Thorax.* Jan 1998;53(1):58-62.
- Van Eldere J, Slack MP, Ladhani S, Cripps AW. Non-typeable Haemophilus influenzae, an under-recognised pathogen. *Lancet Infect. Dis.* Dec 2014;14(12):1281-1292.
- 24. De Schutter I, De Wachter E, Crokaert F, et al. Microbiology of bronchoalveolar lavage fluid in children with acute nonresponding or recurrent community-acquired pneumonia: identification of nontypeable Haemophilus influenzae as a major pathogen. *Clin. Infect. Dis.* Jun 15 2011;52(12):1437-1444.
- 25. Hare KM, Binks MJ, Grimwood K, Chang AB, Leach AJ, Smith-Vaughan H. Culture and PCR detection of Haemophilus influenzae and Haemophilus haemolyticus in Australian Indigenous children with bronchiectasis. J. Clin. Microbiol. Jul 2012;50(7):2444-2445.

- 26. Thanavala Y, Lugade AA. Role of nontypeable Haemophilus influenzae in otitis media and chronic obstructive pulmonary disease. *Adv. Otorhinolaryngol.* 2011;72:170-175.
- 27. King P. Haemophilus influenzae and the lung (Haemophilus and the lung). *Clinical and translational medicine*. 2012;1(1):10.
- 28. Sethi S, Murphy TF. Bacterial infection in chronic obstructive pulmonary disease in 2000: a state-of-the-art review. *Clin. Microbiol. Rev.* Apr 2001;14(2):336-363.
- 29. Butt HL, Clancy RL, Cripps AW, et al. Bacterial colonisation of the respiratory tract in chronic bronchitis. *Aust. N. Z. J. Med.* Feb 1990;20(1):35-38.
- Clancy RL, Dunkley M. Acute exacerbations in COPD and their control with oral immunization with non-typeable haemophilus influenzae. *Frontiers in immunology*. 2011;2:7.
- Bilton D, Pye A, Johnson MM, et al. The isolation and characterization of non-typeable
 Haemophilus influenzae from the sputum of adult cystic fibrosis patients. *Eur. Respir. J.*Jun 1995;8(6):948-953.
- Suzuki K, Bakaletz LO. Synergistic effect of adenovirus type 1 and nontypeable
 Haemophilus influenzae in a chinchilla model of experimental otitis media. *Infect. Immun.* May 1 1994;62(5):1710-1718.
- Cole PJ. Inflammation: a two-edged sword--the model of bronchiectasis. *Eur. J. Respir. Dis. Suppl.* 1986;147:6-15.
- Kapur N, Masters IB, Chang AB. Longitudinal growth and lung function in pediatric noncystic fibrosis bronchiectasis: what influences lung function stability? *Chest.* Jul 2010;138(1):158-164.

- 35. Chang AB, Robertson CF, Van Asperen PP, et al. A multicenter study on chronic cough in children : burden and etiologies based on a standardized management pathway. *Chest.* Oct 2012;142(4):943-950.
- 36. Smith-Vaughan HC, Binks MJ, Marsh RL, et al. Dominance of Haemophilus influenzae in ear discharge from Indigenous Australian children with acute otitis media with tympanic membrane perforation. *BMC ear, nose, and throat disorders.* 2013;13(1):12.
- 37. Nosworthy E, Price E, Sarovich D, et al. Accurate and rapid identification ofHaemophilus influenzae using real-time PCR. OMOZ; 2014; Melbourne.

	PBB	PBB total	Controls	P-value ¹
	followed-up (N=106)	(N=161)	(N=25)	i vuide
Sex, M:F	74:32	106:55	12:13	0.085
Age - Mo	23 (14, 53)	22 (13, 50)	44 (7, 97)	0.061
Prior pneumonia – X-ray confirmed ²	24 (23%)	35 (22%)	2 (8%)	0.109
Household tobacco smoke exposure	36 (34%)	55 (34%)	6 (24%)	0.662
Aboriginal or Torres Strait Islander	3 (3%)	10 (6%)	0	0.363
Current cough	89 (84%)	134 (83%)	2 (8%)	<0.001
Length of current cough – wks, median (IQR)	26 (7, 52)	26 (6, 52)	0	<0.001
Current antibiotics	16 (15%)	19 (12%)	1/15 (7%)	0.695
>5 doctor visits past yr for	95 (90%)	140 (87%)	3/15 (20%)	<0.001

Table 1: Baseline demographic and clinical characteristics of study participants

Abbreviations: PBB followed-up = Subset of patients with PBB who completed 2-year follow-up;

PBB total = all patients with PBB irrespective of whether 2-year follow-up was completed.

¹ Comparison of 'PBB total' to controls; ² Parent-reported

Characteristic	PBB followed-up (N=106)	Controls (N=25)	P-value
Current cough	47 (44%)	3 (12%)	0.005
Current antibiotics	20/98 (20%)	0	0.012
Recurrent OM (>5 episodes)	58/97 (60%)	7/20 (35%)	0.042
>5 doctor visits past yr for cough	43/97 (44%)	1/20 (5%)	0.001
Asthma diagnosed ever	57/95 (60%)	4/20 (20%)	0.001
Eczema diagnosed ever	29/92 (32%)	3/19 (16%)	0.265
Wheeze ever ¹	76/95 (80%)	7/19 (37%)	<0.001
Wheeze past 12 months	54/93 (58%)	3/19 (16%)	0.001
Family history of atopy ²	12/88 (14%)	1/20 (5%)	0.455
Recurrent PBB ³	67 (63%)	0	<0.001
CT chest (any indication)	20	3	0.564

Table 2: Clinical characteristics of study participants at 2-year follow-up

Abbreviations: PBB followed-up = *Subset of patients with PBB who completed 2-year follow-up; PBB total* = *all patients with PBB irrespective of whether 2-year follow-up was completed.*

¹ Parent-reported ('has your child ever had a high-pitched whistling or wheezing sound')

² Asthma, eczema and/or hay fever in first-degree relative/s

³ In 12-month period after enrolment

Risk factor	Group 1 BE present (N=13)	Group 2 BE absent (N=148)	Odds ratio (95% CI)	P-value
Univariate analysis				
Sex, M:F	8:5	98:50	0.816 (0.25, 2.63)	0.733
Recruitment age – mths, median (IQR)	29 (10, 45)	22 (13, 50)	0.99 (0.97, 1.01)	0.521
Prior pneumonia	3 (23%)	32 (22%)	1.09 (0.28, 4.19)	0.903
Recurrent PBB $(>3 \text{ ep/year})^1$	11 (85%)	56 (38%)	9.04 (1.93, 42.27)	0.005
Household tobacco smoke exposure	3 (23%)	52 (35%)	0.55 (0.14, 2.08)	0.377
Maternal smoking in pregnancy	3 (23%)	18/85 (21%)	1.40 (0.34, 5.81)	0.647
Aboriginal or Torres Strait Islander	1 (8%)	9 (6%)	1.29 (0.15, 11.03)	0.818
No. children in household, median (IQR)	3 (2, 4)	2 (2, 3)	1.28 (0.91, 1.80)	0.156
\geq 2 siblings \geq 1 sibling	8 (62%) 13 (100%)	47 (32%) 123 (83%)	3.44 (1.07, 11.08)	0.039
Childcare attendance, ever	8/9 (89%)	71/82 (87%)	1.24 (0.14, 10.90)	0.847
BAL organism Adenovirus positive (on PCR) ²	3/12 (25%)	26/138 (19%)	1.44 (0.36, 5.68)	0.606
H. influenzae	11 (85%)	72 (49%)	5.81 (1.24, 27.10)	0.025
M. catarrhalis	4 (31%)	43 (29%)	1.09 (0.32, 3.71)	0.896
S. pneumoniae	3 (23%)	41 (28%)	0.78 (0.21, 2.99)	0.720

 Table 3: Uni- and multi-variate analysis of risk factors for bronchiectasis (BE) in children with PBB

	S. aureus	1 (8%)	12 (8%)	0.94 (0.11, 7.90)	0.958
Multivariate	analysis				
H. influenzae	2			7.60 (1.53, 37.79)	0.013
≥2 siblings				3.53 (0.98, 12.70)	0.054
Recurrent PE ep/year)	BB (>3			11.48 (2.33, 56.50)	0.003

¹ As determined at 1-year time-point in study. 7 children had not reached the 1-year time-point,

for the purposes of the multi-variate analysis these children were assumed to have non-recurrent

PBB.

² As compared to previously published rate of 4% (adenovirus positivity) in BAL samples of control children.⁸

1 able 4: Survival analysis – r DD to bronci	neclasis (months) (n-101)	
	HR (95% CI)	P-value
Univariable H. influenzae ¹	5.81 (1.28 - 26.37)	0.022
≥ 2 siblings	3.18 (1.04 - 9.73)	0.042
Recurrent PBB (>3 episodes/yr)	7.65 (1.67 – 34.92)	0.009
<i>Multivariable</i> H. influenzae ¹	7.55 (1.66 – 34.28)	0.009
Recurrent PBB (>3 episodes/yr)	9.77 (2.13 – 44.80)	0.003

 Table 4: Survival analysis – PBB to bronchiectasis (Months) (n=161)

¹ Lower airway infection with *H. influenzae* defined as $\geq 10^4$ cfu/ml growth on BAL fluid culture

Cellularity,	BE present	BE absent	P-value
median (IQR)	$(N=12)^{1}$	$(N=138)^{1}$	
BAL TCC x 10 ⁶ /L	240 (210, 560)	240 (120, 420)	0.29
BAL neutr%,	35 (12, 75)	25 (10, 55)	0.35
BAL macro%,	56 (25, 72)	61 (35, 80)	0.47
BAL lymph%,	6 (4, 9)	9 (4, 14)	0.51
BAL eosin%,	1 (0, 2)	0 (0, 0)	0.001 ²

Table 5: Cellularity from BAL of children with and without bronchiectasis (BE)

¹Cellularity was unavailable, due to insufficient sample, from 11 participants (10 'BE absent' children and 1 'BE present' child).

² When outlier (16% eosinophils) is removed from 'BE present' group, the difference between groups, for eosinophil percentage, was no longer significant.

FIGURE LEGENDS

Figure 1: CONSORT diagram



CHAPTER 7

7.1 INTRODUCTION TO CHAPTER 7

Previous chapters have focused on original research relating to chronic wet cough and PBB in children. Chapter 7 is comprised of 2 manuscripts that reviewed the current literature relating to cough treatments. Chapter 7.1 was a Cochrane meta-analysis examining the efficacy of short-course antibiotics (treatment duration of 4 weeks or less), in the treatment of bronchiectasis in children and adults. Chapter 7.2 was a review of available literature pertaining to cough remedies in children. Both manuscripts are highly applicable to clinical practice, with the latter having direct relevance to clinicians working in general practice.

7.2 Cochrane systematic review: short courses of antibiotics for children and adults with bronchiectasis

Short courses of antibiotics for children and adults with bronchiectasis (Review)

Wurzel D, Marchant JM, Yerkovich ST, Upham JW, Masters IB, Chang AB



This is a reprint of a Cochrane review, prepared and maintained by The Cochrane Collaboration and published in *The Cochrane Library* 2011, Issue 6

http://www.thecochranelibrary.com



Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

TABLE OF CONTENTS

HEADER	1
ABSTRACT	1
PLAIN LANGUAGE SUMMARY	2
BACKGROUND	2
OBJECTIVES	3
METHODS	3
RESULTS	5
DISCUSSION	7
AUTHORS' CONCLUSIONS	7
ACKNOWLEDGEMENTS	7
REFERENCES	8
CHARACTERISTICS OF STUDIES	0
DATA AND ANALYSES	4
Analysis 1.1. Comparison 1 Change in general medical condition, Outcome 1 Improvement in general medical condition.	5
Analysis 2.1. Comparison 2 Respiratory system adverse events, Outcome 1 Respiratory system adverse events 1	5
Analysis 2.2. Comparison 2 Respiratory system adverse events, Outcome 2 Hospitalised for respiratory events	6
Analysis 3.1. Comparison 3 Microbiological response, Outcome 1 Microbiological response.	6
Analysis 4.1. Comparison 4 Development of pseudomonas resistance (when initially sensitive), Outcome 1 Pseudomonas	
resistance	7
HISTORY	7
CONTRIBUTIONS OF AUTHORS	7
DECLARATIONS OF INTEREST	7
SOURCES OF SUPPORT	7
DIFFERENCES BETWEEN PROTOCOL AND REVIEW	8
INDEX TERMS	8

[Intervention Review]

Short courses of antibiotics for children and adults with bronchiectasis

Danielle Wurzel^{1,2}, Julie M Marchant^{1,2}, Stephanie T Yerkovich³, John W Upham⁴, I Brent Masters^{1,2}, Anne B Chang^{1,2,5}

¹Queensland Children's Respiratory Centre, Royal Children's Hospital, Brisbane, Australia. ²Queensland Children's Medical Research Institute, The University of Queensland, Brisbane, Australia. ³QLD Centre for Pulmonary Transplantation and Vascular Disease, The Prince Charles Hospital, Brisbane, Australia. ⁴School of Medicine, University of Queensland & Respiratory Medicine, Princess Alexandra Hospital, The University Queensland & Princess Alexandra Hospital, Brisbane, Australia. ⁵Menzies School of Health Research, Charles Darwin University, Casuarina, Australia

Contact address: Danielle Wurzel, daniellewurzel@netspace.net.au.

Editorial group: Cochrane Airways Group. **Publication status and date:** New, published in Issue 6, 2011. **Review content assessed as up-to-date:** 8 February 2011.

Citation: Wurzel D, Marchant JM, Yerkovich ST, Upham JW, Masters IB, Chang AB. Short courses of antibiotics for children and adults with bronchiectasis. *Cochrane Database of Systematic Reviews* 2011, Issue 6. Art. No.: CD008695. DOI: 10.1002/14651858.CD008695.pub2.

Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

ABSTRACT

Background

Bronchiectasis is an important cause of respiratory morbidity in both developing and developed countries. Antibiotics are considered standard therapy in the treatment of this condition but it is unknown whether short courses (four weeks or less) are efficacious.

Objectives

To determine whether short courses of antibiotics (i.e. less than or equal to four weeks) for treatment of acute and stable state bronchiectasis, in adults and children, are efficacious when compared to placebo or usual care.

Search methods

The Cochrane Central Register of Controlled Trials (CENTRAL, *The Cochrane Library*), MEDLINE, EMBASE, OLDMEDLINE, CINAHL, AMED and PsycINFO and handsearching of respiratory journals and meeting abstracts were performed by the Cochrane Airways Group up to February 2011.

Selection criteria

Only randomised controlled trials were considered. Adults and children with bronchiectasis (defined clinically or radiologically) were included. Patients with cystic fibrosis were excluded.

Data collection and analysis

Two review authors independently reviewed the titles, abstracts and citations to assess eligibility for inclusion. Only one study fulfilled the inclusion criteria and thus meta-analysis could not be performed.

Main results

The single eligible study showed a small benefit, when compared to placebo, of four weeks of inhaled antibiotic therapy in adults with bronchiectasis and pseudomonas in their sputum. There were no studies in children and no studies on oral or intravenous antibiotics.

Authors' conclusions

There is insufficient evidence in the current literature to make reasonable conclusions about the efficacy of short course antibiotics in the management of adults and children with bronchiectasis. Until further evidence is available, adherence to current treatment guidelines is recommended.

PLAIN LANGUAGE SUMMARY

Short courses of antibiotics for children and adults with bronchiectasis

There is a paucity of evidence to conclude whether short courses of antibiotics (i.e. less than or equal to four weeks) are equivalent or superior to placebo in the treatment of stable or exacerbation state bronchiectasis. One single study showed some benefit of short-course inhaled antibiotics over placebo, in terms of microbiological response and subjective improvement in medical condition, although this was balanced against an increase in adverse effects and antimicrobial resistance in the treatment group. Given the potential benefits of shorter duration antibiotic therapy in bronchiectasis, further RCTs are clearly needed to answer this important question.

BACKGROUND

Non cystic-fibrosis bronchiectasis, previously termed an 'orphan disease' is increasingly recognised as a major cause of respiratory morbidity both in developing countries as well as affluent countries, and particularly within Indigenous populations (Chang 2008; Edwards 2003; Singleton 2000).

There are many known aetiological associations with bronchiectasis. Severe lower respiratory tract infection has previously been identified as the most common preceding medical condition in children diagnosed with bronchiectasis (Chang 2008b; Eastham 2004; Karakoc 2009; Singleton 2000), and continues to be a major cause of bronchiectasis worldwide (Callahan 2002; Edwards 2003; Karakoc 2009). However, diseases that affect the pulmonary system (such as immunodeficiency) have become increasingly identified as important causes of bronchiectasis within developed countries (Li 2005; Shoemark 2007).

In adults, the major aetiological associations include post-infection pneumonia or TB, primary ciliary dyskinesia, allergic bronchopulmonary aspergillosis and immunodeficiencies (Shoemark 2007). Primary lung diseases such as chronic obstructive pulmonary disease (COPD) (O'Brien 2000; Patel 2004), sarcoidosis (Lewis 2002), and bronchiolitis obliterans (Chang 1998) have been associated with bronchiectasis as a secondary manifestation. In children, immunodeficiency, aspiration, primary ciliary dyskinesia and congenital airway anomalies are important considerations (Li 2005). Despite intensive investigations, an underlying cause is often not found in many children and adults with bronchiectasis (Shoemark 2007).

Description of the condition

Bronchiectasis is a disease that primarily affects the airways in the initial disease phase. The postulated pathophysiology includes a 'vicious circle' of infection, inflammation and impairment of mucociliary clearance mechanisms eventually leading to airway destruction and bronchial dilatation (Cole 1986).

Adult patients usually present with cough, daily sputum production, dyspnoea, rhinosinusitis, haemoptysis and recurrent pleurisy (King 2004). Children usually present with chronic wet cough or recurrent pneumonia (Singleton 2000). They may have clinical signs such as crackles and wheeze on chest auscultation and sometimes peripheral clubbing (King 2004). With appropriate treatment (usually antibiotics and airway clearance), the chronic cough can totally resolve (Kapur 2009), as can radiological evidence of bronchiectasis (Gaillard 2003). High-resolution Computed Tomography (HRCT) of the chest is considered the gold-standard in diagnosis of bronchiectasis. However, controversy exists as to the normal cut-off of broncho-arterial ratio (particularly in children) (Chang 2008b). Hence, bronchiectasis is sometimes defined clinically in children (Chang 2008b; Singleton 2000). Recent studies suggest that volumetric scans acquired using multi-detector CT are substantially more sensitive and accurate than conventional HRCT for assessing bronchiectasis (Hill 2009).

A combination of airway clearance techniques and antibiotic therapy, with or without other therapies such as anti-inflammatory agents and bronchodilators, are current recommended treatment for non-cystic fibrosis bronchiectasis (Chang 2008b). Specific treatments targeted to the underlying aetiology (such as intravenous immunoglobulin for common variable immunodeficiency) is also important in the management of bronchiectasis.

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd. Early recognition and treatment of bronchiectasis may improve long term outcomes (Kapur 2010).

Description of the intervention

Antibiotics are the mainstay of therapy in the management of bronchiectasis (Prasad 2007). There are various methods to deliver antibiotics to the pulmonary system (oral, intravenous or inhaled). The type of antibiotics given may be targeted to the patient's known airway organism(s) or used empirically.

How the intervention might work

Infection and inflammation are key components in the aetiopathogenesis of bronchiectasis (Cole 1986). Bronchiectatic airways facilitate chronic bacterial colonisation and predispose them to recurrent infections (Loebinger 2007). Chronic bacterial infection elicits a systemic inflammatory response with local release of inflammatory cytokines (including TNF- α and IL-8) causing migration of inflammatory cells such as neutrophils and lymphocytes (Gaga 1998; Shum 2000). Neutrophils release proteolytic enzymes e.g. neutrophil elastase (Amitani 1995) and reactive oxygen species (Anderson 1996), into the airway lumen which cause epithelial damage and stimulate mucous production (Adler 1990). Antibiotics can potentially halt the bacterial infection and subsequently limit ongoing neutrophilic inflammation.

An existing Cochrane Review found that prolonged courses of antibiotics for bronchiectasis, i.e. greater than four weeks in length, are effective in reducing sputum volume and purulence but have a limited impact on the natural history of the condition (Evans 2007). We have reviewed the literature to determine the impact of shorter courses of antibiotics during stable state and exacerbation state in bronchiectasis.

Why it is important to do this review

Most adults and children with bronchiectasis are given frequent courses of antibiotics, the optimal duration of which is unknown. The published Cochrane Review on antibiotics for bronchiectasis is limited to prolonged courses greater than four weeks in duration (Evans 2007).

In clinical practice, short courses of antibiotics during stable and exacerbation states of bronchiectasis often result in improvement in symptoms. Objective measures of airway inflammation also improve, as some studies have shown, for example use of short-course inhaled gentamicin resulted in improvement in airway hypersecretion and inflammation (Lin 1997).

There is currently a paucity of evidence on the optimal duration of antibiotics in stable and exacerbation states. However, there is a trend towards use of shorter courses (i.e. less than two weeks) amongst clinicians. The risks of antibiotic side effects and resistance are a significant concern when using longer courses of antibiotics. A review of the literature to determine the evidence for use of shorter courses of antibiotics in adults and children in stable and exacerbation states of bronchiectasis is important to assist in guiding clinical practice.

OBJECTIVES

To evaluate the efficacy of short courses (i.e. four weeks or less) of antibiotics in children and adults with bronchiectasis;

(a) during stable state bronchiectasis; and

(b) for reducing the severity and frequency of acute respiratory exacerbations.

METHODS

Criteria for considering studies for this review

Types of studies

Any randomised controlled trial comparing outcomes with use of antibiotics (intravenous, oral or inhaled) versus placebo or usual care as the control, for periods of less than, or equal to, four weeks, in non-cystic fibrosis bronchiectasis. Participants are allowed to have adjunctive therapies (such as airway clearance) as long as they have equal chance of having these adjunctive therapies.

Types of participants

Adults and children with bronchiectasis (defined clinically or radiologically) not related to cystic fibrosis.

Types of interventions

Any short (four weeks or less) course of antibiotics given by intravenous, oral or nebulised routes.

Types of outcome measures

Primary outcomes

The primary outcome measures were change in:

1. Symptom score (e.g. bronchiectasis severity control, coughspecific/respiratory-specific quality of life (QoL) or generic health-specific QoL)

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

2. Lung function indices (airway resistance or airway calibre measurements)

3. Adverse events

Secondary outcomes

1. Sputum or airway markers (weight, purulence, colour (Bronkotest), inflammatory profiles)

- 2. Microbiological data (density, resistance patterns)
- 3. Exacerbation data (length, time to next exacerbation)

Search methods for identification of studies

Electronic searches

Randomised controlled trials were identified using the Cochrane Airways Group Specialised Register of trials, which is derived from systematic searches of bibliographic databases including the Cochrane Central Register of Controlled Trials (CENTRAL), MEDLINE, EMBASE, OLDMEDLINE, CINAHL, AMED and PsycINFO, and handsearching of respiratory journals and meeting abstracts (please see the Airways Group search methods for further details). We searched all records in the Specialised Register coded as 'bronchiectasis' using the following terms:

antibiotic* OR *cillin OR *tetracycline OR *mycin OR macrolide* OR quinolone* OR *oxacin OR trimethoprim OR *sulpha OR *ceph or anti-bacteri* or "anti bacteri* OR anti-microbial* OR anti*

The search was conducted in February 2011.

Searching other resources

We handsearched all the papers and reviews identified for further references and contacted authors to request their identification of any unpublished or missed trials. We contacted researchers directly as required to establish whether other unpublished or ongoing studies were available for assessment. We handsearched clinical trials web sites (www.clinicalstudyresults.org; www.clinicaltrials.gov; www.fda.gov).

Data collection and analysis

Selection of studies

Following electronic literature searches, AC and DW independently reviewed the searches to identify potentially relevant trials for full review. We searched bibliographies and texts to identify additional studies. From the full text using specific criteria, AC and DW independently selected trials for inclusion. There was complete agreement between review authors.

Data extraction and management

We extracted information from each study for the following characteristics:

1. Design (description of randomisation, blinding, allocation, number of study centres and location, withdrawals)

2. Participants (N, mean age, age range of the study, baseline lung function, duration of antibiotics < 4 weeks versus no antibiotics)

3. Intervention (type and duration of antibiotic,

appropriateness of antibiotic choice, dosing schedules of groups)4. Outcomes (type of outcomes and results of outcomes)

We requested further information from the trial authors where required. This occurred with Barker 2000 as explained in the included studies section of the Results.

Studies were translated to English where possible.

Assessment of risk of bias in included studies

We assessed trial bias protection in the following domains and study quality according to whether studies met the following prespecified quality criteria (as met, unmet or unclear) using the risk of bias (RoB) table in Review Manager 5.

- 1. Sequence generation
- 2. Allocation concealment
- 3. Blinding of participants and investigators
- 4. Loss to follow-up

Measures of treatment effect

We extracted data for each of the outcomes (where data were available) from the trial publication that fulfilled the inclusion criteria.

We performed an initial qualitative comparison of all the individually analysed studies taking into account differences in study populations, inclusion/exclusion criteria, interventions, outcome assessment and estimated effect size, to examine whether pooling of results (meta-analysis) was reasonable.

Unit of analysis issues

We sought to obtain data that were reported with patients (rather than events) as the unit of analysis for the primary outcomes.

Dealing with missing data

The proportion of randomised patients who provided data for the main outcomes was reported and we had planned to compare this with the number of patients with events in each outcome category.

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

Assessment of heterogeneity

We had planned to describe and explore heterogeneity between the study results, and to use the 95% confidence interval, estimated using a random-effects model, whenever there were concerns about statistical heterogeneity.

Assessment of reporting biases

If combining data was possible, we had planned to assess publication bias using a funnel plot. We planned to identify and report on any selective reporting in the included trials.

Data synthesis

We combined data using Review Manager 5, with a view to using a fixed-effect mean difference (calculated as a weighted mean difference) for continuous data variables. If different scales were combined, we had intended to use the standardised mean difference.

For the dichotomous outcome variables of each individual study, we had planned to calculate odds ratios using a modified intention-to-treat analysis (i.e. failure assumed if participant drops out of study). This analysis assumes that children or adults not available for outcome assessment have not improved (and probably represents a conservative estimate of effect).

For the primary outcomes we intended to calculate a number needed to treat (benefit or harm) when possible for the different levels of risk as represented by control group event rates over a specified time period using the pooled Odds Ratio and its confidence interval using an online calculator, Visual Rx (Cates 2003). A summary of findings (SoF) table would be constructed for the primary outcomes.

Subgroup analysis and investigation of heterogeneity

We had intended to perform an a priori subgroup analysis for: 1. Children versus adults (adult studies will be considered as

- those which recruited participants from 18 years upwards)2. Type of antibiotics (oral, intravenous, inhaled)
- 3. Type of control arm (placebo/no treatment or control
- antibiotic i.e. prolonged duration antibiotics)
- 4. State during enrolment (stable state or exacerbation)

Sensitivity analysis

Sensitivity analyses were also planned to assess the impact of the potentially important factors on the overall outcomes:

1. variation in the inclusion criteria;

2. differences in the medications used in the intervention and comparison groups;

- 3. analysis using random-effects model;
- 4. analysis by "treatment received"; and
- 5. analysis by "intention-to-treat".

RESULTS

Description of studies

See: Characteristics of included studies; Characteristics of excluded studies.

Results of the search

From the searches, the Cochrane Airways Group specialised register/search identified 187 potentially relevant titles. After assessing the titles/abstracts, we obtained19 papers for consideration for inclusion in the review. We reviewed 11 full text articles and 8 further abstracts. We excluded 18 studies (see Excluded studies). We found two potentially eligible studies, however they appeared to have used the same patient population, leaving only one eligible study in adults. We identified no studies in children.

Included studies

The included studies were Barker 2000 and Couch 2001a, which described the same study populations. Couch 2001a was an "early review of data" for a chest supplement on aerosolised therapeutics as discussed by Fiel 2001 in his editorial. For this reason, we used the final data in Barker's paper in this Cochrane Review.

The sole study was a multicentre parallel RCT that examined the effect of nebulised tobramycin (Tobramycin Solution for Inhalation) versus placebo, in adults with CT confirmed bronchiectasis and *Pseudomonas aeruginosa* infection. There were 16 study sites, 74 adults enrolled with completion rate of 81% (60/74). Further details are described in the Characteristics of included studies.

Excluded studies

Eighteen studies were excluded as they did not fulfil criteria for the review. Fifteen of the eighteen studies excluded had no suitable comparator/placebo group, one study did not specify bronchiectasis as an inclusion criterion, one study had treatment periods greater than four weeks, two studies shared a common study population, resulting in the exclusion of one.

Bilton 2006 and Shrewsbury 2004 examined short courses of antibiotics in bronchiectasis and compared oral ciprofloxacin (in treatment doses) plus placebo to oral ciprofloxacin plus inhaled tobramycin. Although these studies almost fulfilled criteria for inclusion, there was no "antibiotic-free" placebo for comparison, resulting in their exclusion from this review.

Couch 2001a was excluded as it was an early review of the same data used in the included study Barker 2000. Attempts to contact the corresponding author were unsuccessful, but this was confirmed in the editorial Fiel 2001 which accompanied the paper.

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

Risk of bias in included studies

Allocation concealment was unclear in this study (Barker 2000). Block randomisation was utilised to balance group sizes. The study was double-blinded, with both investigators and participants being blinded to study group allocation to reduce selection bias. Six patients were withdrawn from the placebo group due to need for an antibiotic other than the study drug.

Allocation

Block randomisation of patients was used. Eligible patients were randomly assigned in blocks of two to parallel groups at each study centre to receive either inhaled tobramycin or placebo.

Blinding

The study was double-blinded i.e. both participants and investigators were blinded to the study drug assignment until completion of the follow-up visit and data were collected from all study sites. The placebo used (quinine sulphate) was chosen because of its similar taste to tobramycin. It was unclear whether investigators were blinded to the study hypothesis.

Incomplete outcome data

The number of participants withdrawn (due to adverse events or use of antibiotic other than the study drug) and those lost to follow-up were reported for both inhaled tobramycin and placebo groups.

Selective reporting

There was no suggestion that selective reporting had occurred.

Effects of interventions

The included trial involved 74 patients with 60 completing the study. In the absence of additional suitable studies, there were insufficient data to perform a meta-analysis. The outcomes from the single study (also presented in the forest plot) were:

Primary Outcomes

I. Symptom score (e.g. bronchiectasis severity control, cough-specific/respiratory-specific quality of life (QoL) or generic health-specific QoL)

The study did not specifically measure any symptoms to determine a symptom score. Instead, the investigator's subjective assessment of a change in the patient's general medical condition ("improved" or "not improved") was recorded at week six: 23 of 37 (62%) patients in the tobramycin-treated group significantly improved compared to 14 of 35 (40%) in the placebo group that improved (Analysis 1.1; OR 0.37; 95% CI 0.14 to 0.95).

2. Lung function indices (airway resistance or airway calibre measurements)

Percent change in forced expiratory volume in one second (FEV₁) percent predicted and in Forced Vital Capacity (FVC) percent predicted from baseline to week four were not statistically significant between the tobramycin and placebo groups (-2.2% versus 1.5% respectively, P = 0.41) for FEV₁ % predicted and FVC % predicted (-2.8% versus 2.2%, P = 0.19). Airway reactivity (percent change in FEV1 from pre- to post-study drug administration) was not significantly different from zero percent for either the tobramycin group (mean = -1%, Week 0, -3% Week 4) or placebo group (mean = -3% Week 0, -1% Week 4).

3. Adverse events

Thirty-one of 37 (84%) patients in each treatment group reported at least one adverse event. Respiratory system adverse events were reported by 26 (70%) tobramycin patients and by 19 (51%) placebo patients. There was no statistical significance, but results favoured the placebo group (OR 2.24; 95% CI 0.86 to 5.82; Analysis 2.1). Five patients in the tobramycin group and one patient in the placebo group were hospitalised and treated for an exacerbation of their pulmonary disease (OR 5.63; 95% CI 0.62 to 50.73; Analysis 2.2).

Secondary outcomes

1. Microbiological response (i.e. change in *P. aeruginosa* density from baseline to week 4 and week 6 and antibiotic resistance)

At the end of the trial, significantly more subjects in the tobramycin group (13 of 37) had *P. aeruginosa* eradicated from their sputum compared to the placebo group (0 of 37); (OR 0.03; 95% CI 0.01 to 0.14; Analysis 3.1.) Furthermore, a further 12 patients in the tobramycin group had reduced *P. aeruginosa* carriage (PA cfu/g decreased by at least 2 \log_{10} at week 4), compared to two patients with reduced carriage in the placebo group.

However, at follow-up two weeks post cessation of antibiotics, the mean reduction in the tobramycin group was smaller than in previous weeks, suggesting some regrowth of organisms after ceasing the antibiotic.

Four of 36 patients in the tobramycin group and one of 32 in the placebo group, who began the study with susceptible *P. aeruginosa*, had resistant *P. aeruginosa* at their last visit (P = 0.36). Three of the four patients in the tobramycin group who developed resistant *P. aeruginosa* showed no microbiological response. All four patients

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

were considered to have not improved when their general medical condition was assessed.

DISCUSSION

This review is limited to a sole study in adults. The data suggest that short term inhaled antibiotics show some benefit in the treatment of patients with bronchiectasis and *P. aeruginosa*. The study followed the patients for only two weeks post-treatment to assess longevity of response. Despite its weaknesses, this single study supports previous Cochrane Reviews on the use of longer term antibiotics in bronchiectasis (Evans 2007), in showing subjective improvement in general medical condition as assessed by an investigator.

This review has highlighted the fact that, although antibiotics are the mainstay of treatment in adults and children with bronchiectasis, there remains a paucity of data, particularly high quality data, to support this practice. In today's evidence based medical era and with bronchiectasis becoming increasingly recognised as a major cause of respiratory morbidity, there will be an increasing need to provide evidence-based answers to these common questions.

A small number of patients in the study antibiotic arm developed "resistant" pseudomonas. This highlights an important complication of antibiotic therapy, particularly when antibiotic use in any one patient is prolonged or recurrent. Previous research by Hillier 2007 has shown that antibiotic resistance increases with increased duration of antibiotic therapy. These findings are related to urinary tract infections but are likely to be applicable to infections elsewhere in the body including the lung. This issue, combined with the likely improved quality of life for patients receiving shorter courses of antibiotics, highlights the need for further studies to determine the optimal duration of antibiotic therapy in bronchiectasis.

The included study followed patients for only two weeks posttherapy. Given that antibiotics are known to indirectly limit neutrophilic inflammation in the airways and ideally eradicate the bacterial infection, it is not surprising that patients showed improvement over this follow-up period. It is unknown if two weeks is an optimal timeframe to assess bacterial eradication adding further to the limitations of this study. To assess longevity of symptom response and microbiological eradication, a longer timeframe of follow-up would have been required.

Evans 2007's Cochrane Review on prolonged antibiotics found a significant benefit of prolonged antibiotics (i.e. four or more weeks) in terms of response rates. They found no significant difference between placebo and prolonged antibiotics in terms of exacerbation rates and lung function. This would concur with Barker 2000, which found a subjective improvement in patients' general medical condition with inhaled tobramycin therapy as compared to placebo, with no significant differences in lung function between treatment groups.

The single included study (Barker 2000) showed some benefit of four weeks of inhaled antibiotic therapy compared with placebo in patients with bronchiectasis and *P. aeruginosa* infection in terms of microbiological response (at the expense of increased resistance) and improvement in general medical condition. The lack of other suitable studies for this Cochrane Review precluded a meta-analysis, and as a result we could not draw any firm conclusions on the topic.

AUTHORS' CONCLUSIONS

Implications for practice

When considering implications for practice, one must acknowledge the fact that this review was based on one study of adult patients and the intervention was "inhaled" antibiotics. This study showed a small benefit in microbiological response and overall subjective improvement in general medical condition, with no effect on lung function and an increase in adverse events in the intervention group. The study failed to address primary outcome measures with symptom scores or objective outcomes. Therefore, although this one study is suggestive of benefit of short courses of antibiotics for bronchiectasis, one would conclude, for current practice implications, that there is insufficient evidence available in the current literature to make any reasonable conclusions. However until further evidence is available, clinicians should adhere to guidelines (Chang 2010; Pasteur 2010) that include data from non RCTs.

Implications for research

Well-designed, double-blinded, parallel, randomised controlled trials are required to assess the role of short courses of antibiotics in the treatment of bronchiectasis. These studies should include validated outcome measures such as improvements in symptom score, QOL and lung function, balanced against adverse effects as primary outcome measures. Trials including oral, inhaled and intravenous antibiotic administration methods are needed. Future RCTs should be undertaken in paediatric as well as adult patients. Well-designed studies should include sufficient longitudinal follow-up of patients to assess longevity of intervention response and therefore applicability of evidence to practice. Such well-designed research would have the potential to improve the quality of life for individuals with bronchiectasis.

ACKNOWLEDGEMENTS

We thank Toby Lasserson, Dr Chris Cates, Elizabeth Stovold (neé Arnold) and Susan Ann Hansen from the Cochrane Airways

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd. Group for their advice, supportive role and comments on the protocol and review. We also thank Martina Franke and Elizabeth Stovold for kindly assisting with German and Spanish translation respectively.

REFERENCES

References to studies included in this review

Barker 2000 {published data only}

Barker AF, Couch L, Fiel SB, Gotfried MH, Ilowite J, Meyer KC, et al. Tobramycin solution for inhalation reduces sputum Pseudomonas aeruginosa density in bronchiectasis. *American journal of respiratory and critical care medicine* 2000;**162**(2 Pt 1):481–5.

References to studies excluded from this review

Bilton 2006 {published data only}

Bilton D, Henig N, Morrissey B, Gotfried M. Addition of inhaled tobramycin to ciprofloxacin for acute exacerbations of Pseudomonas aeruginosa infection in adult bronchiectasis. *Chest* 2006;**130**(5):1503–10.

Chrysanthopoulos 1989 {published data only}

Chrysanthopoulos CJ, Starakis JC, Skoutelis AT, Bassaris HP. Brief report: Sequential intravenous/oral therapy with ciprofloxacin in severe infection. *American Journal of Medicine* 1989;**87**(5 A):225S–7S.

Couch 2001a {published data only}

Couch L. Treatment with tobramycin solution for inhalation in bronchiectasis patients with Pseudomonas aeruginosa. *Chest* 2001;**120**(3 Suppl 11):4S–7S.

Douglas 1957 {published data only}

Douglas AC, Somner AR, Marks BL, Grant IW. Effect of antibiotics on purulent sputum in chronic bronchitis and bronchiectasis. *Lancet* 1957;**273**(6988):214–8.

Howie 1976 {published data only}

Howie JG. Antibiotics and purulent sputum. *British Medical Journal* 1976;**2**(6040):882.

Lam 1989 {published data only}

Lam WK, Chau PY, So SY, Leung YK, Chan JC, Ip M, et al.Ofloxacin compared with amoxycillin in treating infective exacerbations in bronchiectasis. *Respiratory medicine* 1989; **83**(4):299–303.

Lin 1996 {published data only}

Lin HC, Wang CH, Liu CY, Yu CT, Kuo HP. Amoxicillin downregulates the release of neutrophil derived proinflammatory cytokines and C5a in patients with bronchiectasis [abstract]. *European Respiratory Journal. Supplement.* 1996;**9 Suppl 23**:400s.

Matsumoto 1986 {published data only}

Matsumoto K, Takahashi A, Yamamoto M. A study on the evaluation of the efficacy of chemotherapy for chronic respiratory tract infections. [Japanese]. *Chemotherapy* 1986; **34**(4):316–30.

May 1972 {published data only}

May JR, Ingold A. Amoxycillin in the treatment of chronic non-tuberculous bronchial infections. *British Journal of Diseases of the Chest* 1972;**66**(3):185–91.

Mehta 1991 {published data only}

Mehta AP, Phutane L, Patel MH, Rupwate RU, Kamat SR. Comparative study of amoxycillin and amoxycillin/ clavulanic acid in lower respiratory infections. *Journal of the Association of Physicians of India* 1991;**39**(3):251–3.

Messens 1973 {published data only}

Messens Y, Oger A, Cornette M, Calay G. [Double-blind study of the sulfamethoxazole-trimethoprim association in the treatment of pulmonary infections]. *Acta Clinica Belgica* 1973;**28**(2):92–9.

Mijuskovic 1972 {published data only}

Mijuskovic B, Macanovic J. Bisolvomycin and Bisolvon in the treatment of chronic bronchitis, bronchial asthma and bronchiectasis. *Plucne Bolesti i Tuberkuloza* 1972;**24**(3): 229–33.

Nagy 1968 {published data only}

Nagy M, Meszaros G. [Late results of the conservative treatment of bronchiectasis in adults]. [German]. Zeitschrift fur Tuberkulose und Erkrankungen der Thoraxorgane 1968; **127**(5):283–90.

Oki 1993 {published data only}

Oki Y, Morishita M, Miyachi A. An evaluation of low dose long term erythromycin chemotherapy in bronchiectasis. [Japanese]. *Japanese Journal of Chest Diseases* 1993;**52**(1): 54–8.

Santiveri 1995 {published data only}

Santiveri C, Orriols R, Roig J, Balcells E, Bellver P, Ferrer A, et al. Effectiveness of inhaled antibiotic treatment for pseudomonas aeruginosa in outpatients with colonized bronchiectasis without mucoviscidosis. *Archivos de Bronconeumologia* 1995;**31**(Suppl):42.

Shrewsbury 2004 {published data only}

Shrewsbury SB, Bilton D, Gotfried M, Jones SA. The TABLE study (TOBI in acute bronchiectasis: additional treatment for exacerbations): rationale and methodolgy [Abstract]. American Thoracic Society 100th International Conference, May 21-26, 2004, Orlando. 2004:A42 Poster A53.

Tagaya 2002 {published data only}

Tagaya E, Tamaoki J, Kondo M, Nagai A. Effect of a short course of clarithromycin therapy on sputum production in patients with chronic airway hypersecretion. *Chest* 2002; **122**(1):213–8.

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

Twiss 2008 {published data only}

Twiss J, Byrnes CA. Nebulized antibiotics reduce symptoms, bacterial density and oral antibiotic usage in children with non cystic fibrosis bronchiectasis [Abstract]. American Thoracic Society International Conference, May 16-21, 2008, Toronto. 2008:A681[#C40].

Additional references

Adler 1990

Adler KB, Holden-Stauffer WJ, Repine JE. Oxygen metabolites stimulate release of high-molecular-weight glycoconjugates by cell and organ cultures of rodent respiratory epithelium via an arachidonic acid-dependent mechanism. *The Journal of clinical investigation* 1990;**85**(1): 75–85.

Amitani 1995

Amitani R, Taylor G, Elezis EN, Llewellyn-Jones C, Mitchell J, Kuze F, et al.Purification and characterization of factors produced by Aspergillus fumigatus which affect human ciliated respiratory epithelium. *Infection and immunity* 1995;**63**(9):3266–71.

Anderson 1996

Anderson R, Feldman C, Theron AJ, Ramafi G, Cole PJ, Wilson R. Anti-inflammatory, membrane-stabilizing interactions of salmeterol with human neutrophils in vitro. *British Journal of Pharmacology* 1996;**117**(7):1387–94.

Callahan 2002

Callahan CW, Redding GJ. Bronchiectasis in children: orphan disease or persistent problem?. *Pediatric Pulmonology* 2002;**33**(6):492–6.

Cates 2003

Cates C. Visual Rx. Online NNT Calculator. http:// www.nntonline.net/. London: Cates C, 2003.

Chang 1998

Chang AB, Masel JP, Masters B. Post-infectious bronchiolitis obliterans: clinical, radiological and pulmonary function sequelae. *Pediatric Radiology* 1998;**28**(1):23–9.

Chang 2008

Chang AB, Bilton D. Exacerbations in cystic fibrosis: 4--Non-cystic fibrosis bronchiectasis. *Thorax* 2008;**63**(3): 269–76.

Chang 2008b

Chang AB, Grimwood K, Macguire G, King PT, Morris PS, Torzillo PJ. Management of bronchiectasis and chronic suppurative lung disease (CSLD) in Indigenous children and adults from rural and remote Australian communities. *Medical Journal of Australia* 2008;**189**:386–93.

Chang 2010

Chang AB, Bell SC, Byrnes CA, Grimwood K, Holmes PW, King PT, et al.Bronchiectasis and chronic suppurative lung disease (CSLD) in children and adults in Australian and New Zealand: Thoracic Society of Australia and New Zealand and Australian Lung Foundation Position Statement. *Medical Journal of Australia* 2010;**193**:356–65.

Cole 1986

Cole PJ. Inflammation: a two-edged sword--the model of bronchiectasis. *European Journal of Respiratory Diseases Supplement* 1986;**147**:6–15.

Eastham 2004

Eastham KM, Fall AJ, Mitchell L, Spencer DA. The need to redefine non-cystic fibrosis bronchiectasis in childhood. *Thorax* 2004;**59**(4):324–7.

Edwards 2003

Edwards EA, Asher MI, Byrnes CA. Paediatric bronchiectasis in the twenty-first century: experience of a tertiary children's hospital in New Zealand. *Journal of Paediatrics and Child Health* 2003;**39**(2):111–7.

Evans 2007

Evans DJ, Bara AI, Greenstone M. Prolonged antibiotics for purulent bronchiectasis in children and adults. *Cochrane Database of Systematic Reviews* 2007, Issue 2. [DOI: 10.1002/14651858.CD001392.pub2]

Fiel 2001

Fiel SB. History and evolution of aerosolized therapeutics: overview and introduction. *Chest* 2001;**120**(3 Suppl): 87S–8S.

Gaga 1998

Gaga M, Bentley AM, Humbert M, Barkans J, O'Brien F, Wathen CG, et al.Increases in CD4+ T lymphocytes, macrophages, neutrophils and interleukin 8 positive cells in the airways of patients with bronchiectasis. *Thorax* 1998;**53** (8):685–91.

Gaillard 2003

Gaillard EA, Carty H, Heaf D, Smyth RL. Reversible bronchial dilatation in children: comparison of serial high-resolution computer tomography scans of the lungs. *European Journal of Radiology* 2003;47:215–20.

Hill 2009

Hill LE, Ritchie G, Wightman AJ, Hill AT, Murchison JT. Comparison between conventional interrupted highresolution CT and volume multidetector CT acquisition in the assessment of bronchiectasis. *British Journal of Radiology* 2009;**63**:67–70.

Hillier 2007

Hillier S, Roberts Z, Dunstan F, Butler C, Howard A, Palmer S. Prior antibiotics and risk of antibiotic-resistant community-acquired urinary tract infection: a case-control study. *The Journal of Antimicrobial Chemotherapy* 2007;**60** (1):92–9.

Kapur 2009

Kapur N, Masters IB, Chang AB. Exacerbations in non cystic fibrosis bronchiectasis: Clinical features and investigations. *Respiratory Medicine* 2009;**103**:1681–7.

Kapur 2010

Kapur N, Masters IB, Chang A. Longitudinal growth and lung function in pediatric non-CF bronchiectasis what influences lung function stability?. Chest 2010, issue EPub ahead of print Feb 19, 2010. [DOI: 10.1378/ chest.09-2932]

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

Karakoc 2009

Karakoc GB, Inal A, Yilmaz M, Altintas DU, Kendirli SG. Exhaled breath condensate MMP-9 levels in children with bronchiectasis. *Pediatric Pulmonology* 2009;44(10):1010–6.

King 2004

King PT, Freezer NJ, Holmes PW, Holdsworth SR, Forshaw K, Sart DD. Role of CFTR mutations in adult bronchiectasis. *Thorax* 2004;**59**(4):357–8.

Lewis 2002

Lewis MM, Mortelliti MP, Yeager H, Tsou E. Clinical bronchiectasis complicating pulmonary sarcoidosis: case series of seven patients. *Sarcoidosis, vasculitis, and diffuse lung diseases : official journal of WASOG / World Association of Sarcoidosis and Other Granulomatous Disorders* 2002;**19** (2):154–9.

Li 2005

Li AM, Sonnappa S, Lex C, Wong E, Zacharasiewicz A, Bush A, et al.Non-CF bronchiectasis: does knowing the aetiology lead to changes in management?. *The European respiratory journal : official journal of the European Society for Clinical Respiratory Physiology* 2005;**26**(1):8–14.

Lin 1997

Lin HC, Cheng HF, Wang CH, Liu CY, Yu CT, Kuo HP. Inhaled gentamicin reduces airway neutrophil activity and mucus secretion in bronchiectasis. *American Journal of Respiratory and Critical Care Medicine* 1997;**155**(6):2024–9.

Loebinger 2007

Loebinger MR, Wilson R. Pharmacotherapy for bronchiectasis. *Expert Opinion on Pharmacotherapy* 2007;**8** (18):3183–93.

O'Brien 2000

O'Brien C, Guest PJ, Hill SL, Stockley RA. Physiological and radiological characterisation of patients diagnosed with chronic obstructive pulmonary disease in primary care. *Thorax* 2000;**55**(8):635–42.

Pasteur 2010

Pasteur MC, Bilton D, Hill AT. British Thoracic Society guideline for non-CF bronchiectasis. *Thorax* 2010;**65**: i1–i58.

Patel 2004

Patel IS, Vlahos I, Wilkinson TM, Lloyd-Owen SJ, Donaldson GC, Wilks M, et al.Bronchiectasis, exacerbation indices, and inflammation in chronic obstructive pulmonary disease. *American Journal of Respiratory and Critical Care Medicine* 2004;**170**(4):400–7.

Prasad 2007

Prasad M, Tino G. Bronchiectasis, part 2: Management. *The Journal of respiratory diseases* 2007;**29**(1):20–5.

Review Manager 5

The Nordic Cochrane Centre, The Cochrane Collaboration. Review Manager (RevMan). 5.0. Copenhagen: The Nordic Cochrane Centre, The Cochrane Collaboration, 2008.

Shoemark 2007

Shoemark A, Ozerovitch L, Wilson R. Aetiology in adult patients with bronchiectasis. *Respiratory Medicine* 2007;**10**: 1163–70.

Shum 2000

Shum DK, Chan SC, Ip MS. Neutrophil-mediated degradation of lung proteoglycans: stimulation by tumor necrosis factor-alpha in sputum of patients with bronchiectasis. *American Journal of Respiratory and Critical Care Medicine* 2000;**162**(5):1925–31.

Singleton 2000

Singleton R, Morris A, Redding G, Poll J, Holck P, Martinez P, et al.Bronchiectasis in Alaska Native children: causes and clinical courses. *Pediatric Pulmonology* 2000;**29**(3):182–7.

* Indicates the major publication for the study

CHARACTERISTICS OF STUDIES

Characteristics of included studies [ordered by study ID]

Barker 2000

Methods	The study was a randomised placebo-controlled, double-blinded trial with antibiotics administered for 4 weeks. Eligible patients were randomly assigned in blocks of two to parallel groups at each of 16 study sites across the United States to receive either Tobramycin solution for inhalation or placebo. This was administered twice daily for 4 weeks in bronchiectasis patients whose sputum contained <i>Pseudomonas aeruginosa</i> . Patients were then observed for 2 weeks after administration of their last dose. Of patients enrolled, they were divided equally between the two treatment groups. Patients were withdrawn if they required any additional antibiotics at any stage during study participation Patients were screened 2 weeks prior to their initial dose of the study drug, were dosed for 4 weeks, and then observed for 2 weeks after their last dose. Thus the total duration of the study was 8 weeks. At each visit a sputum sample was obtained and the density of the <i>P. aeruginosa</i> in sputum was measured. Pulmonary function testing (FEV1 and FVC) was performed at baseline and at the final treatment visit (week 4). Tobramycin levels were measured. Adherence was measured at week 4 by counting the number of vials of study drug used. A subjective clinical assessment of the patient's general medical condition was made, by a study investigator, at the follow-up visit on week 6 There were 6 patients who withdrew from the tobramycin group (3 for adverse events, 2 for use of antibiotics other than study drug, and 1 was lost to follow-up.) 8 withdrew from the placebo group (2 for adverse events, and 6 for use of antibiotics other than the study drug.)
Participants	125 patients were screened and 74 patients (45 female), mean age 66.6 (tobramycin group) and 63.2 (placebo group) were enrolled. Patients were block randomised to receive either 300 mg of inhaled tobramycin or placebo twice daily for 4 weeks. 37 received inhaled tobramycin and 37 received placebo Inclusion: Bronchiectasis diagnosed by conventional or high-resolution CT and sputum containing at least 10 ⁴ cfu/mL <i>P. aeruginosa</i> . Exclusion: cystic fibrosis, allergic bronchopulmonary aspergillosis, acute pulmonary process requiring medical intervention as indicated by a new infiltrate on a chest radiograph, significant recent haemoptysis, or had received antibiotics within 2 weeks of the screening visit
Interventions	Nebulised tobramycin solution for inhalation (300 mg tobramycin) twice daily or placebo for 4 weeks
Outcomes	Primary end point was (1) Change in <i>P. aeruginosa</i> density from baseline to week 4. Additional endpoints included: (1) change in <i>P. aeruginosa</i> density from baseline values to week 2 and to week 6; (2) an investigator's subjective assessment of a change in the patient's general medical condition ("improved" or "not improved") was made and recorded at week 6, (3) the percent change in FEV1 percent predicted and in FVC percent predicted from week 0 to week 4 and (4) Safety endpoints included the incidence of adverse events, change in serum chemistry and haematology measurements, and airway reactivity

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

Barker 2000 (Continued)

Each patient's microbiological response was categorised according to whether *P. aeruginosa* was eradicated, reduced by treatment, or did not respond to treatment. *P. aeruginosa* was considered eradicated if it was not detected at week 6 or if it was not detected at week 4 and the patient was unable to produce sputum at week 6. A patient's response was defined as reduced by treatment if *P. aeruginosa* was recovered from the week 6 sputum and reduced by at least 2 log 10 cfu/g at week 4 compared with baseline. A patient had no microbiological response if *P. aeruginosa* did not decrease 2 log 10 cfu/g at week 4 or if the patient withdrew from the study

Notes

Risk of bias

Bias	Authors' judgement	Support for judgement
Allocation concealment (selection bias)	Unclear risk	Unclear
Blinding of participants and investigators.	Low risk	Patients and investigators were blinded to the study drug assignment
Incomplete outcome data assessed?	Low risk	Number of dropouts and loss to follow-up included.
Free of selective reporting?	Low risk	No suggestion that selective reporting may have occurred.
Free of other bias?	Low risk	No other bias identified
Sequence generation.	Low risk	Sequence generation was referred to, how- ever details were not provided

Characteristics of excluded studies [ordered by study ID]

Study	Reason for exclusion
Bilton 2006	No suitable placebo/comparator group. (Comparator was not adequate to satisfy inclusion criteria, i.e. no "antibiotic-free/placebo" or long-term antibiotic group for comparison.)
Chrysanthopoulos 1989	No suitable placebo/comparator group.
Couch 2001a	Overlap of study participants with included study.
Douglas 1957	No suitable placebo/comparator group.
Howie 1976	Study participants had unknown bronchiectasis status.

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

(Continued)

Lam 1989	No suitable placebo/comparator group.
Lin 1996	No suitable placebo/comparator group.
Matsumoto 1986	No suitable placebo/comparator group. Not an RCT.
May 1972	No suitable placebo/comparator group.
Mehta 1991	No suitable placebo/comparator group.
Messens 1973	Translated from French. No suitable placebo/comparator group
Mijuskovic 1972	No suitable placebo/comparator group.
Nagy 1968	No suitable placebo/comparator group. (Compared antibiotic treatment to surgery.)
Oki 1993	No suitable placebo/comparator group. (Examined long term antibiotic treatment, no short course treat- ment arm.)
Santiveri 1995	No suitable placebo/comparator group.
Shrewsbury 2004	No suitable placebo/comparator group. (Comparator was not adequate to satisfy inclusion criteria, i.e. no "antibiotic-free/placebo" or long-term antibiotic group for comparison.)
Tagaya 2002	No suitable placebo/comparator group.
Twiss 2008	Longer-course antibiotics (i.e. 2 to 3 months) compared to placebo. No short course group
DATA AND ANALYSES

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Improvement in general medical condition	1		Odds Ratio (M-H, Fixed, 95% CI)	Subtotals only

Comparison 1. Change in general medical condition

Comparison 2. Respiratory system adverse events

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Respiratory system adverse events	1		Odds Ratio (M-H, Fixed, 95% CI)	Subtotals only
2 Hospitalised for respiratory events	1		Odds Ratio (M-H, Fixed, 95% CI)	Subtotals only

Comparison 3. Microbiological response

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Microbiological response	1		Odds Ratio (M-H, Fixed, 95% CI)	Subtotals only

Comparison 4. Development of pseudomonas resistance (when initially sensitive)

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Pseudomonas resistance	1		Odds Ratio (M-H, Fixed, 95% CI)	Subtotals only

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright 0 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

Analysis I.I. Comparison I Change in general medical condition, Outcome I Improvement in general medical condition.

Review: Short courses of antibiotics for children and adults with bronchiectasis

Comparison: I Change in general medical condition

Outcome: I Improvement in general medical condition

Study or subgroup	TSI	Placebo	C	Odds Ratio	Odds Ratio
	n/N	n/N	M-H,Fi>	(ed,95% Cl	M-H,Fixed,95% Cl
Barker 2000	14/37	23/37		-	0.37 [0.14, 0.95]
Subtotal (95% CI)	0	0			0.0 [0.0, 0.0]
Total events: 14 (TSI), 23 (Placebo	o)				
Heterogeneity: not applicable					
Test for overall effect: $Z = 0.0$ (P	< 0.00001)				
			<u> </u>	<u> </u>	
			0.01 0.1	1 10 100	
			Favours TSI	Favours placebo	

Analysis 2.1. Comparison 2 Respiratory system adverse events, Outcome 1 Respiratory system adverse events.

Review: Short courses of antibiotics	for children and adul	ts with bronchiectasis			
Comparison: 2 Respiratory system a	adverse events				
Outcome: I Respiratory system adv	verse events				
Study or subgroup	TSI	Placebo	C	Odds Ratio	Odds Ratio
	n/N	n/N	M-H,Fi>	«ed,95% Cl	M-H,Fixed,95% Cl
Barker 2000	26/37	19/37			2.24 [0.86, 5.82]
Subtotal (95% CI) Total events: 26 (TSI), 19 (Placebo) Heterogeneity: not applicable Test for overall effect: Z = 0.0 (P < 0.0	0	0			0.0 [0.0, 0.0]
			0.01 0.1 Favours TSI	I IO IOO Favours placebo	

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

Analysis 2.2. Comparison 2 Respiratory system adverse events, Outcome 2 Hospitalised for respiratory events.

Review: Short courses of antibiotics for children and adults with bronchiectasis

Comparison: 2 Respiratory system adverse events

Outcome: 2 Hospitalised for respiratory events

Study or subgroup	Tobramycin	Placebo		Odds Ratio	Odds Ratio
	n/N	n/N	M-H,	Fixed,95% CI	M-H,Fixed,95% CI
Barker 2000	5/37	1/37			5.63 [0.62, 50.73]
Subtotal (95% CI)	0	0			0.0 [0.0, 0.0]
Total events: 5 (Tobramycin), I (P	'lacebo)				
Heterogeneity: not applicable					
Test for overall effect: $Z = 0.0$ (P	< 0.00001)				
			1 1		
			0.01 0.1	I IO IOO	
			Favours experimental	Favours control	

Analysis 3.1. Comparison 3 Microbiological response, Outcome 1 Microbiological response.

Review: Short courses of antibiotics for children and adults with bronchiectasis

Comparison: 3 Microbiological res	sponse				
Outcome: I Microbiological respo	onse				
Study or subgroup	TSI n/N	Placebo n/N	C M-H,Fi>	Odds Ratio ked,95% Cl	Odds Ratio M-H,Fixed,95% Cl
Barker 2000	12/37	33/35	```		0.03 [0.01, 0.14]
Subtotal (95% CI) Total events: 12 (TSI), 33 (Placebo) Heterogeneity: not applicable Test for overall effect: $Z = 0.0$ (P < 0	0	0			0.0 [0.0, 0.0]
			001 01	1 10 100	
			Favours TSI	Favours placebo	

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

Analysis 4.1. Comparison 4 Development of pseudomonas resistance (when initially sensitive), Outcome I Pseudomonas resistance.

Review: Short courses of antibiotics for children and adults with bronchiectasis

Comparison: 4 Development of pseudomonas resistance (when initially sensitive)

Outcome: I Pseudomonas resistance

Study or subgroup	TSI	Placebo	00	dds Ratio	Odds Ratio
	n/N	n/N	M-H,Fixe	ed,95% CI	M-H,Fixed,95% Cl
Barker 2000	4/36	1/32			3.88 [0.41, 36.63]
Subtotal (95% CI)	0	0			0.0 [0.0, 0.0]
Total events: 4 (TSI), 1 (Placebo)					
Heterogeneity: not applicable					
Test for overall effect: Z = 0.0 (P <	0.00001)				
			0.01 0.1 1	10 100	

Favours experimental Favours control

HISTORY

Protocol first published: Issue 9, 2010

Review first published: Issue 6, 2011

CONTRIBUTIONS OF AUTHORS

DW and AC wrote the protocol, based on previous protocols on cough in children. DW and AC selected articles from the search and together with JM, wrote the manuscript. BM, JU and SY contributed by reviewing the manuscript.

DECLARATIONS OF INTEREST

No conflicts of interest.

SOURCES OF SUPPORT

Internal sources

• Australian Cochrane Airways Group Network Graduate Scholarship, Australia. Support to attend Cochrane Airways course

Short courses of antibiotics for children and adults with bronchiectasis (Review) Copyright © 2011 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

External sources

- TSANZ Allen and Hanbury's Paediatric Respiratory Grant-in-aid, Australia.
- To support paediatric respiratory training for DW
- Queensland Children's Medical Research Institute, Australia.
- Top-up scholarship for DW; Program grant for AC
 - NHMRC, Australia.
- Practitioner Fellowship for AC

DIFFERENCES BETWEEN PROTOCOL AND REVIEW

The duration of short course antibiotics was originally specified in the protocol as less than four weeks, however, in the actual review this was amended to four weeks or less to enable inclusion of Barker 2000.

INDEX TERMS

Medical Subject Headings (MeSH)

Anti-Bacterial Agents [administration & dosage; *therapeutic use]; Bronchiectasis [*drug therapy]; Drug Administration Schedule; Randomized Controlled Trials as Topic

MeSH check words

Adult; Child; Humans

7.3 Drug treatments of childhood cough

Drug treatments of childhood coughs

SUMMARY

Appropriate management of cough in children depends upon accurate assessment. The diagnosis is often unclear at the initial presentation.

Acute cough is frequently caused by a viral infection, and often no specific therapy is indicated. Urgent treatment may be needed if history suggests a more serious disorder such as a foreign body or pneumonia.

When treating children with chronic cough, paediatric-specific algorithms should be used. Empirical use of medicines without looking for a specific cause should be avoided.

In the absence of an alternative specific cause of cough, chronic wet cough (lasting at least four weeks) is most frequently due to protracted bacterial bronchitis. Antibiotics are indicated.

Introduction

Cough is the most common symptom presented to GPs and pharmacists in Australia. An Australian study found that 'one in three (28.7%) respiratory episodes were associated with a doctor's visit, and one in four (23%) necessitated time off school or work'.¹ When a child first presents with cough, determining the precise diagnosis is not always possible.

Acute cough

Acute cough in a child may represent a variety of pathologies, from self-resolving viral-induced acute respiratory infection to acute severe respiratory disease or an acute presentation of an underlying chronic disorder. Appropriate management depends on accurate assessment. Patient history should include:²

- cough duration (acute <2 weeks, sub-acute 2-4 weeks, chronic >4 weeks)
- characteristics of cough (whooping cough, wet vs dry cough)
- questions about choking episodes and previous respiratory illness
- associated wheeze
- other symptoms such as weight loss, appetite or rash
- immunisation history.

In the differential diagnosis, it is important to consider inhaled foreign body, pneumonia and other treatable infections like pertussis and underlying lung disease such as bronchiectasis.

Uncomplicated acute upper respiratory infections

It is commonly said that young children have up to 6–12 acute respiratory infections per year. However,

a Melbourne-based community study involving 600 families showed fewer episodes and an age-dependent trend (see Table).¹ The mean duration of episodes was 6.3 days (range 1–70 days) and younger children were more likely to have a longer duration of cough (6.8 days in youngest age group and 5.5 days in oldest group).

Management

Supportive therapy is the mainstay of treatment for viral acute respiratory infections. Paracetamol and ibuprofen are useful for related symptoms. Over-the-counter cough and cold medicines are not recommended due to a lack of proven efficacy and the possibility that they may present a safety risk.³ The Therapeutic Goods Administration now recommends that they should not be used in children under 6 years and only in children aged 6–11 years on advice from a doctor.⁴

Honey,^{5,6} and menthol-based rubs⁷ may reduce the impact of nocturnal cough. It is reasonable to recommend one teaspoon of honey before bedtime for children aged over one year. Honey should be avoided in children under one year due to the risk of botulism.

TableAustralian rates of uncomplicated
acute upper respiratory infections
in children and young adults 1

Age (years)	Mean number of episodes a year
0–1	3.8
2-3	3.3
4-5	2.8
6–10	2.2
11-20	2

Danielle F Wurzel

Paediatric respiratory physician Clinical research fellow^{1,2}

Julie M Marchant Paediatric respiratory physician^{1,2}

Anne B Chang

Paediatric respiratory physician¹ Professor Paediatrics^{1,3,4}

¹Queensland Children's Respiratory Centre Queensland Children's Medical Research Institute Royal Children's Hospital Brisbane ² The University of Queensland ³ Child Health Division Menzies School of Health Research Darwin ⁴ Queensland University of Technology

Key words

asthma, bronchiectasis, bronchiolitis, bronchitis, croup, pertussis, pneumonia

Aust Prescr 2014;37:115-9

Childhood coughs

Antibiotics should be avoided for the treatment of acute cough associated with mild upper respiratory tract infection, as the cough is most likely viral in origin. A recent Cochrane review reported that in cases of confirmed or suspected exposure to influenza in healthy children, oseltamivir shortens the time to first alleviation of symptoms by 29 hours (95% confidence interval 12–47 hours, p=0.001).⁸ No effect however was seen in children with asthma. Oseltamivir may reduce the risk of otitis media in children aged 1–5 years, especially if commenced within the first 12 hours, but is associated with a significantly increased risk of vomiting.⁹ Laboratory-based polymerase chain reaction (PCR) techniques enable rapid influenza diagnosis.¹⁰

Management of acute cough should include counselling and advice on:

- the expected duration of cough (typically 5–7 days, but up to 3 weeks)
- when to come back and see the GP and when to seek urgent medical review (for example suspected foreign body, tachypnoea, dyspnoea, vomiting, inability to feed, persistent fever, lethargy)
- avoidance of passive smoke exposure.

Specific causes of acute cough

A number of specific diseases need to be considered in a child presenting with acute cough. Many of these have specific symptoms and signs.

Croup

The acute or sub-acute onset of a barking 'brassy' cough, hoarse voice, stridor with or without evidence of upper airway obstruction, is characteristic of croup. It often begins with a viral upper respiratory tract infection (for example rhinorrhoea, sore throat with or without fever) and typically affects children aged 1–6 years. Children outside this age range or with severe or recurrent stridor or other symptoms require careful evaluation for an underlying airway lesion. Children with bacterial causes of stridor such as tracheitis or epiglottitis usually appear more toxic.

Prednisolone 1–2 mg/kg orally for two consecutive days is effective for croup. Dexamethasone 0.15 mg/kg orally is an appropriate alternative therapy. In severe croup, when a child has ongoing stridor at rest, increasing fatigue and marked tachycardia with or without signs of impending hypoxaemia (for example, lethargy and increased irritability), urgent transfer to an emergency facility is recommended. Potentially distressing interventions, such as throat examination, should be avoided, as these may worsen respiratory obstruction.

Pneumonia

Children with pneumonia often have cough, fever and tachypnoea, but occasionally present with fever and

upper abdominal pain. Signs of severity include grunt and intercostal recession. Wheeze is usually absent in bacterial pneumonia.

A chest X-ray does not need to be performed routinely in all children with suspected pneumonia. However, it should be considered in any child with an atypical presentation (recurrent pneumonia, prolonged fever, signs of pleural effusion) or severe pneumonia requiring hospital admission.¹¹

Recommendations for antimicrobial therapy vary according to the age of the child, context, presence of underlying disease (risk factors), presence of hypoxaemia, non-respiratory symptoms (such as vomiting), length and severity of symptoms and the presence of complications. Guidelines for antimicrobial therapy should be consulted.¹¹⁻¹³ For a child with subacute onset and prominent cough (with or without headache or sore throat), or who is not improving, mycoplasma pneumonia should be suspected.¹³

Indications for hospitalisation for community-acquired pneumonia include:

- very young children (less than 6 months) with suspected bacterial pneumonia¹²
- clinical evidence of moderate to severe pneumonia, including hypoxaemia and signs of respiratory distress¹²
- significant comorbidities or factors which predispose to more severe disease e.g. immunodeficiency, congenital heart disease, bronchiectasis¹¹
- pneumonia suspected or confirmed to be secondary to a pathogen with increased virulence e.g. community-acquired methicillin-resistant Staphylococcus aureus (MRSA)¹²
- dehydration or inability to tolerate oral therapies¹¹
- significant parental concern or anxiety¹¹
- family unable to provide appropriate care or adhere to management plan¹²
- toxic-looking child e.g. pale or cyanotic, lethargic or inconsolably irritable
- complicated pneumonia e.g. empyema
- poor response after 48 hours of oral antibiotics.

All children with suspected pneumonia should be followed up regularly to ensure complete resolution of their symptoms. A repeat chest X-ray is not routinely performed following simple pneumonia unless there are persisting symptoms.¹¹

Bronchiolitis

Children under two years presenting acutely with cough, tachypnoea (with or without poor feeding) and often with a history of a viral prodrome may have viral bronchiolitis. Clinical examination reveals hyperinflation with widespread wheeze and crackles on chest auscultation. Respiratory syncytial virus is the most common infection associated with bronchiolitis.

Any infant with apnoeas, hypoxia (oxygen saturations ≤92%), dehydration or poor feeding requires hospital admission for supplemental oxygen with or without hydration therapy. Children frequently worsen in the first 72 hours before showing improvement. The cough can persist for 2–3 weeks after other symptoms resolve. There is no evidence for the routine use of antibiotics, steroids or asthma drugs in viral bronchiolitis.

Pertussis

Pertussis (whooping cough) typically presents with cough lasting two or more weeks with cough paroxysms, inspiratory whoop or post-tussive vomiting. Confirmation with a PCR-positive nasopharyngeal aspirate or swab is recommended. If there is a high clinical suspicion, start antibiotics before receiving the test results. Clarithromycin (7.5 mg/kg up to 500 mg orally, 12-hourly for 7 days) or erythromycin (10 mg/kg up to 250 mg orally, 6-hourly for 7 days) is recommended.¹³ Treat early to improve symptoms (within 1–2 weeks of start of symptoms) and reduce the infectious period. Patients are seldom infectious after having a cough for longer than three weeks and antibiotics are not recommended at this point.

Chronic cough

The common causes of chronic cough in children differ from those in adults¹⁴ so adult-type management approaches directed at asthma, rhinitis and gastro-oesophageal reflux disease do not apply. In a multicentre study involving 346 new referrals to respiratory paediatricians for chronic cough, the most common diagnoses included protracted bacterial bronchitis (41%), asthma (15.9%) and bronchiectasis (9%). In 13.9% of children, cough resolved without a specific diagnosis.¹⁵

A detailed respiratory history and examination as well as use of a chronic cough algorithm (see Fig.)¹⁶ assist in the assessment and diagnosis of chronic cough.

Fig. Simplified paediatric chronic cough algorithm



Adapted from reference 16

ARTICLE

Childhood coughs

The cough algorithm also significantly improves quality of life and reduces duration of cough.¹⁶ This approach is based on determining the cause of the cough (through systematic history taking and a thorough examination), in addition to spirometry (in a child >5 years of age) and chest X-ray. Indications for referral to a specialist are listed in the Box.

Box Common indications for specialist referral in chronic childhood cough

Chronic cough (>4 weeks) of unclear aetiology (with or without failure to thrive) Suspected airway malformation e.g. tracheo-oesophageal fistula, vascular ring Cough and feeding difficulties (suspected aspiration disease) Clinical features of chronic lung disease e.g. clubbing Persisting auscultatory findings e.g. crepitations Recurrent pneumonias Abnormalities on chest X-ray or spirometry Failure to respond to treatment e.g. in asthma

Protracted bacterial bronchitis

Protracted bacterial bronchitis is the most common cause of chronic wet cough in Australian children.^{14,16} It is defined as:

- cough lasting more than four weeks
- response to two weeks of antibiotic therapy
- absence of specific pointers indicating an alternative cause.

A history of a preceding viral infection is common. Protracted bacterial bronchitis is more common in boys than girls and in those aged 1–3 years.

Lower airway bacterial infection is frequently found on bronchoalveolar lavage sampling and is usually accompanied by elevated neutrophils suggestive of active airway inflammation. The major bacterial organisms found are *Haemophilus influenzae*, *Moraxella catarrhalis* and *Streptococcus pneumoniae*. After exclusion of other causes of chronic cough, a two-week course of amoxycillin-clavulanate is recommended. Children should receive follow-up after 2–3 weeks to ensure complete resolution of cough. A chest X-ray should be performed in any child with clinical suspicion of an alternative cause of chronic cough or if their cough persists despite antibiotic therapy.

Bronchiectasis

Bronchiectasis is another important cause of wet cough to consider, and should be suspected in any child with the following:

- chronic wet cough lasting longer than eight weeks
- two or more episodes of chronic wet cough (lasting ≥4 weeks) per year responding to antibiotics

 chest radiographic changes lasting more than six weeks despite appropriate antibiotic therapy.¹⁷

Antibiotic therapy is usually started at the onset of wet cough in children known to have bronchiectasis. Antibiotic selection is based upon lower airway culture, local antibiotic susceptibility patterns and clinical severity. If symptoms do not respond promptly or adequately to oral antibiotic therapy patients should be hospitalised for intravenous antibiotics. Regular physiotherapy, physical exercise, avoidance of triggers (for example tobacco smoke) and routine vaccinations are recommended.¹⁷ Aboriginal and Torres Strait Islander children are at increased risk of bronchiectasis and doctors should be aware that cough may be under-reported by those from remote communities.¹⁸

Asthma and chronic cough

While asthma can cause chronic cough, isolated chronic cough without any other symptoms in children is rarely due to asthma.^{19,20} Other symptoms usually present in asthma are wheeze, dyspnoea, chest tightness or exercise limitation. Risk factors such as eczema, hay fever, allergies or a family history of asthma in a first-degree relative are often present.

Spirometry and measurements of airway responsiveness (for example exercise challenge) in children aged over five years can help to diagnose asthma. The presence of atopy does not distinguish asthma from other causes of chronic cough. Previous response to asthma therapies may be helpful, however response on a single occasion does not necessarily mean that the child has asthma. Guidelines for the management of asthma are available from the National Asthma Council of Australia (www.nationalasthma.org.au/handbook).

Conclusion

Accurate diagnosis of cough in children depends upon a thorough clinical history and examination to guide appropriate prescribing. The nature of the cough and its chronicity provide important diagnostic clues as to a specific cause of cough. Cough guidelines and algorithms further enhance diagnostic accuracy and may help to ensure more effective prescribing of cough therapies in children.

D Wurzel is funded by a National Health and Medical Research Council (NHMRC) postgraduate scholarship and the Queensland Children's Medical Research Institute, and has received previous funding from the Thoracic Society of Australia and New Zealand (Allen and Hanbury's Award). A Chang is supported by an NHMRC fellowship and various NHMRC grants including a Centre of Research Excellence in Lung Health of Aboriginal and Torres Strait Islander Children grant.

Q:

SELF-TEST QUESTIONS

True or false?

1. Aboriginal and Torres Strait Islander children have an increased risk of bronchiectasis.

2. There is evidence for use of steroids in viral bronchiolitis.

Answers on page 143

REFERENCES

- Leder K, Sinclair MI, Mitakakis TZ, Hellard ME, Forbes A. A community-based study of respiratory episodes in Melbourne, Australia. Aust N Z J Public Health 2003;27:399-404.
- Chang AB, Landau LI, Van Asperen PP, Glasgow NJ, Robertson CF, Marchant JM, et al. Cough in children: definitions and clinical evaluation. Med J Aust 2006;184:398-403.
- Smith SM, Schroeder K, Fahey T. Over-the-counter medications for acute cough in children and adults in ambulatory settings. Cochrane Database Syst Rev 2012;8:CD001831.
- Therapeutic Goods Administration. OTC cough and cold medicines for children - Final outcomes of TGA review. Canberra: Australian Government Department of Health; 2012.

www.tga.gov.au/industry/otc-notices-cough-cold-reviewoutcomes.htm [cited 2014 Jul 11]

- 5. Cranswick N. Cough and cold remedies for children. Aust Prescr 2013;36:e1.
- Oduwole O, Meremikwu MM, Oyo-Ita A, Udoh EE. Honey for acute cough in children. Cochrane Database Syst Rev 2012;3:CD007094.
- Paul IM, Beiler JS, King TS, Clapp ER, Vallati J, Berlin CM Jr. Vapor rub, petrolatum, and no treatment for children with nocturnal cough and cold symptoms. Pediatrics 2010;126:1092-9.
- Jefferson T, Jones MA, Doshi P, Del Mar CB, Hama R, Thompson MJ, et al. Neuraminidase inhibitors for preventing and treating influenza in healthy adults and children. Cochrane Database Syst Rev 2014;4:CD008965.
- Winther B, Block SL, Reisinger K, Dutkowski R. Impact of oseltamivir treatment on the incidence and course of acute otitis media in children with influenza. Int J Pediatr Otorhinolaryngol 2010;74:684-8.
- 10. Foo H, Dwyer DE. Rapid tests for the diagnosis of influenza. Aust Prescr 2009;32:64-7.

- Harris M, Clark J, Coote N, Fletcher P, Harnden A, McKern M, et al. British Thoracic Society guidelines for the management of community acquired pneumonia in children: update 2011. Thorax 2011;66 Suppl 2:ii1-23.
- Bradley JS, Byington CL, Shah SS, Alverson B, Carter ER, Harrison C, et al. The management of community-acquired pneumonia in infants and children older than 3 months of age: clinical practice guidelines by the Pediatric Infectious Diseases Society and the Infectious Diseases Society of America. Clin Infect Dis 2011;53:e25-76.
- Antibiotic Expert Group. eTG complete [internet]. Melbourne: Therapeutic Guidelines Limited; 2010. www.tg.org.au [cited 2014 Jul 11]
- Marchant JM, Masters IB, Taylor SM, Cox NC, Seymour GJ, Chang AB. Evaluation and outcome of young children with chronic cough. Chest 2006;129:1132-41.
- Chang AB, Robertson CF, Van Asperen PP, Glasgow NJ, Mellis CM, Masters IB, et al. A multicenter study on chronic cough in children: burden and etiologies based on a standardized management pathway. Chest 2012;142:943-50.
- Chang AB, Robertson CF, van Asperen PP, Glasgow NJ, Masters IB, Teoh L, et al. A cough algorithm for chronic cough in children: a multicenter, randomized controlled study. Pediatrics 2013;131:e1576-83.
- Chang AB, Bell SC, Byrnes CA, Grimwood K, Holmes PW, King PT, et al. Chronic suppurative lung disease and bronchiectasis in children and adults in Australia and New Zealand. Med J Aust 2010;193:356-65.
- Morey MJ, Cheng AC, McCallum GB, Chang AB. Accuracy of cough reporting by carers of Indigenous children. J Paediatr Child Health 2013;49:E199-203.
- 19. McKenzie S. Cough but is it asthma? Arch Dis Child 1994;70:1-2.
- Wright AL, Holberg CJ, Morgan WJ, Taussig LM, Halonen M, Martinez FD. Recurrent cough in childhood and its relation to asthma. Am J Respir Crit Care Med 1996;153:1259-65.

FURTHER READING

Centre for Clinical Practice at NICE. Evidence review and recommendations. In: Respiratory Tract Infections - Antibiotic Prescribing: Prescribing of antibiotics for self-limiting respiratory tract infections in adults and children in primary care. NICE Clinical Guidelines 69. London: NICE; 2008.

Chang A. Cough. Pediatr Clin North Am 2009;56:19-31.

Chang AB, Redding GJ, Everard ML. Chronic wet cough: protracted bronchitis, chronic suppurative lung disease and bronchiectasis. Pediatr Pulmonol 2008;43:519-31.

Wurzel DF, Marchant JM, Clark JE, Masters IB, Yerkovich ST, Upham JW, et al. Wet cough in children: infective and inflammatory characteristics in broncho-alveolar lavage fluid. Pediatr Pulmonol 2014:49:561-8.

Australian Prescriber books

As an *Australian Prescriber* reader you are offered a 15% discount by McGraw-Hill on two books. The books are comprehensive collections of articles originally published in *Australian Prescriber*.

Use the promotional code AUSCRIBE when ordering. Orders over \$40 will be shipped free.

Contact McGraw-Hill Education ANZ www.mcgraw-hill.com.au



Pharmacokinetics Made Easy D Birkett

'Presents a complex subject in a simple, easy-to-understand manner'



Phone: 1800 646 572

Abnormal Laboratory Results

Email: cservice_sydney@mheducation.com

G Kellerman, editor Also available as an e-book

'Examines the dilemmas of abnormal tests and outlines how to approach them'

CHAPTER 8

8.1 DISCUSSION

Over the past two decades, the significance of chronic wet cough has been increasingly appreciated. Using a systematic approach to evaluate the most common cause of chronic cough in children, the relationship between wet cough, bacterial infection and neutrophilic airway inflammation was identified and the clinical entity of protracted bacterial bronchitis (PBB) was first described in 2006 [2]. Early research into PBB was subsequently supported by studies by other research groups, both locally and internationally [75-77]. Defining the clinical entity of PBB has been pivotal in our understanding of chronic wet cough in children. It has also facilitated further studies on the most common aetiologies of chronic cough in children [16, 27].

Prior to this PhD thesis, several research gaps existed in the area of PBB and chronic wet cough in children. These included:

- 1) the significance of wet cough as a symptom
- 2) the role of viruses in PBB
- 3) the clinical profile of children with PBB and

4) the validity and accuracy of the postulated link between PBB and bronchiectasis.

Addressing these questions in this PhD has resulted in a body of research that has direct relevance to the clinician, whilst also providing insight into areas that warrant further study. Our findings in relation to wet cough progress our understanding of the 'grass roots' symptom (and sign) of PBB.

My PhD work has demonstrated that children with wet cough are significantly more likely to harbour lower airways infection with bacteria, viruses and viral-bacterial co-infection, when compared to children without wet cough [130] (Chapter 2). We have shown that viral-bacterial co-infection is associated with maximal neutrophilic lower airway inflammation, suggesting that microbial synergy may also play a role in the pathogenesis of chronic bronchitis in children [130]. In further exploring the potential role of viruses, chapter 3 showed that adenoviruses and rhinoviruses are the most common viruses detected in the lower airways of children with chronic respiratory symptoms (most of whom had chronic cough). Further, adenovirus detection (via PCR) on NPA or BAL is associated with significantly increased lower airway neutrophilic inflammation, when compared to adenovirus negativity [131]. This finding was independent of the presence of bacterial infection, suggesting that adenovirus is an independent predictor of neutrophilic lower airway inflammation in children. Further supporting the potential relevance of this finding was the fact that adenovirus was significantly more likely to be detected in the BAL of children with PBB compared to controls (23% vs 4% respectively) [132](Chapter 5) suggesting a possible role in disease pathogenesis.

As adenovirus had not previously been described in PBB, the logical next step was to examine the genotypes of adenovirus in the lower airways of children with PBB and bronchiectasis (Chapter 4). We showed that adenovirus species C (genotypes 1 and 2) is the major adenovirus species detected in the lower airways of children with PBB and bronchiectasis [133]. Adenovirus C is known to establish latent infection within lymphocytes and is believed to remain dormant within tonsillar and adenoidal tissues of young children [102, 134] and to persist in the lung [135]. We also found adenovirus to be positively associated with lower airway bacterial infection with *H. influenzae* and *M. catarrhalis* [133]. These findings provide further evidence for a role of (viral-bacterial) microbial synergy in the development of PBB and/or bronchiectasis.

In further delineating the clinical entity of PBB, chapter 5 describes the typical child who presents with PBB. We have shown that young boys, in their second year of life, who have attended childcare and have a history of protracted wet cough and parent-reported wheeze, typify the average child with PBB [132]. We have also shown that basic immune function in these children is intact. However, natural killer (NK) cell levels are often elevated [132]. This latter finding is in keeping with the observation that viruses are significantly more likely to be detected in the lower airways of children with PBB compared to controls, a common factor in NK cell elevation. However, the postulated link between PBB and bronchiectasis is perhaps the most significant theme addressed in this PhD thesis in chapter 6.

We prospectively followed a group of young children with PBB over a two-year period and showed that a significant proportion (8.1%) were subsequently diagnosed with

bronchiectasis on HRCT chest [136] (chapter 6). Predictors of bronchiectasis included: recurrent episodes of PBB (defined as >3 episodes in the first year after recruitment) and lower airway infection with *H. influenzae* (most commonly *non-typeable H. influenzae*) [136]. These findings are first to lend support to the proposed clinical spectrum linking PBB and bronchiectasis, suggesting that this theory [51] is likely to be accurate and robust.

I have thus far summarised the major novel findings arising from the original research encapsulated within this PhD. In addition, a review of the broader medical literature on major cough treatments available for children was summarised in chapter 7. This was followed by a Cochrane review examining the efficacy of short-course antibiotic therapy for the treatment of bronchiectasis. The relevance of this chapter extends beyond the paediatric community and has direct utility for general practice.

The findings of the original research studies contribute to previous work on this topic and allow us to come closer to a hypothesis on the complete pathophysiological process underlying chronic wet cough and PBB in children. The following schematic diagram depicts this proposed theory and I have further elaborated in the text below. The theory expands on Cole's vicious circle hypothesis [81] and Chang's proposed clinical spectrum of PBB and bronchiectasis [51]. I have intentionally omitted discussion of extrinsic factors, such as antibiotic therapy, that are likely to attenuate the infection/inflammation process, as it was beyond the scope of our findings.



8.1 Schematic diagram of proposed pathophysiological process uniting PBB and bronchiectasis (figure adapted from Chang's proposed clinical spectrum of PBB and bronchiectasis [51]).

A primary insult to the lung occurs in a window of susceptibility. This window of susceptibility is defined by an immature immune system as well as structural immaturity of the lung. The insult is an extrinsic initiating event, such as adenovirus C infection. This event impairs host respiratory defences, including mucociliary clearance and immune surveillance mechanisms. This impairment in lung defences increases host susceptibility to secondary bacterial infection. Infection with respiratory bacteria e.g. *non-typeable H influenzae* occurs, or, bacterial load increases if infection is already present. Microbial synergy between pathogens heightens mucosal inflammation and dysfunction further promoting persistence of both virus and bacteria in the airways. Viral-bacterial co-infection underlies and drives the vicious cycle of airway inflammation and infection. For a period of time (weeks/months), airway epithelial injury occurs at the same rate as repair. As lung growth and immune system maturation occur, airway clearance improves and epithelial repair exceeds injury and the vicious cycle abates. This 'reversible airway epithelial injury' occurs in the majority of children with PBB.

In a subset of susceptible individuals (approximately 8.1%[136]), however, a threshold of 'irreversible airway epithelial injury' whereby injury has exceeded repair, is reached. This airway damage acts as a nidus for ongoing infection and inflammation. As the rate of airway injury continues to exceed repair, the discrepancy between the two increases, and the trajectory from PBB to bronchiectasis begins.

8.2 LIMITATIONS OF THIS THESIS

The limitations of this thesis are outlined in the manuscripts contained within each chapter. I will summarise some of the major limitations common to the majority of studies in this thesis.

With the exception of the longitudinal follow-up component of Chapter 6, studies were cross-sectional in nature. Hence, interpretation of lower airway findings, obtained from single time-point sampling, is limited to the observation of association. Cause-effect relationships cannot be assumed. Nevertheless, there is substantial value in the analysis of lower airway data from such a large cohort of young children. Due to the high numbers of included participants, our findings provide invaluable clues to the state of the lower airways in children with (and without) chronic cough. The inability to perform longitudinal

studies involving BAL in children is based upon ethical constraints that are inherent to studies in children.

A second limitation that merits discussion is the duration of follow-up of children with PBB. In Chapter 6, the clinical outcomes of a cohort of children with PBB followed for a 2-year period were described. Indeed, this relatively short period of follow-up allowed detection of a number of children with bronchiectasis. However, it is likely that this observed value would be higher with a longer period of follow-up e.g. 5 years. Furthermore, only a subset of children with PBB underwent CT scan of their chest.

Due to the ethical issues pertaining to the performing of CT chest primarily for research purposes, children in this study did not have routine CT scan at study entry. Only those with clinical suspicion of bronchiectasis were subjected to CT scan. The consequences of this are two-fold. First, as not all children had CT at baseline, we cannot assume that those subsequently diagnosed with bronchiectasis entered the study free of bronchiectasis. Hence, progression from PBB to bronchiectasis per se cannot be assumed. Second, as not all children had CT at follow-up, bronchiectasis may have been unrecognised in a subset that did in fact develop it, hence under-estimating the true rate of this disease in the cohort.

8.3 FUTURE PROJECTS AND FURTHER RESEARCH

The findings from the research arising from this PhD thesis provide the foundation for future research in the area of chronic wet cough and PBB. Potential studies are outlined below. Concepts are presented in preliminary format.

BIRTH-COHORT STUDY

The exact influence of early life respiratory tract infections upon the trajectory of lung health is unknown. Findings from our research suggest that viral-bacterial co-infection is likely to be relevant to the pathogenesis of chronic suppurative lung diseases in children. In addition to organism type/s, it is possible that the timing and sequence of early respiratory tract infections is of clinical significance and impacts upon the risk of development of bronchiectasis.

A birth cohort study would enable documentation of early viral and bacterial respiratory tract infections. Frequent, time-intervalled sampling of respiratory tract secretions when a child is in the well-state, with additional samples during periods of illness, would enable determination of both the timing and sequence of viral and bacterial respiratory tract infections and their relationship to clinical symptoms.

NPA would be used for detection of an extended panel of common respiratory viruses via PCR. Further speciation and genotyping of adenoviruses and rhinoviruses would establish the duration of infection with individual virus types and identify chronic carriage, if any. Cough swab or induced sputum would be used to obtain respiratory specimens for bacteriology.

Documentation of clinical symptoms and PBB episodes, undertaken over time, with regular follow-up with a specialist respiratory physician, would identify those children requiring further investigation for bronchiectasis. The frequency of PBB episodes in relation to presence and timing of viral and bacterial infection would be analysed. Risk factors for recurrent PBB and bronchiectasis (as separate outcome measures) would be determined using longitudinally collected data.

CHRONIC COUGH AND OBSTRUCTIVE SLEEP APNOEA

The observation in clinical practice that chronic cough is often associated with obstructive sleep apnoea (OSA) symptoms and that the chronic cough often resolves after effective intervention for OSA, has been the focus of recent research [137, 138]. Several theories in relation to the cause of cough in children with OSA have been proposed. These range from aspiration of upper airway secretions during sleep (resulting in bacterial bronchitis) to mechanical irritation of the epiglottis by enlarged tonsillar tissue [139]. Indeed an association between asthma and sleep-disordered breathing has also been observed in both adults [140] and children [140-143].

In this PhD we have also collected data on OSA symptoms in a large number of children who have undergone bronchoscopy for investigation of chronic cough. Hence, a crosssectional study, further examining the relationship between chronic cough and OSA, with exploration of clinical and laboratory factors influencing this association, would be highly clinically relevant. Further, it is possible that PBB is the cause of chronic cough in children with large adenoids and tonsils and analysis of already collected data would enable further evaluation of this postulated link.

8.4 CONCLUSION

The novel findings of this PhD thesis complement current knowledge on chronic wet cough and PBB in children in several ways. We have shown that viruses, in addition to bacteria, are likely to play a role in wet cough and PBB. We have also provided further depth to our understanding of PBB as a diagnostic entity by delineating the major clinical features of this condition. We have shown that adenovirus C is the major adenovirus species in the airways of children with PBB and bronchiectasis and, in demonstrating an association with bacterial co-infection, we have provided evidence for the possible role of microbial synergy. Lastly, we have evaluated the accuracy of the proposed clinical spectrum of PBB and bronchiectasis and shown that this is likely to be robust. We have identified key risk factors for bronchiectasis and highlighted important areas for future research.

REFERENCES

- 1. Marchant JM, Newcombe PA, Juniper EF, Sheffield JK, Stathis SL, Chang AB. What is the burden of chronic cough for families? Chest **2008**; 134(2): 303-9.
- 2. Marchant JM, Masters IB, Taylor SM, Cox NC, Seymour GJ, Chang AB. Evaluation and outcome of young children with chronic cough. Chest **2006**; 129(5): 1132-41.
- Thomson F, Masters IB, Chang AB. Persistent cough in children and the overuse of medications. J Paediatr Child Health 2002; 38(6): 578-81.
- Marchant JM, Gibson PG, Grissell TV, Timmins NL, Masters IB, Chang AB. Prospective assessment of protracted bacterial bronchitis: airway inflammation and innate immune activation. Pediatr Pulmonol **2008**; 43(11): 1092-9.
- 5. Marchant JM, Masters IB, Taylor SM, Chang AB. Utility of signs and symptoms of chronic cough in predicting specific cause in children. Thorax **2006**; 61(8): 694-8.
- Gibson PG, Chang AB, Glasgow NJ, et al. CICADA: Cough in Children and Adults: Diagnosis and Assessment. Australian cough guidelines summary statement. Med J Aust 2010; 192(5): 265-71.
- Chang AB, Glomb WB. Guidelines for evaluating chronic cough in pediatrics: ACCP evidence-based clinical practice guidelines. Chest **2006**; 129(1 Suppl): 260S-83S.
- 8. Chang AB, Glasgow NJ, Robertson CF, Marchant JM, Zealand TSoAaN. Cough in children: definitions and clinical evaluation. The Medical journal of Australia **2006**; 184(8): 398-403.
- Shields MD, Bush A, Everard ML, McKenzie S, Primhak R. BTS guidelines: Recommendations for the assessment and management of cough in children. Thorax 2008; 63 iii1-iii15.
- 10. Hay AD, Wilson A, Fahey T, Peters TJ. The duration of acute cough in pre-school children presenting to primary care: a prospective cohort study. Fam Pract **2003**; 20(6): 696-705.

- Butler CC, Hood K, Kinnersley P, Robling M, Prout H, Houston H. Predicting the clinical course of suspected acute viral upper respiratory tract infection in children. Fam Pract 2005; 22(1): 92-5.
- 12. Karakoc F, Karadag B, Akbenlioglu C, et al. Foreign body aspiration: what is the outcome?Pediatr Pulmonol **2002**; 34(1): 30-6.
- Korpas J, Sadlonova J, Salat D, Masarova E. The origin of cough sounds. Bull Eur Physiopathol Respir **1987**; 23 Suppl 10: 47s-50s.
- Hashimoto Y, Murata A, Mikami M, Nakamura S, Yamanaka E, Kudoh S. Influence of the rheological properties of airway mucus on cough sound generation. Respirology **2003**; 8(1): 45-51.
- Korpas J, Widdicombe JG, Vrabec M. Influence of simulated mucus on cough sounds in cats. Respir Med **1993**; 87(1): 49-54.
- 16. Chang AB, Robertson CF, van Asperen PP, et al. A cough algorithm for chronic cough in children: a multicenter, randomized controlled study. Pediatrics **2013**; 131(5): e1576-83.
- 17. Marchant J, Masters IB, Champion A, Petsky H, Chang AB. Randomised controlled trial of amoxycillin clavulanate in children with chronic wet cough. Thorax **2012**.
- Chang AB, Faoagali J, Cox NC, et al. A bronchoscopic scoring system for airway secretions--airway cellularity and microbiological validation. Pediatr Pulmonol **2006**; 41(9): 887-92.
- 19. Chang AB, Gaffney JT, Eastburn MM, Faoagali J, Cox NC, Masters IB. Cough quality in children: a comparison of subjective vs. bronchoscopic findings. Respir Res **2005**; 6: 3.
- 20. Chang AB, Van Asperen PP, Glasgow N, et al. Children with chronic cough: when is watchful waiting appropriate? Development of likelihood ratios for assessing children with chronic cough. Chest **2015**; 147(3): 745-53.
- 21. Elphick HE, Sherlock P, Foxall G, et al. Survey of respiratory sounds in infants. Arch Dis Child **2001**; 84(1): 35-9.

- Mehrotra A, Lasley MV. Early Rattles, Purrs and Whistles as Predictors of Later Wheeze.
 Pediatrics 2008; 122(Supplement): S212-S3.
- 23. Cane RS, Ranganathan SC, McKenzie SA. What do parents of wheezy children understand by "wheeze"? Arch Dis Child **2000**; 82(4): 327-32.
- De Schutter I, Dreesman A, Soetens O, et al. In young children, persistent wheezing is associated with bronchial bacterial infection: a retrospective analysis. BMC Pediatr **2012**; 12(1): 83.
- Faniran AO, Peat JK, Woolcock AJ. Persistent cough: is it asthma? Arch Dis Child **1998**;
 79(5): 411-4.
- 26. Britt H MG, Knox S et al. Bettering the evaluation and care or health: a study of general practice activity. In: Welfare AloHa, 2002.
- Chang AB, Robertson CF, Van Asperen PP, et al. A multicenter study on chronic cough in children : burden and etiologies based on a standardized management pathway. Chest 2012; 142(4): 943-50.
- Cornford CS, Morgan M, Ridsdale L. Why do mothers consult when their children cough?
 Fam Pract **1993**; 10(2): 193-6.
- 29. Davies MJ, Cane R, Ranganathan S, McKenzie SA. Cough, wheeze and sleep. Arch Dis Child **1998**; 79(5): 465.
- Dicpinigaitis PV, Tso R, Banauch G. Prevalence of depressive symptoms among patients with chronic cough. Chest **2006**; 130(6): 1839-43.
- McGarvey LP, Carton C, Gamble LA, et al. Prevalence of psychomorbidity among patients with chronic cough. Cough **2006**; 2: 4.
- Morice AH, Fontana GA, Belvisi MG, et al. ERS guidelines on the assessment of cough.
 Eur Respir J 2007; 29(6): 1256-76.
- Chung KF, Pavord ID. Prevalence, pathogenesis, and causes of chronic cough. Lancet
 2008; 371(9621): 1364-74.

- Canning BJ, Chang AB, Bolser DC, et al. Anatomy and neurophysiology of cough: CHEST Guideline and Expert Panel report. Chest **2014**; 146(6): 1633-48.
- Kantar A, Shields M, Cardinale F, Chang A. Cough. In: Eber E, Midulla F. ERS handbook: Paediatric Respiratory Medicine:. Sheffield, UK: The European Respiratory Society, 2013:44-9.
- 36. Chang AB. Cough, cough receptors, and asthma in children. Pediatr Pulmonol **1999**; 28(1):59-70.
- 37. Mazzone SB, McLennan L, McGovern AE, Egan GF, Farrell MJ. Representation of capsaicin-evoked urge-to-cough in the human brain using functional magnetic resonance imaging. Am J Respir Crit Care Med **2007**; 176(4): 327-32.
- Dinwiddie R. Anatomy and development of the respiratory system. In: Eber E, Midulla F. ERS handbook: Paediatric Respiratory Medicine:. Sheffield, UK: The European Respiratory Society, **2013**:1-10.
- Taussig LM, Landau LI, Le Souëf PN. C. Gaultier and A. Denjean. Developmental Anatomy and Physiology of the Respiratory System. In: Pediatric respiratory medicine. Vol. 2nd; 2. Philadelphia: Mosby/Elsevier, 2008.
- 40. Devlieger H, Daniels H, Marchal G, Moerman P, Casaer P, Eggermont E. The diaphragm of the newborn infant: anatomical and ultrasonographic studies. J Dev Physiol **1991**; 16(6): 321-9.
- 41. Papastamelos C, Panitch HB, England SE, Allen JL. Developmental changes in chest wall compliance in infancy and early childhood. J Appl Physiol (1985) **1995**; 78(1): 179-84.
- 42. Curzi-Dascalova L. Thoracico-abdominal respiratory correlations in infants: constancy and variability in different sleep states. Early Hum Dev **1978**; 2(1): 25-38.
- 43. Curzi-Dascalova L, Peirano P, Morel-Kahn F. Development of sleep states in normal premature and full-term newborns. Dev Psychobiol **1988**; 21(5): 431-44.

- 44. Kohyama J, Shiiki T, Shimohira M, Hasegawa T. Asynchronous breathing during sleep.Arch Dis Child **2001**; 84(2): 174-7.
- Foster WM. Mucus hypersecretion and mucus clearance in cough. In: Chung KF, Widdicombe JG, Boushey HA, editors Cough: Causes, Mechanisms and Therapy: London: Blackwell Publishing Ltd, **2003**:207-16.
- Chang AB. Pediatric cough: children are not miniature adults. Lung **2010**; 188 Suppl 1: S33-40.
- 47. Chang AB, Berkowitz RG. Cough in the pediatric population. Otolaryngol Clin North Am**2010**; 43(1): 181-98, xii.
- 48. Chang AB, Phelan PD, Sawyer SM, Del Brocco S, Robertson CF. Cough sensitivity in children with asthma, recurrent cough, and cystic fibrosis. Arch Dis Child **1997**; 77(4): 331-4.
- 49. Kelsall A, Decalmer S, McGuinness K, Woodcock A, Smith JA. Sex differences and predictors of objective cough frequency in chronic cough. Thorax **2009**; 64(5): 393-8.
- 50. Gewolb IH, Bosma JF, Taciak VL, Vice FL. Abnormal developmental patterns of suck and swallow rhythms during feeding in preterm infants with bronchopulmonary dysplasia. Dev Med Child Neurol **2001**; 43(7): 454-9.
- 51. Chang AB, Redding GJ, Everard ML. Chronic wet cough: Protracted bronchitis, chronic suppurative lung disease and bronchiectasis. Pediatr Pulmonol **2008**; 43(6): 519-31.
- 52. Roberts CR, Pare PD. Composition changes in human tracheal cartilage in growth and aging, including changes in proteoglycan structure. Am J Physiol **1991**; 261(2 Pt 1): L92-101.
- Scherer PW, Burtz L. Fluid mechanical experiments relevant to coughing. J Biomech **1978**; 11(4): 183-7.
- 54. Yeates DB, Aspin N, Levison H, Jones MT, Bryan AC. Mucociliary tracheal transport rates in man. J Appl Physiol **1975**; 39(3): 487-95.

- 55. Bennett WD, Foster WM, Chapman WF. Cough-enhanced mucus clearance in the normal lung. J Appl Physiol (1985) **1990**; 69(5): 1670-5.
- 56. Camner P, Mossberg B, Philipson K, Strandberg K. Elimination of test particles from the human tracheobronchial tract by voluntary coughing. Scand J Respir Dis **1979**; 60(2): 56-62.
- 57. Goyal V, Grimwood K, Marchant J, Masters IB, Chang AB. Does failed chronic wet cough response to antibiotics predict bronchiectasis? Arch Dis Child **2014**; 99(6): 522-5.
- 58. Lewis TC, Stout JW, Martinez P, et al. Prevalence of asthma and chronic respiratory symptoms among Alaska Native children. Chest **2004**; 125(5): 1665-73.
- Seitz AE, Olivier KN, Steiner CA, Montes de Oca R, Holland SM, Prevots DR. Trends and burden of bronchiectasis-associated hospitalizations in the United States, 1993-2006. Chest **2010**; 138(4): 944-9.
- 60. Australian Institute of Health and Welfare. Bronchiectasis: Australian Government, 2013:1-6.
- Valery PC, Torzillo PJ, Mulholland K, Boyce NC, Purdie DM, Chang AB. Hospital-based case-control study of bronchiectasis in indigenous children in Central Australia. Pediatr Infect Dis J 2004; 23(10): 902-8.
- Edwards EA, Asher MI, Byrnes CA. Paediatric bronchiectasis in the twenty-first century: experience of a tertiary children's hospital in New Zealand. J Paediatr Child Health 2003; 39(2): 111-7.
- 63. Singleton R, Morris A, Redding G, et al. Bronchiectasis in Alaska Native children: causes and clinical courses. Pediatr Pulmonol **2000**; 29(3): 182-7.
- 64. Chang AB, Masel JP, Boyce NC, Wheaton G, Torzillo PJ. Non-CF bronchiectasis: clinical and HRCT evaluation. Pediatr Pulmonol **2003**; 35(6): 477-83.

- Chang AB, Grimwood K, Mulholland EK, Torzillo PJ, Working Group on Indigenous Paediatric Respiratory H. Bronchiectasis in indigenous children in remote Australian communities. Med J Aust 2002; 177(4): 200-4.
- 66. Massie RJ, Olsen M, Glazner J, Robertson CF, Francis I. Newborn screening for cystic fibrosis in Victoria: 10 years' experience (1989-1998). Med J Aust **2000**; 172(12): 584-7.
- 67. Morey MJ, Cheng AC, McCallum GB, Chang AB. Accuracy of cough reporting by carers of Indigenous children. J Paediatr Child Health **2013**; 49(3): E199-203.
- 68. Massie J, Curnow L, Gaffney L, Carlin J, Francis I. Declining prevalence of cystic fibrosis since the introduction of newborn screening. Arch Dis Child **2010**; 95(7): 531-3.
- Australian Bureau of Statistics. Population distribution, aboriginal and Torres Strai Islander Australians;. 4705.02006.
- 70. Laumbach RJ. Outdoor air pollutants and patient health. Am Fam Physician **2010**; 81(2): 175-80.
- Asilsoy S, Bayram E, Agin H, et al. Evaluation of chronic cough in children. Chest 2008;
 134(6): 1122-8.
- 72. Field CE. Bronchiectasis in childhood; aetiology and pathogenesis, including a survey of 272 cases of doubtful irreversible bronchiectasis. Pediatrics **1949**; 4(2): 231-48.
- 73. Irwin RS. Diagnosis and Management of Cough Executive Summary: ACCP Evidence-Based Clinical Practice Guidelines. Chest **2006**; 129(1_suppl): 1S-23S.
- 74. Pohunek P, Svobodova T. Bacterial bronchitis and chronic wet lung. In: Eber E, Midulla F. ERS handbook: Paediatric Respiratory Medicine:. Sheffield, UK: The European Respiratory Society, **2013**:266-9.
- 75. Donnelly D, Critchlow A, Everard ML. Outcomes in children treated for persistent bacterial bronchitis. Thorax **2007**; 62(1): 80-4.
- 76. Kompare M, Weinberger M. Protracted Bacterial Bronchitis in Young Children: Association with Airway Malacia. The Journal of pediatrics **2012**; 160(1): 88-92.

- 77. Zgherea D, Pagala S, Mendiratta M, Marcus MG, Shelov SP, Kazachkov M. Bronchoscopic findings in children with chronic wet cough. Pediatrics **2012**; 129(2): e364-9.
- 78. Douglas AC, Somner AR, Marks BL, Grant IW. Effect of antibiotics on purulent sputum in chronic bronchitis and bronchiectasis. Lancet **1957**; 273(6988): 214-8.
- 79. Boyd GL. Bronchiectasis in Children. Can Med Assoc J **1931**; 25(2): 174-82.
- Fitch PS, Brown V, Schock BC, Taylor R, Ennis M, Shields MD. Chronic cough in children: bronchoalveolar lavage findings. Eur Respir J **2000**; 16(6): 1109-14.
- Cole PJ. Inflammation: a two-edged sword--the model of bronchiectasis. Eur J Respir Dis Suppl **1986**; 147: 6-15.
- Priftis KN, Litt D, Manglani S, et al. Bacterial bronchitis caused by Streptococcus pneumoniae and nontypable Haemophilus influenzae in children: the impact of vaccination. Chest 2013; 143(1): 152-7.
- 83. Hare KM, Binks MJ, Grimwood K, Chang AB, Leach AJ, Smith-Vaughan H. Culture and PCR detection of Haemophilus influenzae and Haemophilus haemolyticus in Australian Indigenous children with bronchiectasis. J Clin Microbiol **2012**; 50(7): 2444-5.
- 84. Marchant JM, Morris P, Gaffney JT, Chang AB. Antibiotics for prolonged moist cough in children. Cochrane database of systematic reviews (Online) **2005**; (4): CD004822.
- 85. Marchant J, Masters IB, Champion A, Petsky H, Chang AB. Randomised controlled trial of amoxycillin clavulanate in children with chronic wet cough. Thorax **2012**; 67(8): 689-93.
- 86. Narang R, Bakewell K, Peach J, et al. Bacterial distribution in the lungs of children with protracted bacterial bronchitis. PLoS ONE **2014**; 9(9): e108523.
- 87. Van Eldere J, Slack MP, Ladhani S, Cripps AW. Non-typeable Haemophilus influenzae, an under-recognised pathogen. Lancet Infect Dis **2014**.
- Kapur N, Mackay IM, Sloots TP, Masters IB, Chang AB. Respiratory viruses in exacerbations of non-cystic fibrosis bronchiectasis in children. Arch Dis Child **2014**; 99(8): 749-53.

- 89. Becroft DM. Bronchiolitis obliterans, bronchiectasis, and other sequelae of adenovirus type
 21 infection in young children. J Clin Pathol **1971**; 24(1): 72-82.
- 90. Fox JP, Brandt CD, Wassermann FE, et al. The virus watch program: a continuing surveillance of viral infections in metropolitan New York families. VI. Observations of adenovirus infections: virus excretion patterns, antibody response, efficiency of surveillance, patterns of infections, and relation to illness. Am J Epidemiol **1969**; 89(1): 25-50.
- 91. Franz A, Adams O, Willems R, et al. Correlation of viral load of respiratory pathogens and co-infections with disease severity in children hospitalized for lower respiratory tract infection. J Clin Virol **2010**; 48(4): 239-45.
- 92. Antunes H, Rodrigues H, Silva N, et al. Etiology of bronchiolitis in a hospitalized pediatric population: prospective multicenter study. J Clin Virol **2010**; 48(2): 134-6.
- Calvo C, Pozo F, Garcia-Garcia ML, et al. Detection of new respiratory viruses in hospitalized infants with bronchiolitis: a three-year prospective study. Acta Paediatr 2010; 99(6): 883-7.
- 94. Stempel HE, Martin ET, Kuypers J, Englund JA, Zerr DM. Multiple viral respiratory pathogens in children with bronchiolitis. Acta Paediatr **2009**; 98(1): 123-6.
- 95. Bateman ED, Hayashi S, Kuwano K, Wilke TA, Hogg JC. Latent adenoviral infection in follicular bronchiectasis. Am J Respir Crit Care Med **1995**; 151(1): 170-6.
- 96. Teper AM, Kofman CD, Maffey AF, Vidaurreta SM. Lung function in infants with chronic pulmonary disease after severe adenoviral illness. J Pediatr **1999**; 134(6): 730-3.
- 97. Simila S, Linna O, Lanning P, Heikkinen E, Ala-Houhala M. Chronic lung damage caused by adenovirus type 7: a ten-year follow-up study. Chest **1981**; 80(2): 127-31.
- 98. Warner JO. Factors predisposing to abnormal pulmonary function after adenovirus type 7 pneumonia. Arch Dis Child **1985**; 60(1): 82.

- Edmond K, Scott S, Korczak V, et al. Long term sequelae from childhood pneumonia; systematic review and meta-analysis. PLoS ONE **2012**; 7(2): e31239.
- 100. Kalu SU, Loeffelholz M, Beck E, et al. Persistence of adenovirus nucleic acids in nasopharyngeal secretions: a diagnostic conundrum. The Pediatric infectious disease journal 2010; 29(8): 746-50.
- 101. Macek V, Sorli J, Kopriva S, Marin J. Persistent adenoviral infection and chronic airway obstruction in children. Am J Respir Crit Care Med **1994**; 150(1): 7-10.
- 102. Garnett CT, Erdman D, Xu W, Gooding LR. Prevalence and quantitation of species C adenovirus DNA in human mucosal lymphocytes. J Virol **2002**; 76(21): 10608-16.
- 103. Neumann R, Genersch E, Eggers HJ. Detection of adenovirus nucleic acid sequences in human tonsils in the absence of infectious virus. Virus Res **1987**; 7(1): 93-7.
- 104. Matsuse T, Hayashi S, Kuwano K, Keunecke H, Jefferies WA, Hogg JC. Latent adenoviral infection in the pathogenesis of chronic airways obstruction. Am Rev Respir Dis **1992**; 146(1): 177-84.
- 105. Thorburn K, Harigopal S, Reddy V, Taylor N, van Saene HKF. High incidence of pulmonary bacterial co-infection in children with severe respiratory syncytial virus (RSV) bronchiolitis. Thorax **2006**; 61(7): 611-5.
- 106. Wolf DG, Greenberg D, Shemer-Avni Y, Givon-Lavi N, Bar-Ziv J, Dagan R. Association of human metapneumovirus with radiologically diagnosed community-acquired alveolar pneumonia in young children. J Pediatr **2010**; 156(1): 115-20.
- 107. Giebink GS. Studies of Streptococcus pneumoniae and influenza virus vaccines in the chinchilla otitis media model. Pediatr Infect Dis J **1989**; 8(1 Suppl): S42-4.
- Suzuki K, Bakaletz LO. Synergistic effect of adenovirus type 1 and nontypeable Haemophilus influenzae in a chinchilla model of experimental otitis media. Infect Immun 1994; 62(5): 1710-8.

- Ohashi Y, Nakai Y, Esaki Y, Ohno Y, Sugiura Y, Okamoto H. Influenza A virus-induced otitis media and mucociliary dysfunction in the guinea pig. Acta Otolaryngol Suppl **1991**; 486: 135-48.
- Park K, Bakaletz LO, Coticchia JM, Lim DJ. Effect of influenza A virus on ciliary activity and dye transport function in the chinchilla eustachian tube. Ann Otol Rhinol Laryngol **1993**; 102(7): 551-8.
- 111. McBride TP, Doyle WJ, Hayden FG, Gwaltney JM, Jr. Alterations of the eustachian tube, middle ear, and nose in rhinovirus infection. Arch Otolaryngol Head Neck Surg **1989**; 115(9): 1054-9.
- Doyle WJ, Skoner DP, Hayden F, Buchman CA, Seroky JT, Fireman P. Nasal and otologic effects of experimental influenza A virus infection. Ann Otol Rhinol Laryngol **1994**; 103(1): 59-69.
- 113. Turner BW, Cail WS, Hendley JO, et al. Physiologic abnormalities in the paranasal sinuses during experimental rhinovirus colds. J Allergy Clin Immunol **1992**; 90(3 Pt 2): 474-8.
- 114. Bakaletz LO. Viral potentiation of bacterial superinfection of the respiratory tract. Trends Microbiol **1995**; 3(3): 110-4.
- 115. Mandell GL, Bennett JE, Raphael D. Mandell, Douglas and Bennett's Principles and Practice of Infectious Diseases. 6th edition ed. Philadelphia: Elsevier, **2010**.
- 116. Adrian T, Schafer G, Cooney MK, Fox JP, Wigand R. Persistent enteral infections with adenovirus types 1 and 2 in infants: no evidence of reinfection. Epidemiol Infect **1988**; 101(3): 503-9.
- 117. Patel J, Faden H, Sharma S, Ogra PL. Effect of respiratory syncytial virus on adherence, colonization and immunity of non-typable Haemophilus influenzae: implications for otitis media. Int J Pediatr Otorhinolaryngol **1992**; 23(1): 15-23.
- 118. Plotkowski MC, Bajolet-Laudinat O, Puchelle E. Cellular and molecular mechanisms of bacterial adhesion to respiratory mucosa. Eur Respir J **1993**; 6(6): 903-16.

- 119. Hakansson A, Kidd A, Wadell G, Sabharwal H, Svanborg C. Adenovirus infection enhances in vitro adherence of Streptococcus pneumoniae. Infect Immun **1994**; 62(7): 2707-14.
- 120. Korppi M, Leinonen M, Makela PH, Launiala K. Mixed infection is common in children with respiratory adenovirus infection. Acta Paediatr Scand **1991**; 80(4): 413-7.
- 121. Thorburn K, Harigopal S, Reddy V, Taylor N, van Saene HK. High incidence of pulmonary bacterial co-infection in children with severe respiratory syncytial virus (RSV) bronchiolitis. Thorax **2006**; 61(7): 611-5.
- 122. Zimmerman B, Silverman FS, Tarlo SM, Chapman KR, Kubay JM, Urch B. Induced sputum: comparison of postinfectious cough with allergic asthma in children. J Allergy Clin Immunol **2000**; 105(3): 495-9.
- 123. Gibson PG, Chang AB, Glasgow NJ, Holmes PW, Landau LI, Newcombe P. CICADA: Cough in Children and Adults: Diagnosis and Assessment. Australian cough guidelines summary statement. The Medical journal of Australia **2010**; 192(5): 265-71.
- 124. Pasteur MC, Helliwell SM, Houghton SJ, et al. An investigation into causative factors in patients with bronchiectasis. Am J Respir Crit Care Med **2000**; 162(4 Pt 1): 1277-84.
- 125. King P, Holdsworth S, Freezer N, Holmes P. Bronchiectasis. Intern Med J 2006; 36(11): 729-37.
- 126. Pritchard MG, Lenney W, Gilchrist FJ. Outcomes in children with protracted bacterial bronchitis confirmed by bronchoscopy. Arch Dis Child **2014**.
- 127. Grissell TV, Chang AB, Gibson PG. Reduced toll-like receptor 4 and substance P gene expression is associated with airway bacterial colonization in children. Pediatr Pulmonol 2007; 42(4): 380-5.
- 128. Parkin J, Cohen B. An overview of the immune system. Lancet **2001**; 357(9270): 1777-89.
- 129. Whitsett JA, Alenghat T. Respiratory epithelial cells orchestrate pulmonary innate immunity. Nat Immunol **2014**; 16(1): 27-35.

- 130. Wurzel DF, Marchant JM, Clark JE, et al. Wet cough in children: infective and inflammatory characteristics in broncho-alveolar lavage fluid. Pediatr Pulmonol **2014**; 49(6): 561-8.
- 131. Wurzel DF, Marchant JM, Clark JE, et al. Respiratory virus detection in nasopharyngeal aspirate versus bronchoalveolar lavage is dependent on virus type in children with chronic respiratory symptoms. J Clin Virol **2013**; 58(4): 683-8.
- 132. Wurzel DF, Marchant JM, Yerkovich ST, et al. Prospective characterization of protracted bacterial bronchitis in children. Chest **2014**; 145(6): 1271-8.
- 133. Wurzel DF, Mackay IM, Marchant JM, et al. Adenovirus Species C Is Associated With Chronic Suppurative Lung Diseases in Children. Clin Infect Dis **2014**; 59(1): 34-40.
- 134. Garnett CT, Talekar G, Mahr JA, et al. Latent species C adenoviruses in human tonsil tissues. J Virol **2009**; 83(6): 2417-28.
- 135. Matsuse T, Hayashi S, Kuwano K, Keunecke H, Jefferies WA, Hogg JC. Latent adenoviral infection in the pathogenesis of chronic airways obstruction. Am Rev Respir Dis **1992**; 146(1): 177-84.
- 136. Wurzel D, Marchant JM, ST Y, JW U, HL P, AB C. Protracted Bacterial Bronchitis in Children: natural history and risk factors for bronchiectasis. . In progress.
- 137. Birring SS, Ing AJ, Chan K, et al. Obstructive sleep apnoea: a cause of chronic cough.Cough (London, England) 2007; 3: 7.
- 138. Sundar KM, Daly SE. Chronic cough and OSA: a new association? J Clin Sleep Med **2011**;7(6): 669-77.
- 139. Gurgel RK, Brookes JT, Weinberger MM, Smith RJ. Chronic cough and tonsillar hypertrophy: a case series. Pediatr Pulmonol **2008**; 43(11): 1147-9.
- 140. Teodorescu M, Barnet JH, Hagen EW, Palta M, Young TB, Peppard PE. Association between asthma and risk of developing obstructive sleep apnea. JAMA **2015**; 313(2): 156-64.

- 141. Bhattacharjee R, Choi BH, Gozal D, Mokhlesi B. Association of adenotonsillectomy with asthma outcomes in children: a longitudinal database analysis. PLoS Med **2014**; 11(11): e1001753.
- 142. Kheirandish-Gozal L, Dayyat EA, Eid NS, Morton RL, Gozal D. Obstructive sleep apnea in poorly controlled asthmatic children: effect of adenotonsillectomy. Pediatr Pulmonol 2011; 46(9): 913-8.
- 143. Kheirandish-Gozal L, Gozal D. Obesity, asthma, and sleep-disordered breathing. J Pediatr
 2012; 160(5): 713-4.

APPENDIX 1

SUPPLEMENTARY MATERIAL FOR CHAPTERS 5 AND 6



CHEST

Prospective Characterization of Protracted Bacterial Bronchitis in Children

Danielle F. Wurzel, MBBS; Julie M. Marchant, PhD; Stephanie T. Yerkovich, PhD; John W. Upham, PhD; Ian M. Mackay, PhD; I. Brent Masters, PhD; and Anne B. Chang, PhD

e-Appendix 1.

Bronchoscopy and BAL were undertaken as per European Respiratory Society guidelines.[1] The first aliquot was utilized for microbiology (bacteria and viruses), the second for total cell counts and cellular differentials.[2, 3] Bacterial infection was defined as growth of $\geq 10^4$ colony-forming units (cfu)/mL of any of the following pathogens: *Streptococcus pneumoniae*, *Moraxella catarrhalis*, *Haemophilus influenza* or *Staphylococcus aureus* using standard culture methods previously described.[3]

REFERENCES

- de Blic J, Midulla F, Barbato A, et al. Bronchoalveolar lavage in children. ERS Task Force on bronchoalveolar lavage in children. European Respiratory Society. *Eur Respir J*. 2000;15(1):217-31.
- Marchant JM, Masters IB, Taylor SM, et al. Evaluation and outcome of young children with chronic cough. *Chest.* 2006;129(5):1132-41.
- 3. Marchant JM, Gibson PG, Grissell TV, et al. Prospective assessment of protracted bacterial bronchitis: airway inflammation and innate immune activation. *Pediatr Pulmonol.* 2008;43(11):1092-9.

Online supplements are not copyedited prior to posting.

1

©2014 American College of Chest Physicians. Reproduction of this article is prohibited without written permission from the American College of Chest Physicians. See online for more details. DOI: 10.1378/chest.13-2442 CHEST 2014; 145(6):1271-1278

E-Supplement – Chapter 6

Protracted Bacterial Bronchitis in Children: Natural History and Risk Factors for Bronchiectasis

Indications for flexible bronchoscopy in controls

Indications for bronchoscopy in the 15 control participants undergoing respiratory evaluation included: stridor in 6 (40%), apnoeas in 3 (20%), dyspnoea in 3 (20%), haemoptysis in 1 (7%), congenital lung lesion in 1 (7%) and imaging abnormality in 1 (7%).

H. influenzae typing

H. influenzae characterisation was based upon colony morphology and hemin and nicotinamide adenine dinucleotide (X and V factor) dependence.³⁶ Non-typeable *H. influenzae* (NTHi) were identified by failure to react with capsular antisera (Phadebact® Haemophilus coagglutination test). A *fucP*-based real time PCR assay was used for differentiation of NTHi from *H. haemolyticus* ³⁷.

Characteristic	Recurrent PBB (N=67)	Non- recurrent PBB (N=94)	Odds ratio (95% CI)	P- value
Sex, M:F, %male	48:18, 73%	57:36, 61%	1.767 (0.90, 3.49)	0.101
Recruitment age, mths, median (IQR)	24 (16, 53)	20 (12, 49)	1.003 (0.994, 1.012)	0.508

eTable 1: Univariate analysis of predictors of recurrent PBB (>3 episodes/yr)¹

Prior x-ray confirmed pneumonia	16 (24%)	19 (20%)	1.24 (0.583, 2.63)	0.578
Household tobacco smoke exposure	22 (33%)	33 (36%)	0.89 (0.46, 1.73)	0.728
Maternal smoking in pregnancy	13/60 (22%)	8/36 (22%)	0.97 (0.36, 2.62)	0.949
Aboriginal or Torres Strait Islander	1 (2%)	9 (10%)	0.143 (0.018, 1.158)	0.068
No. children in household, median (IQR)	2 (2, 3)	2 (2, 3)	1.005 (0.784, 1.288)	0.971
\geq 3 children	22 (33%)	33 (35%)	0.904 (0.466, 1.753)	0.765
Childcare attendance, ever	37/41 (90%)	42/50 (84%)	1.762 (0.490, 6.331)	0.385
TCC x 10 ⁶ /L	260 (100, 420)	200 (130, 415)	1.000 (0.999, 1.000)	0.476
BAL neutr%, median (IQR)	25 (8, 55)	27 (11, 56)	0.997 (0.986, 1.010)	0.680
BAL macro%, median (IQR)	63 (30, 79)	60 (35, 80)	1.002 (0.990, 1.014)	0.734
BAL lymph%, median (IQR)	6 (5, 12)	9 (4, 15)	0.997 (0.964, 1.032)	0.869
BAL eosin%, median (IQR)	0 (0, 0)	0 (0, 0)	1.164 (0.926, 1.463)	0.194
HAdV on BAL	10/60 (17%)	19/90 (21%)	0.747 (0.320, 1.743)	0.500
H. influenzae [*]	30 (45%)	53 (56%)	0.627 (0.334, 1.179)	0.147
M. catarrhalis*	16 (24%)	31 (33%)	0.638 (0.314, 1.293)	0.212
S. pneumoniae*	18 (27%)	26 (218%)	0.961 (0.475, 1.943)	0.911
------------------------	----------	-----------	----------------------	-------
S. aureus [*]	6 (9%)	7 (7%)	1.222 (0.392, 3.817)	0.729

APPENDIX 2

PUBLICATIONS NOT INCLUDED IN THIS THESIS

Chang AB, Yerkovich ST, Gibson PG, Anderson-James S, Petsky HL, Carroll ML, Masters IB, Marchant JM, **Wurzel DF**, Upham JW. *Pulmonary innate immunity in children with protracted bacterial bronchitis*. The Journal of Pediatrics. 2012;161(4): 621-5.

Wurzel DF, Masters IB, Isles AF. *A Case For Early Bronchoscopic Airway Assessment After Disc Battery Ingestion.* Pediatric Pulmonology 2014 Mar;49(3):E72-4. doi: 10.1002/ppul.22858.

Pulmonary Innate Immunity in Children with Protracted Bacterial Bronchitis

Anne B. Chang, PhD^{1,2}, Stephanie T. Yerkovich, PhD^{3,4}, Peter G. Gibson, MD⁵, Sophie Anderson-James, BSN¹, Helen L. Petsky, BSN¹, Melanie L. Carroll, BSc³, I. Brent Masters, PhD¹, Julie M. Marchant, PhD¹, Danielle Wurzel, MBBS¹, and John W. Upham, PhD³

Objective To determine bronchoalveolar lavage (BAL) levels of 3 innate immunity components (human β -defensin-2 [hBD2], mannose-binding lectin [MBL], and surfactant protein-A [SP-A]), the relationship with airway neutrophilia and infection, and cytokine production of stimulated BAL cells in children with current protracted bacterial bronchitis (PBB), children with resolved PBB (PBB well), and controls.

Study design BAL of 102 children (mean age 2.8 years) fulfilling predefined criteria of current PBB (n = 61), PBB well (n = 20), and controls (n = 21) was cultured (quantitative bacteriology) and viruses examined by polymerase chain reaction. hBD2, MBL, and SP-A were measured, and cytokine production by lipopolysaccharide-stimulated BAL cells was determined.

Results Median hBD2 and MBL levels were significantly higher in the current PBB group (hBD2 = 164.4, IQR 0-435.5 pg/mL; MBL = 1.7, 0.4-4 ng/mL) than in the PBB well group (hBD2 = 0, IQR 0-85.2; MBL = 0.6, IQR 0.03-2.9) and controls (hBD2 = 3.6, IQR 0-126; MBL = 0.4, IQR 0.02-79). hBD2 was significantly higher in children with airway infection (n = 54; median 76.9, IQR 0-397.3) compared with those without (n = 48; 0, IQR 0-236.3), P = .04. SP-A levels and cytokine production of stimulated BAL cells were similar between groups.

Conclusion In children's airways, hBD2, but not MBL and SP-A, relates to inflammation and infection. In children with PBB, mechanisms involving airway hBD2 and MBL are augmented. These pulmonary innate immunity components and the ability of BAL cells to respond to stimuli are unlikely to be deficient. (*J Pediatr 2012;161:621-25*).

rotracted bacterial bronchitis (PBB) is a relatively new and significant diagnostic clinical entity characterized by chronic wet cough and resolution of cough within 2 weeks of treatment with appropriate antibiotics.^{1,2} PBB, increasingly recognized as an important pediatric condition worldwide,²⁻⁶ is a common cause of chronic cough in children.² Not only is PBB easily treatable (thus improving health-related quality of life^{7,8}) but, if left untreated, it may progress to chronic suppurative lung disease and radiological bronchiectasis.^{3,7}

The lower airways of children with PBB are characterized by bacterial infection and airway neutrophilia.^{1,7} The latter suggests that pulmonary innate immunity likely plays a pivotal role.^{9,10} However, there are few prospective studies on PBB; only 2^{9,10} have evaluated mechanistic links. Innate immune defense involves recognition and clearing responses. In the recognition arm, specific pattern recognition receptors are key components and include Toll-like receptors (TLRs) and lectin-like molecules such as, surfactant protein-A (SP-A) and mannose-binding lectin (MBL).¹¹ Pulmonary collectins (eg, SP-A, MBL) play roles in pulmonary innate immunity and regulate inflammatory responses.^{12,13} SP-A also has antimicrobial properties.¹⁴ Serum MBL deficiency is associated with bronchiectasis but there is only 1 article on MBL in children's bronchoalveolar lavage (BAL). Fidler et al¹⁵ reported MBL in the BAL of children with lung infection. The MBL levels were not associated with bacteria or systemic inflammation but

correlated with neutrophil elastase.¹⁵ However, quantitative bacteria culture was not defined, polymerase chain reaction for viruses was undertaken in only 4 children, and no other innate immunity components were examined.¹⁵

The clearing mechanism of the innate immune system includes antimicrobials such as the defensin family. In the lungs, epithelial production of β -defensin sub-types 1-3 are likely important.^{14,16} In the middle ear, human β -defensin-2 (hBD2) has bactericidal activity against *Streptococcus pneumoniae*, *Moraxella catarrhalis*,

BAL	Bronchoalveolar lavage
hBD2	Human β -defensin-2
IL	Interleukin
LPS	Lipopolysaccharide
MBL	Mannose-binding lectin
PBB	Protracted bacterial bronchitis
SP-A	Surfactant protein-A
TLR	Toll-like receptor

From the ¹Queensland Children's Respiratory Centre, and Queensland Children's Medical Research Institute, Royal Children's Hospital, Brisbane; ²Child Health Division, Menzies School of Health Research, Charles Darwin University, Darwin, Northern Territory; ³School of Medicine, The University of Queensland; ⁴Queensland Centre for Pulmonary Transplantation and Vascular Disease, The Prince Charles Hospital, Brisbane; and ⁵Department of Respiratory Medicine, John Hunter Hospital, University of Newcastle, New South Wales, Australia

Funded by the Financial Markets Foundation for Children (grant 2010-005) and the Royal Children's Hospital Foundation. A.C. (grant 545216), P.G. (grant 569240), and J.U. (grant 511019) are supported by National Heath and Medical Research Council fellowships. The study sponsors had no role in study design; the collection, analysis, and interpretation of data; the writing of the report; or the decision to submit the paper for publication. The authors declare no conflicts of interest.

0022-3476/\$ - see front matter. Copyright © 2012 Mosby Inc. All rights reserved. 10.1016/j.jpeds.2012.03.049 and nontypeable *Haemophilus influenzae*.¹⁷ These bacteria are the same dominant pathogens in the lower airways of children with PBB⁷ and bronchiectasis.¹⁸ Yet there is only 1 small study (n = 20) that has measured airway human β -defensin in children.¹⁶ In children, human β -defensin is relevant to several conditions including inflammatory gut disease like Crohn's disease.¹⁷

In a cohort of 102 children, we determined whether BAL levels of hBD2, SP-A, and MBL: (1) differed between children with current PBB, children with resolved PBB (PBB well), and controls; and (2) were related with airway neutrophilia and endobronchial infection. We further examined if the cytokine production of lipopolysaccharide (LPS)-stimulated BAL cells were different in these groups. We tested the hypothesis that current PBB would be associated with deficient levels of hBD2, SP-A, and MBL.

Methods

Children undergoing flexible bronchoscopy for clinical reasons were eligible for enrollment. Informed consent was obtained and the study was approved by the Ethics Committees of the Royal Children's Hospital (No. 2003/017) and the University of Queensland School of Medicine (No. 2008000064). Enrolled children had a standardized medical history taken that focused on the respiratory system including cough quality (wet/dry)¹⁹ and key demographics that contribute to frequency of respiratory infections (day care attendance, smoke exposure, number of siblings in household). Post bronchoscopy, children with a wet cough were treated with amoxillin-clauvanate (45 mg/kg/d in 2 doses).¹ A validated cough diary card²⁰ was used to document response to antibiotics, defined as absence of cough or >75% reduction in score (for \geq 3 days) within 2 weeks of antibiotic use post bronchoscopy.

Flexible bronchoscopy was performed under general anesthesia and BAL was obtained using European Respiratory Society guidelines as previously described¹; 3 aliquots of sterile normal saline were instilled. The first aliquot was used for microbiology examination; the second and third aliquots were pooled for cytology, fluid phase measurements, and cell cultures. Microbiological assessment (previously described^{1,21}) included quantitative aerobic cultures of bacteria and positive bacterial culture and was defined as growth of $\geq 10^5$ colony-forming units/mL.^{9,10} Viral assessment on BAL to detect respiratory syncytial virus, adenovirus, parainfluenza virus types 1-3, influenza, *Mycoplasma pneumoniae*, and human meta-pneumovirus²¹ were performed using polymerase chain reaction.²¹

For this study, only children who fulfilled the clinical definitions of PBB, PBB well, and controls were included. Children were classified before innate immunity studies were performed. We defined children as PBB if there was a history of chronic (>4 weeks) wet cough and response to antibiotic treatment with resolution of the cough within 2 weeks in the absence of signs and symptoms of other disease.² PBB well were children who previously had a wet cough that had responded to antibiotics and were treated by a pediatric pulmonologist but were no longer coughing when the bronchoscopy was undertaken. Controls were children having bronchoscopy for the assessment of the airways (eg, stridor) with no history of chronic cough and no acute respiratory infection in the preceding 2 weeks. In all children, immune deficiencies and cystic fibrosis were excluded through serum immunoglobulins (IgG, IgM, IgA), neonatal screening for controls, and sweat test for PBB and PBB well. Children with recurrent pneumonia (>2), bronchiectasis (on high-resolution computed tomography scan), and neurological abnormalities were also excluded.

Measurement of hBD2, MBL, and SP-A

The cellular and supernatant components of the BAL fluid were separated within 3 hours of collection, with the cellular portion used immediately to assess cytokine production and the supernatant stored at -20° C for later batch analysis. hBD2 (Phoenix Pharmaceuticals, Burlingame, California), MBL (Bioporto Diagnostics, Gentofte, Denmark), and SP-A (Biovendor, Modrice, Czech Republic) were measured using commercial kits according to the manufacturers' directions. The limits of detection were 3.9 pg/mL (hBD2), 200 pg/mL (MBL), and 500 pg/mL (SP-A).

Measurement of Cytokine Production by LPS-Stimulated BAL Cells

Where available, BAL cells were resuspended in Roswell Park Memorial Institute with 10% fetal calf serum at 2×10^4 cells/ well in round-bottom culture plates. Cells were stimulated with 1 ng/mL LPS (Alexis Biochemicals, Lausen, Switzerland) for 24 hours at 37°C. Cytokine (interleukin [IL]-6, IL-8, IL-10, and tumor necrosis factor- α) production was measured in culture supernatants using in-house enzyme-linked immunosorbent assays with commercially available paired antibodies and recombinant cytokines (Becton Dickinson, Franklin Lakes, New Jersey). The limit of detection was 15 pg/m.

Statistical Analyses

Data were examined for type of distribution using normality plots. Data that had a normal distribution were described using mean and SD values; medians and IQRs were used for nonparametric data. Chi-squared tests were used for comparison of categorical data, Mann-Whitney for 2-group comparisons, Kruskal-Wallis used for >2-group comparisons, and ANOVA for age comparison between groups. Spearman correlation (r_s) was used to examine correlations among variables. The "enter" method was used for multiple regression analyses. SPSS (SPSS Inc, Chicago, Illinois) was used, and a 2-tailed *P* value of <.05 was considered significant.

Results

Of the 164 children enrolled, 62 were excluded because they did not fulfil the criteria of current PBB, PBB well, or controls. The demographics of the cohort of the 102 children (**Table**) showed that the groups significantly differed in age

Table. Baseline data				
Demographics	Current PBB group (n = 61)	PBB well group ($n = 20$)	Control group (n = 21)	Р
Age, y (mean, SD)	2.5 (2.3)	4.2 (3.0)	2.2 (2.8)	.018
Sex, M:F	47:14	10:10	11:10	.025
Length of current cough, wk*	28 (8-57)	0	0	.0001
Household tobacco smoke exposure, n (%)	20 (32.8%)	6 (30%)	7 (33.3%)	.97
No. of children in household (median, range)	2 (1-8)	2 (1-10)	2 (1-6)	.13
BAL data (median, IQR)				
Total cell count ($\times 10^{6}$ /L)	305 (157-498)	131 (78.5-277.5)	100 (45-139)	.0001
% Macrophage	45.5 (26.0-68.5)	87.0 (80.5-93.0)	91.0 (81-95)	.0001
% Neutrophil	41.5 (18.7-69.0)	5.0 (2.5-7.8)	2.0 (1.0-8.0)	.0001
% Eosinophil	0 (0-8.0)	0 (0-0.0)	0 (0-0.0)	.021
% Lymphocyte	9.0 (4.0-12.0)	5.5 (4.0-12.0)	5.0 (3.0-9.0)	.171

Bolded values are P < .05. *From day of bronchoscopy.

and sex. The PBB group had the highest number of boys, and children in the PBB well group were older. There was no significant difference in exposure to tobacco smoke or number of children in the household.

Analysis by Clinical Group

Figure 1 depicts the significantly higher median hBD2 level in the PBB group compared with the PBB well and the control groups. There was no significant difference between the PBB well group and controls. For MBL, the PBB group also had significantly higher levels than controls but was not significantly different from the PBB well group (**Figure 2**). There was no difference in SP-A between groups (**Figure 3**, Kruskil-Wallis P = .505). Similarly, there was no difference in the LPS-stimulated cytokine response by BAL cells between the groups (**Figure 4**; available at www.jpeds.com). There was no correlation between any cytokine and the proportion of either macrophages or neutrophils within the cellular component (data not shown).

Analysis by Presence/Absence of Bacteria, Viruses, and Other Potential Factors

Considering the whole cohort, hBD2 was significantly higher in the group with BAL presence of either bacteria or virus (n = 54, median 76.9 pg/mL, IQR 0-397.3) compared with the absence of both bacteria and virus antigens (n = 48, median 0, IQR 0-236.3) (P = .04). There was no significant difference in MBL or SP-A between groups (P = .17 and .61, respectively). However, when the children were grouped by the presence of positive bacteria culture (n = 43) compared with the absent (n = 59), there was no significant difference between groups for hBD2, MBL, or SP-A (P = .393, .051, and .393, respectively). Likewise, there was no significant difference in hBD2, MBL, or SP-A when the groups were divided by presence/absence of virus (present in 18, absent in 84) (P = .196, .61, and .664, respectively).

There were modest but significant correlations between hBD2 with MBL ($r_s = 0.22$, P = .027), BAL% neutrophils ($r_s = 0.43$, P = .0001), BAL% macrophages ($r_s = 0.32$, P = .0001), and blood% neutrophils ($r_s = 0.27$, P = .010) but not with SP-A (P = .25), blood% monocytes (P = .79), age (P = .78), and sex (P = .21). MBL also significantly cor-

related with BAL% neutrophils ($r_s = 0.44$, P = .0001) and BAL% macrophages ($r_s = 0.4$, P = .0001). There was no significant correlation between MBL and any of the other factors (blood% neutrophils, SP-A, blood% monocytes, age, and sex; P = .25-.57). SP-A did not correlate to any factor (P = .25-.99).

In multivariate regression analysis, hBD2 significantly related to BAL% neutrophils (β coefficient 5, 95% CI, 1.3-8.9, P = .012) and MBL (β coefficient = 16.5, 95% CI, 7.2-25.9, P = .001) but did not relate to BAL% macrophages (P = .26), blood neutrophils (P = .13), age (P = .8), or sex (P = .7). With MBL as the dependent variable, the only significant factor in multivariate regression analysis was hBD2 (β coefficient 0.01, 95% CI, 0-0.013, P = .0001). There was no significant association in any of the other factors (BAL% neutrophils, BAL% macrophages, age, and sex, P range .23-.94). Because lower airway bacteria infection causes airway neutrophilia and both are present in those with PBB,¹ we excluded both these factors in our regression analyses.

Discussion

In a cohort of 102 children, we found that hBD2 and MBL levels in BAL were significantly higher in children with PBB



Figure 1. hBD2 levels in the BAL of children grouped into current PBB, PBB well children, and controls. Median hBD2 levels were significantly higher in the PBB group compared with the PBB well and the control groups. There was no difference between PBB well and controls.



Figure 2. MBL levels in the BAL of children grouped into current PBB, PBB well, and controls. The median MBL level in the PBB group was significantly higher than that of controls but was not significantly different from the PBB well group.

compared with children who have recovered from a PBB episode (PBB well) and controls. We have also found that hBD2 levels were associated with airway infection and are related to airway neutrophilia and MBL levels. However, SP-A levels in the BAL and cytokine production of stimulated BAL cells were similar between the groups.

Discoveries of the complex innate immune system and its contributions to health and disease are rapidly expanding. Augmentation of the innate system may have a therapeutic role,²² yet there are relatively little BAL data relating to innate immunity in children, outside of cystic fibrosis. In this study, we examined key components of the human pulmonary innate immunity defense system with respect to PBB, an increasingly recognized clinical entity.¹⁻⁷ Innate immunity may be particularly important in light of the recognition of the likely association between PBB, especially when recurrent, and chronic suppurative lung disease and bronchiectasis.^{3,7}

hBD2 is 1 of 6 β -defensins identified in humans.²³ Only hBD2-4 can be induced and produced by airway epithelial cells. As hBD2 has been shown to have bactericidal activity against *S pneumoniae*, *M catarrhalis*, and nontypeable *H influenzae* in otitis media,¹⁷ we considered it to be a key innate immunity component relevant to PBB and bronchiectasis because these bacteria are major lower airways pathogens



Figure 3. SP-A levels in the BAL of children grouped into current PBB, PBB well, and control groups. Median SP-A levels did not differ between groups.

in these conditions.^{7,18} We also examined MBL given its relevance to bronchiectasis and lung infection.¹⁵ If children with PBB have a deficiency in these selected innate immunity mechanisms, reduced levels in BAL fluid would have been found. Instead, we found increased hBD2 and MBL levels in the BAL from children with PBB compared with controls. Although the PBB well group formed an intermediate group, their values were similar to controls and different from the current PBB group. This suggests that children with PBB are unlikely to have a deficiency in the hBD2 and MBL components of the complex pulmonary innate immunity system. We also found that the cytokine production of stimulated BAL cells was similar in children with current PBB, children with resolved PBB (PBB well), and controls, supporting our data that these aspects of the pulmonary innate system are not deficient in children with PBB.

However, it remains unknown if children with recurrent PBB differ from those without recurrent PBB as some children with PBB develop recurrence while others do not.⁷ It is also possible that children with PBB have an initial suboptimal innate immune response that becomes excessive later in the course of the illness, in a similar mechanism to influenza-infected BAL cells. Didierlaurent et al showed that influenza-infected ex vivo BAL cells have an initial reduced neutrophil recruitment that later resulted in heightened bacterial load during secondary respiratory infection and increased neutrophilia.²⁴ They also showed that this was related to TLR signals.²⁴ Children with PBB in our cohort have sustained bacterial infection for many weeks and thus it is possible a differential response may be found in the early phase of the infection. Indeed, in our earlier study, we found that mRNA expression of preprotachykinin I, TLR-2, and TLR-4 in BAL cells of children who did not present to respiratory clinics (hence likely to be less severe) and had bacterial infection in their airways were lower than those of controls.¹⁰

In our current cohort, hBD2, but not MBL, varied significantly in relation to airway neutrophilia and airway infection (defined by presence of bacteria or viruses). This suggests that the epithelial–neutrophil interaction is actively engaged in PBB. hBD2 also varied significantly in relation to MBL. Reasons for this include independent stimulation of both MBL and hBD2 responses. It is also biologically plausible that MBL, which functions in part as a pattern recognition receptor and further stimulates pathogen clearance, likely involves hBD2, an important antimicrobial peptide in the lungs. A highly significant association does not prove causality, so this is merely speculative as we were unable to find any published reports that have measured both MBL and hBD2 in BAL.

In adults, increased hBD2 levels in the BAL have been associated with inflammatory lung disease and infection.²³ However, pediatric data are important in light of the known difference in mucosal immunity and other immunity of the airways of children compared with adults.²⁵ Our findings on hBD2 are in contrast to the sole published study in children¹⁶ that has measured BAL hBD2. The study by Chen et al¹⁶ did not find any relationship between hBD2 levels and airway inflammation or infection, in contrast to our study's findings. The most likely explanation relates to sample size; n = 102 in our study and n = 20 in the earlier study.¹⁶ Our data on SP-A are consistent with a study that described no change in BAL SP-A levels in children with acute lung injury.²⁶

Although we have investigated a previously unexplored area in children, we have not examined many other important components of the innate system, especially mechanisms involved in resolution of inflammation, which might explain the persistence of symptoms and neutrophilic inflammation in children with PBB. There are also many other possible contributors to PBB such as impaired airway clearance related to tracheobronchomalacia⁴ (from mechanical interference) and other causes such as mucus hypersecretion, deficient efferocytosis, and epithelial cell dysfunction, among others.

We conclude that innate immune pathways involving hBD2 and MBL are augmented in the lower airways of children with PBB. Components of the pulmonary innate system represented by BAL levels of hBD2, MBL, and SP-A and cytokine production of BAL cells are unlikely to be deficient in children with established PBB. We also conclude that in the airways, hBD2 is strongly associated with MBL, neutrophilia, and infection.

We are grateful to all the parents and children who participated in this study. We also thank Drs Buntain, Francis, Dore, and Isles for allowing us to recruit their patients into the study; Carol Willis for maintaining the database; and Julie Burel from Professor Upham's laboratory for performing the assays.

Submitted for publication Nov 10, 2011; last revision received Feb 24, 2012; accepted Mar 26, 2012.

Reprint requests: Anne B. Chang, Queensland Children's Respiratory Centre, Royal Children's Hospital, Herston, Queensland 4029, Australia. E-mail: annechang@ausdoctors.net

References

- Marchant JM, Masters IB, Taylor SM, Cox NC, Seymour GJ, Chang AB. Evaluation and outcome of young children with chronic cough. Chest 2006;129:1132-41.
- Chang AB, Landau LI, van Asperen PP, Glasgow NJ, Robertson CF, Marchant JM, et al. The Thoracic Society of Australia and New Zealand. Position statement. Cough in children: definitions and clinical evaluation. Med J Aust 2006;184:398-403.
- Douros K, Alexopoulou E, Nicopoulou A, Anthracopoulos MB, Fretzayas A, Yiallouros P, et al. Bronchoscopic and high resolution CT findings in children with chronic wet cough. Chest 2011;140:317-23.
- 4. Kompare M, Weinberger M. Protracted bacterial bronchitis in young children: association with airway malacia. J Pediatr 2012;160:88-92.
- Chipps BE. Evaluation of infants and children with refractory lower respiratory tract symptoms. Ann Allergy Asthma Immunol 2010;104: 279-83.
- 6. Donnelly DE, Critchlow A, Everard ML. Outcomes in children treated for persistent bacterial bronchitis. Thorax 2007;62:80-4.

- Chang AB, Redding GJ, Everard ML. State of the art. Chronic wet cough: protracted bronchitis, chronic suppurative lung disease and bronchiectasis. Pediatr Pulmonol 2008;43:519-31.
- Marchant JM, Newcombe PA, Juniper EF, Sheffield JK, Stathis SL, Chang AB. What is the burden of chronic cough for families? Chest 2008;134:303-9.
- **9.** Marchant JM, Gibson PG, Grissell TV, Timmins NL, Masters IB, Chang AB. Prospective assessment of protracted bacterial bronchitis: airway inflammation and innate immune activation. Pediatr Pulmonol 2008;43:1092-9.
- **10.** Grissell T, Chang AB, Gibson PG. Impaired toll-like receptor 4 and substance P gene expression is linked to airway bacterial colonisation in children. Pediatr Pulmonol 2007;42:380-5.
- Takahashi K, Ip WE, Michelow IC, Ezekowitz RA. The mannose-binding lectin: a prototypic pattern recognition molecule. Curr Opin Immunol 2006;18:16-23.
- Stuart LM, Henson PM, Vandivier RW. Collectins: opsonins for apoptotic cells and regulators of inflammation. Curr Dir Autoimmun 2006;9: 143-61.
- Sano H, Kuronuma K, Kudo K, Mitsuzawa H, Sato M, Murakami S, et al. Regulation of inflammation and bacterial clearance by lung collectins. Respirology 2006;11(Suppl):S46-50.
- Schutte BC, McCray PB Jr. [beta]-defensins in lung host defense. Annu Rev Physiol 2002;64:709-48.
- Fidler KJ, Hilliard TN, Bush A, Johnson M, Geddes DM, Turner MW, et al. Mannose-binding lectin is present in the infected airway: a possible pulmonary defence mechanism. Thorax 2009;64:150-5.
- Chen CI, Schaller-Bals S, Paul KP, Wahn U, Bals R. Beta-defensins and LL-37 in bronchoalveolar lavage fluid of patients with cystic fibrosis. J Cyst Fibros 2004;3:45-50.
- Underwood MA, Bevins CL. Defensin-barbed innate immunity: clinical associations in the pediatric population. Pediatrics 2010;125:1237-47.
- Hare KM, Grimwood K, Leach AJ, Smith-Vaughan HC, Torzillo PJ, Morris P, et al. Respiratory bacterial pathogens in the nasopharynx and lower airways of Australian Indigenous children with bronchiectasis. J Pediatr 2010;157:1001-5.
- Chang AB, Eastburn MM, Gaffney J, Faoagali J, Cox NC, Masters IB. Cough quality in children: a comparison of subjective vs. bronchoscopic findings. Respir Res 2005;6:3.
- 20. Chang AB, Newman RG, Carlin J, Phelan PD, Robertson CF. Subjective scoring of cough in children: parent-completed vs child-completed diary cards vs an objective method. Eur Respir J 1998;11:462-6.
- Chang AB, Clark R, Acworth JP, Petsky HL, Sloots TP. The impact of viral respiratory infection on the severity and recovery from an asthma exacerbation. Pediatr Infect Dis J 2009;28:290-4.
- 22. Clement CG, Evans SE, Evans CM, Hawke D, Kobayashi R, Reynolds PR, et al. Stimulation of lung innate immunity protects against lethal pneumococcal pneumonia in mice. Am J Respir Crit Care Med 2008;177: 1322-30.
- Parker D, Prince A. Innate immunity in the respiratory epithelium. Am J Respir Cell Mol Biol 2011;45:189-201.
- 24. Didierlaurent A, Goulding J, Patel S, Snelgrove R, Low L, Bebien M, et al. Sustained desensitization to bacterial Toll-like receptor ligands after resolution of respiratory influenza infection. J Exp Med 2008;205: 323-9.
- Heier I, Malmstrom K, Sajantila A, Lohi J, Makela M, Jahnsen FL. Characterisation of bronchus-associated lymphoid tissue and antigenpresenting cells in central airway mucosa of children. Thorax 2011;66: 151-6.
- 26. Todd DA, Marsh MJ, George A, Henderson NG, Barr H, Sebastian S, et al. Surfactant phospholipids, surfactant proteins, and inflammatory markers during acute lung injury in children. Pediatr Crit Care Med 2010;11:82-91.



Figure 4. LPS-stimulated **A**, IL-6, **B**, IL-8, **C**, tumor necrosis factor- α , and **D**, IL-10 cytokine production from BAL cell showed no difference between current PBB (n = 27), PBB well (n = 15), and control (n = 13) groups.

A Case for Early Bronchoscopic Airway Assessment After Disc Battery Ingestion

Danielle F. Wurzel, MBBS, FRACP,^{1,2}* I. Brent Masters, MBBS, FRACP,PhD,^{1,2} Kelvin L. Choo, MBBS, FRACS,³ and Alan F. Isles, MBBS Msc FRACP, FRCP (C)^{1,2}

Summary. Disc battery ingestion in children is becoming increasingly common with the proliferation of small battery-powered electronic devices. In the case of esophageal impaction, the likelihood and severity of complications are proportionate to the time between ingestion and removal. Tracheo-esophageal fistulae (TOF) are a recognized complication and can be life-threatening. We describe an interesting case of disc battery ingestion with delayed recognition of a TOF. We document the tracheal mucosal healing process of a large airway defect and describe the role of bronchoscopy in guiding the timing of surgical intervention. This case highlights the important role of early bronchoscopic assessment in management of these patients. **Pediatr Pulmonol. 2014; 49:E72–E74.** (2) 2013 Wiley Periodicals, Inc.

Key words: disc battery ingestion; button battery ingestion; acquired tracheoesophageal fistula; paediatric bronchoscopy.

Funding source: Thoracic Society of Australia, National Health and Medical Research Council and Queensland Children's Medical Research Institute.

INTRODUCTION

Disc battery ingestion by young children can have devastating or even fatal consequences.¹ When impaction in the esophagus occurs, the time between ingestion and removal is critical. When batteries ≥ 20 mm in size are ingested, they are likely to lodge in the esophagus, typically at sites of physiological narrowing, for example, the thoracic inlet. Tissue necrosis occurs predominantly via hydrolysis.¹ We report on a case of delayed recognition of lithium disc battery ingestion and impaction, which resulted in severe esophageal necrosis with

¹Queensland Children's Respiratory Centre, Royal Children's Hospital, Brisbane, Australia.

²Queensland Children's Medical Research Institute, University of Queensland, Australia.

³Department of Paediatric Surgery, Royal Children's Hospital, Brisbane, Australia.

Author's Contributions: Danielle F. Wurzel: Dr. Wurzel was responsible for manuscript preparation, drafting and final submission. I Brent Masters: A/Prof Masters critically reviewed the manuscript and approved the final version. Kelvin Choo: Dr. Choo assisted in manuscript preparation, critically reviewed the manuscript and approved the final version. Alan tracheo-esophageal fistula (TOF) formation. We describe the role of bronchoscopy and advocate early airway assessment in the case of deep esophageal ulceration; initially, for TOF diagnosis, and subsequently for monitoring tracheal mucosal healing during esophageal rest.

CASE REPORT

A 13-month-old child presented with his parents to a regional general practitioner with a history of sudden onset of cough and irritability. The child had rhinorrhea

Isles: Prof. Isles assisted in manuscript preparation, critically reviewed the manuscript and approved the final version.

Conflict of interest: None.

*Correspondence to: D. F. Wurzel, MBBS, FRACP, Queensland Children's Medical Research Institute, Queensland 4006, Australia; Royal Children's Hospital, Brisbane, Australia. E-mail: daniellewurzel@netspace.net.au

Received 22 February 2013; Accepted 10 May 2013.

DOI 10.1002/ppul.22858 Published online 7 October 2013 in Wiley Online Library (wileyonlinelibrary.com).

Bronchoscopy after Disc Battery Ingestion E73



Fig. 1. A: Frontal CXR demonstrating rounded, midline opacification in the inter-clavicular region measuring roughly 2 cm in diameter with lateral bowing of the trachea to the right. The "double rim" or "halo" appearance of the opacity in the AP view confirms the foreign body to be a disc battery.¹ B: Lateral CXR demonstrating anterior orientation of negative node of battery (narrower side).

but an otherwise normal physical examination. He was presumed to have an early viral upper respiratory tract infection and was discharged home.

The family returned the following morning after the child had been unusually irritable overnight and was not tolerating oral intake. He appeared mottled and pale, and was referred to the local Emergency Department where a chest X-ray was performed (Fig. 1). Further history taking revealed the child had experienced a choking episode at 4 pm the day prior.

On examination in the Emergency Department, he was febrile at 38.0°C, tachycardic at 175 bpm, and tachypnoeic with respiratory rate of 42. He had intercostal recession, moist cough, and was drooling. His oxygen saturations were initially 88% on room air, improving with suctioning of his oral secretions. Attempts to examine him in the supine position were unsuccessful due to increased irritability.

The rounded opacity on his chest film was initially thought to be a coin lodged in his upper esophagus. After discussion with pediatric gastroenterologists at a tertiary pediatric hospital, the patient was transferred urgently for endoscopic removal. The patient arrived 23 hr after the choking episode. Three hours later, flexible endoscopy revealed a button battery impacted in the wall of the esophagus, 2.5 cm distal to the cricopharyngeus muscle. Circumferential esophageal sloughing and necrosis was noted at the impaction site. The battery was removed intact and identified as a lithium 3 V disc battery. The depth of the injury initially appeared to be confined to the esophageal mucosa.

The child remained unwell in hospital, with fevers, rhinorrhea, and moist cough. A nasopharyngeal aspirate was positive for respiratory syncytial virus. He continued

ABBREVIATION: TOF tracheo-esophageal fistula to produce large amounts of frothy secretions for several days, requiring frequent oropharyngeal suctioning and supplemental oxygen.

On day 7, a contrast study demonstrated flow of contrast into the trachea and mainstem bronchi, indicative of a TOF. The child proceeded to flexible bronchoscopy. A large posterior tracheal wall defect, 3 cm proximal to the carina, was visualized. Air bubbles could be seen passing through the defect (Fig. 2A). Ventilation during anesthesia was challenging due to gas leak from the large tracheal defect and gastric hyperinflation. Following the bronchoscopy, the child was admitted to the intensive care unit, antibiotic cover was broadened for possible mediastinitis, and surgical consultation obtained.

The surgical opinion was that primary repair of the fistula would involve a high degree of risk due to the child's compromised respiratory state; the large size of the fistula and the degree of tissue necrosis. Esophageal diversion to protect the airways was considered the safest initial option.

The esophagus, approached via a low-neck incision, was divided and diverted proximally as an esophagostomy. Via a laparotomy, the gastro-esophageal junction was also divided and a gastrostomy fashioned for enteral



Fig. 2. A: Acute phase TOF: posterior tracheal wall defect with evidence of patency and free flow of secretions and air from esophagus to trachea. B: Acute phase TOF: fronds of (pulsatile) granulation tissue protruding from the TOF site into the trachea. Pulsatility was likely due to proximity to vessels. C: Chronic phase TOF: by 10 weeks there had been significant interval healing with tracheal mucosalization. Although not visible in this image, during esophageal insufflation a small residual fistula was observed at the lower aspect of the earlier large defect. D: Chronic phase TOF: at 6 months, healing was complete however there was a small residual fistula in the lateral groove of trachea, allowing passage of 0.9 mm guide wire.

E74 Wurzel et al.

feeding. The child improved quickly and was discharged after establishment of gastrostomy feeds on day 22.

A repeat bronchoscopy, 3 weeks post-discharge, showed interval healing with "pulsatile" fronds of granulation tissue, suspicious of vascular proximity to the TOF site (Fig. 2B). A contrast magnetic resonance angiogram showed no evidence of vasculitis or aneurysm. Serial bronchoscopies were performed to monitor healing (Fig. 2C and D) of the TOF. At 6 months post-injury, despite complete resolution of inflammatory granulation tissue, a pinhole fistula remained necessitating surgical closure.

A right thoracotomy was performed, the fistula was divided, and the trachea reinforced with a pedicled intercostal muscle flap. The remaining isolated segment of intra-thoracic esophagus, badly strictured at the fistula site, was sacrificed. Three months later, esophageal reconstruction was undertaken using a retrosternally tunnelled greater curve gastric tube, and the esophagostomy was closed. When full oral nutrition was maintained, gastrostomy closure was performed. At most recent review, the patient was tolerating an oral diet and achieving normal growth and developmental milestones.

DISCUSSION

Disc battery ingestion poses a special hazard to young children, as the initial ingestion may be unwitnessed and the child often presents with non-specific symptoms.¹ This case illustrates how delays in diagnosis and treatment can occur. Although the choking episode was witnessed, the significance was not initially appreciated. The full extent of the injury was determined only after performing a flexible bronchoscopy a week after first presentation.

In the case of esophageal impaction, tissue necrosis has been documented to occur within 2–2.5 hr of battery lodgement.¹ The risk of severe injury after disc battery ingestion is also influenced by the age of the child and the battery size, residual charge, and orientation within the esophagus.¹ This case involved a large (>20 mm) 3-volt lithium manganese battery.

When a battery lodges in the esophagus, tissue injury is caused predominantly by generation of an electrolytic current that hydrolyses tissue fluids and produces hydroxide at the battery's negative terminal, the narrower side of the battery.^{2,3} If the negative terminal is positioned anteriorly in the esophagus, as it was in this case, erosion through the esophagus into the trachea may occur. If the negative terminal is positioned posteriorly, erosion into the mediastinum and aorta may occur.⁴ Bronchoscopy plays an important role in assessing the presence, location, and extent of potential airway injury.

Prior literature emphasizes the role of timely bronchoscopic airway examination in disc battery impaction with

Pediatric Pulmonology

"meticulous endoscopic examination of the esophagus and trachea after battery removal" for early diagnosis of complications.⁵ Bronchoscopy is generally considered superior to esophagoscopy for diagnosing TOF, as eosophagoscopy may not discover a small fistula hidden behind a mucosal fold.⁶ Early bronchoscopic airway assessment may facilitate more timely surgical intervention and improved symptomatic management.

Our case serves to illustrate several important points. Firstly, it reinforces previously published studies highlighting the significance of the orientation of the impacted battery within the esophagus;^{2,3} when the negative node is facing anteriorly, the risk of TOF is increased. Secondly, it demonstrates the extent of tissue injury that can occur from delayed disc battery removal, indicating the need for a high level of suspicion for TOF in children with respiratory symptoms who have ingested a battery. Lastly, it emphasizes the important role of bronchoscopic airway assessment, in the diagnosis and management of TOF after disc battery ingestion.

CONCLUSION

We recommend bronchoscopic airway assessment at the time of endoscopic battery removal, particularly if removal is delayed, the negative pole of the battery is anteriorly orientated within the esophagus and/or respiratory symptoms are present.

ACKNOWLEDGMENTS

D. Wurzel has received funding from Thoracic Society of Australia and New Zealand/Allen and Hanbury's Paediatric Respiratory grant-in-aid; National Health and Medical Research Council post-graduate scholarship (GNT1039688), and Queensland Children's Medical Research Institute.

REFERENCES

- Litovitz T, Whitaker N, Clark L, White NC, Marsolek M. Emerging battery-ingestion hazard: clinical implications. Pediatrics 2010;125:1168–1177.
- Yoshikawa T, Asai S, Takekawa Y, Kida A, Ishikawa K. Experimental investigation of battery-induced esophageal burn injury in rabbits. Crit Care Med 1997;25:2039–2044.
- Litovitz T, Whitaker N, Clark L. Preventing battery ingestions: an analysis of 8648 cases. Pediatrics 2010;125:1178–1183.
- Maves MCJ, Birck HG. Esophageal burns secondary to disk batteries in children. Ann Otol Rhino Laryngol 1984;93:364–369.
- Imamoglu M, Cay A, Kosucu P, Ahmetoglu A, Sarihan H. Acquired tracheo-esophageal fistulas caused by button battery lodged in the esophagus. Pediatr Surg Int 2004;20:292–294 (Epub 2004/05/19).
- Couraud L, Ballester MJ, Delaisement C. Acquired tracheoesophageal fistula and its management. Semin Thorac Cardiovasc Surg 1996;8:392–399 (Epub 1996/10/01).